

Solution proposals to Higher Software Engineering Education's Design-Reality Gaps in Rural India

DIPLOMA THESIS

submitted in partial fulfillment of the requirements for the degree of

Diplom-Ingenieur

in

Business Informatics

by

Lukas Bürstmayr, BSc Registration Number 01326130

to the Faculty of Informatics at the TU Wien

Advisor: Thomas Grechenig

Vienna, 16th August, 2022

Signature Author

Signature Advisor





DIPLOMARBEIT

zur Erlangung des akademischen Grades

Diplom-Ingenieur

im Rahmen des Studiums

Business Informatics

eingereicht von

Lukas Bürstmayr, BSc

Matrikelnummer 01326130

ausgeführt am Institut für Information Systems Engineering Forschungsbereich Business Informatics Forschungsgruppe Industrielle Software der Fakultät für Informatik der Technischen Universität Wien

Betreuung: Thomas Grechenig

Wien, 16. August 2022



Declaration of Authorship

Lukas Bürstmayr, BSc

I hereby declare that I have written this Diploma Thesis independently, that I have completely specified the utilized sources and resources and that I have definitely marked all parts of the work - including tables, maps and figures - which belong to other works or to the internet, literally or extracted, by referencing the source as borrowed.

Vienna, 16th August, 2022

Lukas Bürstmayr



Acknowledgements

An Paul,

für die unglaublich gute Zusammenarbeit in einer höchst dynamischen Umgebung und dein fortwährendes Engagement im ICT4D-Bereich.

An Raoul,

für die richtigen Fragen zum richtigen Zeitpunkt.

An Thomas Grechenig,

für die Betreuung dieser Arbeit und die Sicherstellung der Kooperation mit unseren indischen Partner:innen.

An Adam,

für das mir entgegengebrachte Verständnis und die Spurhalteassistenz. An Othmar & Ulrike, für die Ermöglichung meiner Ausbildung und die stete Begleitung. An Andrea, für die Bekämpfung meiner Sorgen und die Aufrechterhaltung meiner Motivation. An Paulina Johanna, für die härteste Deadline der Welt.

Von ganzem Herzen: Dankeschön!



Abstract

A challenge of the Indian higher education system that exists for more than a decade is a mismatch between the skills of the engineering graduates and the needs of the industry, especially in the field of software engineering. The reasons and recommendations for this skill mismatch have been discussed in the past and explored in the literature. One of those reasons is the high privatization of higher engineering education in India. The majority of India's private higher education sector is composed of over 38.000 affiliated colleges that implement study programs based entirely on the curricula set by their affiliating universities. Furthermore, the majority of these affiliated colleges are located in rural India.

This thesis examined 1 of 80 affiliated colleges that are located in rural India and are implementing the same curriculum of an affiliating university. Through this examination that considered the reality of the students and teachers by conducting semi-structured interviews and observing both students and teachers, 5 gaps (incomplete student skills, unmet employability goal, inadequate teacher skills, improperly implemented teaching methods, missing curriculum evaluation) between a designed curriculum and its implementations were identified with the help of a Design-Reality Gap Analysis. Additionally, 3 further challenges (ever-changing course assignments, difficulties in self-learning, excessive student workload) were found that students and teachers face in their everyday life. For each of the identified gaps, solutions were proposed by the author of this thesis that aim to bridge the respective gap with regard to the identified challenges of students and teachers. Guidance notes were created that aim to enable curriculum designers and curriculum managers to apply the proposed solutions. In future work, the proposed solutions and guidance notes shall be evaluated to determine if they are suitable to reduce the identified gaps and challenges as well as if they can be applied at other colleges located in rural India.

Keywords: Higher Education, Software Engineering Education, Computer Science Curriculum, Design-Reality Gap Analysis, Rural Education, ICT4D, India



Kurzfassung

Eine Herausforderung, vor der das indischen Hochschulsystem seit mehr als einem Jahrzehnt steht, ist die Diskrepanz zwischen den tatsächlichen Fähigkeiten von Hochschulabsolvent:innen und jenen, die von der Industrie nachgefragt werden, insbesondere im Bereich der Softwareentwicklung. Die Gründe dafür und Empfehlungen zur Aufhebung dieser Diskrepanz wurden in der Vergangenheit bereits durch die Literatur diskutiert und erforscht. Einer dieser Gründe ist das hohe Ausmaß an Privatisierung der indischen Hochschulausbildung im Ingenieurbereich. Der größte Teil des privaten indischen Hochschulsektors besteht aus über 38.000 angegliederten Colleges (Engl. "affiliated colleges"), die Studienprogramme durchführen, welche vollständig auf den Curricula jener Universitäten basieren, an welche die Colleges angegliedert sind (Engl. "affiliating university"). Die Mehrheit dieser angegliederten Colleges befindet sich im ländlichen Indien.

Durch die Untersuchung eines von 80 angegliederten Colleges, welche das selbe Curriculum einer Universität umsetzen, wurden in dieser Arbeit mit Hilfe einer "Design-Reality Gap Analysis" fünf Lücken (Engl. "gaps"; unzureichende Fähigkeiten der Studierenden, unerfüllte Ziele in Bezug auf die Beschäftigungsfähigkeit, unzureichende Fähigkeiten der Lehrenden, unsachgemäß eingesetzte Lehrmethoden, fehlende Curriculum-Evaluierung) zwischen einem Curriculum und dessen Umsetzung identifiziert. Dabei wurde die Realität der Studierenden und Lehrenden durch semi-strukturierte Interviews und Beobachtungen von Studierenden und Lehrenden erhoben. Darüber hinaus wurden drei Herausforderungen ermittelt (sich ständig ändernde Kurszuteilungen, Schwierigkeiten beim selbstgesteuerten Lernen, übermäßige Arbeitsbelastung der Studierenden), mit denen Studierende und Lehrende in ihrem Alltag konfrontiert sind. Für jede der identifizierten Lücken wurden Lösungsvorschläge erarbeitet, welche darauf abzielen, die jeweilige Lücke zu schließen. Diese Lösungen basieren auf vorhandener Literatur und beziehen die Herausforderungen der Studierenden und Lehrenden mit ein. Mit Hilfe dieser Lösungsvorschläge wurden Leitfäden (Engl. "guidance notes") erstellt, die es Curriculum-Designer:innen und Curriculum-Manager:innen ermöglichen sollen, diese Lösungsvorschläge anzuwenden. In zukünftigen Arbeiten soll festgestellt werden, ob die Lösungsvorschläge und Leitfäden dazu geeignet sind, die identifizierten Lücken und Herausforderungen zu reduzieren und auf andere Hochschulen im ländlichen Indien übertragen werden können.

Keywords: Hochschulbildung, Software-Engineering-Ausbildung, Informatik Curriculum, Design-Reality Gap Analysis, Bildung in ländlichen Räumen, ICT4D, Indien



Contents

Abstract		
K	urzfassung	xi
1	Introduction	1
	1.1 Problem Statement	. 1
	1.2 Motivation \ldots	3
	1.3 Objectives	4
	1.4 Methodology	5
	1.5 Limitations	7
	1.6 Organizational Issues	8
2	Related Work	9
	2.1 Software Engineering Education in India	9
	2.2 Higher Education Challenges of India	10
	2.3 Contributions of this thesis	13
3	Fundamentals	15
	3.1 Software Engineering Knowledge	15
	3.2 Teaching Software Engineering	19
	3.3 India's Higher Education System at a glance	22
4	Data Collection & Thematic Analysis	31
	4.1 Methodology	. 31
	4.2 Thematic Analysis of Interviews and Observations	38
5	Design-Reality Gap Analysis	59
	5.1 Methodology \ldots	59
	5.2 Design-Reality Gap Analysis of the Curriculum and its Implementation	n 61
	5.3 Identified Gaps and Challenges	74
6	Closing Gaps & Overcoming Challenges	81
	6.1 Methodology \ldots	. 81
	6.2 Bridging Gap 1: Incomplete Student Skills	. 81

xiii

	6.3 Bridging Gap 2: Unmet Employability Goal	82	
	6.4 Bridging Gap 3: Improperly Implemented Teaching Methods	84	
	6.5 Bridging Gap 4: Missing Curriculum Evaluation	85	
	6.6 Bridging Gap 5: Inadequate Teacher Skills	87	
	6.7 Guidance Notes	88	
7	Discussion of Results	93	
	7.1 Comparing the Identified Gaps & Challenges	94	
	7.2 Comparing the Proposed Solutions	95	
8	Conclusion and Future Work	97	
	8.1 Future Work	99	
List of Figures			
List of Tables			
Acronyms			
B	Bibliography		
W	Web References		
A	Questions of the semi-structured interview A.1 Teachers A.2 Students	123 123 125	
в	Course Description of "Advanced Data Structures and Algorithms"	127	
С	Feedback Form	129	
D	Guidance Notes	131	

CHAPTER

Introduction

1.1 Problem Statement

According to the United Nations (UN), more than 736 million people (10% of the world's population) were living below the international poverty line (\$1.90 per day, [1]) in 2015. People living below the international poverty line are considered to live in extreme poverty, which means that they cannot fulfill their basic needs in certain areas of their life, such as health, education, water, and sanitation [2]. The percentage of the world's population living in extreme poverty dropped to 9.2% in 2017 and was expected to drop further to 7.5% in 2021. This estimation has been revised due to the global impact of COVID-19 [3]. Therefore, the extreme poverty rate is expected to rise again up to 9.4% by 2021, after a trend of decline since 1990 [4].

On September 25, 2015, the general assembly of the UN adopted the resolution "Transforming our world: the 2030 Agenda for Sustainable Development". This agenda consists of 17 different "Sustainable Development Goals (SDGs)", which are representing "a universal call to action to end poverty, protect the planet and improve the lives and prospects of everyone, everywhere" [5]. In order to measure the progress of achieving these goals, the UN is using the Multidimensional Poverty Index (MPI). The MPI complements the international poverty line in a way that it includes the deprivations people in extreme poverty are facing and therefore monitors poverty in different forms [6].

Cremin & Nakabugo describe that sustainable economic development needs social as well as human development. Furthermore, social and human development depends on "universal education with equitable access and assured quality" [7]. Education reduces poverty and its deprivations as well as the inequality between the rich and the poor [8, 9]. The SDGs are considering this concern by SDG 4: "Ensure inclusive and equitable

1. INTRODUCTION

quality education and promote lifelong learning opportunities for all" [5]. The World Bank Group highlighted that people with a solid higher (post-secondary) education are required for a country's ability to innovate and to sustain a healthy growth over the long-term [10]. According to the Global MPI Report 2019, India reduced its poverty headcount ratio from 55.1% in 2005/2006 to 27.9% in 2015/2016 [11]. Even though "poverty reduction in rural areas outpaced that in urban areas", rural areas are poorer than urban areas [11]. Additionally, although poverty is decreasing in India, the inequality indicated by consumption, income, and wealth increases [12].

Due to the increasing demand for technically skilled workers as well as the growing employment opportunities in the IT sector in India since the 1980s, Higher Education Institutions (HEIs) in India responded by increasing the number of graduates in engineering disciplines. In order to cope with this growth while decreasing the investments from the government into the Higher Education (HE) sector, several state governments and the government of India urged the private sector to establish self-financed, engineering HEI [13, 14]. These institutions are meant to educate people to become engineers that meet the demands of the industry.

Although the number of engineering graduates exponentially grew from 1998 to 2008, the quality of education students received was insufficient, leading to a mismatch between the needs of the industry and the skills of the graduates [13, 14]. A survey of the World Bank confirmed this mismatch and pointed out that "64% of employers are only somewhat satisfied or worse with the quality of engineering graduates' skills" [15]. This skill mismatch is still observable. The National Employability Report by Aspiring Minds examined a sample of "more than 170,000 engineering students from more than 750 engineering colleges across multiple Indian states which graduated in 2018". According to this report, of the sample group, 52.5% are not able to write functionally correct code and 37.7% are not able to write compilable code. Additionally, they found out that candidates of colleges located in lower populated cities are overall less employable than the candidates from colleges of higher populated cities, regardless of the examined job role [16]. According to the definition of Indian urban and rural areas [17], the lowest-populated-cities bracket in the report of Aspiring Minds includes rural as well as urban areas with a population of up to 0.5 million people [16]. Students in rural areas are facing problems and limitations such as:

- limited access to education [18, 19]
- lower quality of education in general [19]
- insufficient infrastructure or overcrowded classes [20]

According to Aneja, most students in rural areas have a lack of relevant skills to take part in HE courses from the beginning, some students are forced by "the pressure of their parents" to gain HE admission [20]. Additionally, rural HEIs in poorer regions find it difficult to catch up with the adaptability of their developed counterparts, which are able to change their courses and syllabi according to the latest developments of their respective fields. As a result, rural students are falling behind students of HEIs, which are situated in urban India or other, more developed regions [20].

In 2019, more than 90% of the HEIs in India were private, so-called affiliated colleges, which implement curricula designed by nationally recognized state universities, also referred to as affiliating universities in this context [21, 22]. The affiliating universities define curricula, which are taught at affiliated colleges and conduct the examinations of the courses of the curricula. Although the affiliated colleges are eligible to offer additional courses, they are not allowed to adapt the given syllabi of the courses, the curricula or parts of the curricula [23]. The exact implementation of the curricula and the teaching methods used during the courses are decided by the affiliated colleges. The affiliating university assesses the actual implementation of the curricula it has designed on the basis of their standardized examinations at the end of each semester. According to the Annual Report 2018-2019 by the University Grant Commission (UGC), 40.489 such colleges exist in India and 60.53% of these colleges are located in rural areas [21]. According to Dubey et al. one of the reasons for previously mentioned skill mismatch is this huge privatization of the engineering education sector [14].

1.2 Motivation

The author of this thesis studies the Business Informatics master program at the TU Wien [24]. The curriculum of Business Informatics is based on the "Universitätsgesetz 2002" (UG 2002) and according to §1 UG 2002, universities are "responsible for solving problems of humankind as well as contributing to the development of society and natural environment" [25]. Furthermore, the curriculum defines in its qualification profile of their students that the curriculum, among other things, is focusing on the analysis, design, implementation and evaluation of Information Systems and displays them as "socio-technical systems, which are including human and mechanical components" and that "modern Information Systems are playing a central role in nearly every economical, political and social interdependencies. They are a basic requirement for the developing digitalization of the economy and society" [26]. Sufficient Software Engineering (SE) skills are needed for the implementation of such Information Systems. Therefore, successfully teaching sufficient SE skills and competencies enables the development of the digitalization of the economy and society.

Furthermore, the author of this thesis has been involved in the ongoing development of the curricula of the Bachelor's degree program in Business Informatics and the Master's degree programs in Business Informatics and Data Science for several years as part of his voluntary activities within the Study Commission for Business Informatics. The author of this thesis was also a member of the Senate of the TU Wien, where he participated in the further development of curricula models ("Musterstudienpläne") and evaluated changes of various curricula of the TU Wien. Furthermore, the author acquired expertise in the field of quality assurance and quality enhancement in higher education by attending further training courses of the Agentur für Qualitätssicherung und Akkreditierung Austria (AQ Austria) on the accreditation and auditing of HEIs in Austria and by visiting the European Quality Assurance Forum [27, 28]. Additionally, the author of this thesis worked as a freelance expert for the AQ Austria in several program accreditations. The author participated in a total of 4 expert reviews of applications for the accreditation of study programs by private universities and universities of applied sciences [29, 30, 31, 32].

The author of this thesis and the thesis itself are part of the TU Wien, which is one of the Austrian universities with the responsibilities previously mentioned. Therefore, from the author's point of view, it is also up to the author and this thesis to fulfill these responsibilities. Additionally, the knowledge of the author about the development and the quality assurance as well as the quality enhancement of curricula gave the impulse to deal with the objectives of this thesis, which are described in the next section.

1.3 Objectives

The UN formulated specific targets for each SDG, such as target 4.4 of SDG 4: "By 2030, substantially increase the number of youth and adults who have relevant skills, including technical and vocational skills, for employment, decent jobs and entrepreneurship." As already mentioned in Section 1.1, the skills of engineering graduates do not match the needs of the industry in India. According to Dubey et al. this mismatch results from the "de-facto mass privatization of the engineering education", but the reasons why this "de-facto mass privatization" leads to the mismatch are not known [14]. As it also described in Section 1.1, a key characteristic of the private higher education sector is that it is mainly composed of affiliated colleges, which are responsible for implementing a curriculum that was designed by their affiliating university. However, the efficacy of the curriculum implemented by the affiliated college is evaluated only with the help of standardized tests, which in turn are based on the affiliating university's curriculum design.

This thesis aspires to contribute to target 4.4 of SDG 4 by providing insights into underlying problems of higher SE education in rural India. Therefore, the specific goal of this thesis is to examine the previously described key characteristic of the private higher education sector in the context of the private higher SE education sector in rural India. With the help of the resulting findings, solutions to problems of the private higher SE education sector shall be developed that help to increase the number of SE graduates with sufficient skills.

As it is explained in Section 2.1, the vast majority of study programs that also include the cultivation of SE skills are designated as Computer Science (CS) study programs. Therefore, in order to achieve the specific goal, this thesis takes a closer look at CS curricula that are implemented at Indian HEIs of the private higher education sector in order to determine whether and to what extent gaps exists between the implemented curricula and their intended design and how these gaps correspond to the skill mismatch found in the literature. In case such gaps exist, this thesis aims to propose solutions that can subsequently be applied by Indian HEIs in order to minimize these gaps.

Therefore, this thesis addresses the following research questions (RQ):

- **RQ1**: What gaps exist between a designed Computer Science (CS) curriculum and their implementations at Higher Education Institutions (HEIs) in rural India, exemplified by an affiliated college in rural Andra Pradesh, India?
- **RQ2**: What are appropriate solutions that help HEIs in rural India implementing CS curricula to reduce the identified gaps from **RQ1**?

Besides the list of identified gaps and solution proposals from the answers to **RQ1** and **RQ2**, the results of this thesis include a set of guidance notes that is meant to present the answers to **RQ1** and **RQ2** in a suitable, understandable manner for decision makers of the Indian higher education system. Two different guidance notes are developed, one for those who design a curriculum and another one for those who implement it. However, as this thesis only evaluates the implementation of curricula and not the design, the guidance notes for curriculum designers are intended to help them to design a curriculum in a way that reduces the identified gaps of the implementing HEI.

1.4 Methodology

During the first phase of this thesis, further information about SE with a focus on teaching SE as well as the Indian higher education system and the current status of the SE education in India are gathered via literature research.

The second phase of this thesis is an exploratory case study that examines the curriculum implementation of a partner institution, which is 1 of over 80 rural Indian HEIs that implement the same curriculum designed by a nationally accredited state university of Andra Pradesh. A single-case study based on the definition and characteristics of Yin is chosen as this thesis aims to take a deeper look into the realities of teachers and students in order to stand out from related work (see Section 2.3).

In the third phase of this thesis, the results of the first and second phases are used to propose solutions that aim to bridge the found gaps and therefore answer **RQ2**. The solutions are developed by the author of this thesis and are based on literature that examine problematic characteristics similar to those of the identified gaps. With the help of these solutions, guidance notes are created that aim to support the implementation of a curriculum as a curriculum manager or a curriculum designer.

1.4.1 Case Study Design

According to Yin, case studies such as the one in this thesis provide a profound insight into social phenomena from a holistic, real-world perspective. Furthermore, Yin states that the reason for a single-case study is to collect evidence about the circumstances of everyday situations, so that it can subsequently provide information about the social processes studied [33].

The case study of this thesis consists of semi-structured interviews, observations and a Design-Reality Gap Analysis (DRGA) in order to answer **RQ1** (see Sections 4.1 & 5.1). The semi-structured interviews and observations are thematically analyzed regarding the actual implementation of a CS curriculum at a partner institution as well as realities of students and teachers. Through the DRGA, differences between the design and actual implementation of the CS curriculum are then identified. Following Yin's characteristics of a case study, the DRGA triangulates its findings with the help of interviews, observations and information provided by institutions of the Indian higher education system [33]. The design of this thesis' case study is outlined in Figure 1.1.



Figure 1.1: Design of this thesis' case study

1.4.2 Selecting the partner institution

As it is described in Section 2.3, this thesis aims to take a deeper look into the realities of teachers and students in order to stand out from related work. With the help of these realities, solutions should be proposed that are actually realizable for the people involved.

From the point of view of the author of this thesis, it is an advantage if the statements of teachers and students also correspond to the actual reality, which in turn requires an existing relationship of mutual trust.

For this reason, the author of this thesis collaborates with a college that has already worked with the TU Wien for several years through a variety of on-site and remote collaborations. Due to the already existing collaborations, it was possible to work with teachers and students of the partner institution in advance on another cooperative project, which was not part of this thesis. Thus, an initial relationship of trust could be established. Therefore, this college is chosen as a partner institution for this thesis.

The necessity for mutual trust was also confirmed for the author of this thesis during the interviews. For example, some students asked several times during the interviews if the recordings of the interviews might be seen by the faculty of the college, because they were afraid of negative consequences. After being assured by the author of this thesis that the recordings will not be shared with the faculty of the college, they began to open up and talk more about their problems and challenges. Additionally, those teachers with whom the author of this thesis had already worked in advance were, in the view of the author of this thesis, more open and much more willing to talk about existing shortcomings and problems.

1.5 Limitations

In order to distinguish this thesis from related work and to remain within the scope of a master's thesis, the following limitations are defined:

- 1. This thesis evaluates the implementations of a curriculum in relation to their underlying design. The design of the curriculum and whether it is suitable for increasing the number of CS graduates with sufficient skills is not examined by this thesis.
- 2. The results of this thesis are based on one implementation of one curriculum design. It is outside the scope of this thesis to examine whether a different implementation of the same curriculum would yield the same or similar results. Likewise, it is beyond the scope of this thesis to examine whether implementing a different but comparable curriculum design would yield the same or similar results.
- 3. When studying the implementation of the curriculum by examining the realities of students and teachers, social and cultural problems are not taken into account. Also, it is outside the scope of this thesis if labor policies such as employment protection, paid sick leave or unemployment benefits exist.
- 4. While this thesis identifies the gaps between the curriculum design and its implementations, it does not examine the root causes of the identified gaps and challenges.

- 5. The solutions developed for the identified gaps in this thesis are based on relevant literature and take the reality of the affected people into account. Nevertheless, it is beyond the scope of this work to evaluate for each solution whether it is actually suitable for closing the respective gap.
- 6. This thesis is developed from an external, European point of view. Furthermore, as the author of this thesis could not be on site due to a 2-year long international flight ban from March 2020 to March 2022, the collaboration took place exclusively remotely [34].

1.6 Organizational Issues

Due to the limited infrastructural possibilities in rural India, students and teachers often struggled with connection problems, which meant that meetings were often interrupted or cancelled. Difficulties also arose again and again in collaboration, for example, it often took several days for relevant information to reach the author of this thesis, or it was often only given on request. Likewise, it was rarely possible to get fixed plans for periods that lie more than 1-2 months in the future, which meant that it was usually only possible to work with provisional specifications. The partner institution also acted very dynamically in many aspects, so that sometimes exams suddenly took place over several weeks, and the students' time was then fully occupied.

The most serious organizational problem caused the failure of an experiment originally planned to find a suitable teaching method considering the local realities, using an existing mandatory course of the curriculum and change the teaching method in use. Therefore, this was removed from the methodology of this thesis.

8

$_{\rm CHAPTER} \, 2 \, [$

Related Work

This chapter outlines the current status of higher SE education in Indian. The problems and challenges of higher (software) engineering education as well as potential solutions to solve them are presented from the perspective of published literature.

As already described in detail in the problem statement of this thesis, India has been facing for decades the challenge that engineering graduates largely do not meet the needs of the industry (see Section 1.1). The National Employability Report by Aspiring Minds, which examined a sample of "more than 170,000 engineering students from more than 750 engineering colleges across multiple Indian states which graduated in 2018", on the one hand, gives a broad overview of the trend over the last years as well as the actual state of engineering student's employability and, on the other hand, analyzes current issues of the Indian HE system regarding SE education and formulates recommendations for the respective stakeholders [16].

Furthermore, there is evidence in the literature that India is not the only country to have this problem. One of Duell's recent works in this field gives a brief insight into the issues and challenges of the whole educational system itself, regarding the mismatch between the industry's needs and the skills as well as the employability of students, which are graduating from Indian, Indonesian and Thai universities [35]. Lee & Cheng published a field report about changing SE education in Taiwan, where they describe a very similar initial situation as it is written by the report of Aspiring Minds. They have successfully designed and implemented a module-oriented SE curriculum, which led to an increased number of offered courses as well as course participants [36].

2.1 Software Engineering Education in India

In India, the most popular engineering programs are Civil Engineering, Electrical Engineering, Mechanical Engineering as well as Computer Science & Engineering, which are mostly graduated with the academic degrees Bachelor of Technology (BTech) or Bachelor of Engineering (BE) [37]. Although the author of this thesis has not found official statistics from a national body or equivalent, there are websites of private providers, which list numerous study programs all over India. For example, at the time of the examination, "CollegeDekho" listed a total of 202314 courses at 20020 HEIs [38]. Limited to HEIs offering a BTech (filtering by BE was not possible) in CS, 2714 HEIs were listed, which in turn offer 3448 programs in this subject area [39]. In relation to that, filtering by SE, "CollegeDekho" listed only 14 HEIs that implement 16 corresponding programs [40]. In Software Development (SD), there were only 2 programs at 2 HEIs [41]. A similar situation was also found on a comparable website. The website "collegedunia" lists a total of 18317 HEIs [42] located in India. Filtering the list of HEIs by those that offer "BTech/BE" programs in CS, 3809 HEIs were listed [43]. For SE programs only 37 HEIs were listed and SD was not even available as a value for the respective filter [44]. Therefore, a CS and not a SE study program is examined in the context of this thesis.

Garg & Varma have published three different papers regarding SE Education at Indian universities in the last 12 years [45, 46, 47]. They are one of the few people, who have specifically done research regarding the topic of this thesis. They addressed India specific "people issues" (issues related to HE stakeholders such as teachers and students) in SE education and training as well as India specific issues and challenges in general for SE education, from the industry and the academic perspective [45, 46].

2.2 Higher Education Challenges of India

One of the main challenges of the higher education system in India is the number of graduates as well as the quality of education the graduates received [48]. The quantitative lack of engineering graduates led to a huge increase in the number of private engineering HEIs [14, 15]. Specifically in the Indian higher engineering education, the skills of graduates do not match the industry requirements, according to various studies over the last 15 years [14, 15, 49, 50, 51]. Industry requires graduates that do not need additional training [45, 50]

The following is a summarized overview of skills of (software) engineering graduates that are categorized in the literature as missing or insufficient in the context of the analysis of the skill mismatch:

- Communication skills (oral & written) [15, 46, 50, 52, 53]
- Ability to analyze and solve problems [15, 46, 50, 52]
- Collaboration, Teamwork and Social skills [15, 50, 52, 53]
- Creativity [15, 53]
- Reliability [15, 50]

- Self-motivated [15, 50]
- Willingness to learn, understand/take new directions [15, 46, 53]
- Technical skills [15, 53]
- Use of modern (software) tools [15, 52]
- Management skills [53]

The reasons for the skill mismatch are discussed in the literature and include:

- 1. Insufficient qualifications of the teaching faculty members [14, 49, 50]
- 2. Rigid curricula that cannot respond to changing needs of local industry [45, 50, 51]
- 3. Most of the teaching is focused on the theoretical aspects, which means that the practical applicability of what is learned is pushed into the background [45, 49, 51]
- 4. The extended usage of traditional methods such as memory based assessments instead of using project based assessments [45, 49, 50, 51]
- 5. Insufficient or non-existent quality assurance or quality management systems at the HEIs [49, 50]
- 6. Lack of sufficient infrastructure or research facilities [45, 49, 50]
- 7. Lack of cooperation and connections of the HEIs with the industry [49, 50, 51]

Dubey et al. attribute these reasons for the skill mismatch to the increasing privatization of the engineering education sector mentioned above [14].

2.2.1 Resolve the skill mismatch

Similar to the research regarding the challenges in India's higher engineering education, improvements to cope with described skill mismatch have been discussed at least over the last 15 years [15, 46, 49, 50, 51].

Mahanti et al. suggest teaching theories and models of the field with a connection to practices to strengthen students' understanding as well as to teach the applicability of theoretical knowledge. Additionally, when using project-based learning as part of a SE course, it should be ensured that practical projects do not only focus on on the Building part of the Software Development Life Cycle (SDLC) but also on the other parts such as Defining, Designing and Testing. Furthermore, group projects should be conducted in order to improve the teamwork skills of students. Additionally, they mention that the limited time a teacher can spend preparing their lessons leads teachers to reuse old materials and use traditional teaching methods. In order to overcome this problem,

2. Related Work

characteristics of a course such as the contents, the used teaching approaches and the provided study material should be evaluated periodically with the help and feedback of the students. The respective course should then be adapted with the help of the feedback and revised or new SE knowledge as well as requirements by the industry. One way of providing new SE knowledge or industry requirements would be courses taught by the industry. In addition to existing courses taught by faculty members, guest speakers from the industry should give lectures or seminars to students and faculty on technologies currently being used in the industry or other emerging trends in their respective fields [51].

Based on the findings of Mahanti et al., the work of Garg & Varma covers most of the above listed challenges of the Indian HE sector. Garg & Varma suggest that SE programs should consist of basic courses that have both basic SE knowledge and problem-solving skills related to the particular course as course objectives, which is similar to the structure of the knowledge areas discussed the guidelines for CS and SE curricula by Association for Computing Machinery (ACM) & Institute of Electrical and Electronic Engineers (IEEE) Society [54, 55]. However, Garg & Varma note that that these restructured programs would require different teaching approaches and assessment methods as currently used, so that the application of taught knowledge and skills is assessed. Regarding the cooperation between HEIs and industry, Garg & Varma suggest to cooperate via guest lectures and internships for teachers. Through these internships teachers would be able to teach SE knowledge more effectively with the help of their hands-on experiences from the industry [46].

The recommendations of Bloom et al. focus on the assessment methods and teaching approaches used at engineering HEIs. They suggest concentrating more on understanding the course objectives rather than memorizing them. This can be accomplished by using teaching approaches that encourage students to be active learners rather than just listening to lectures, taking notes, and then memorizing course content. These teaching approaches should enable the students to identify, analyze and solve a problem in their area of interest. As a further consequence, HEIs need to adopt assessment methods that are appropriate for assessing these "higher-ordering thinking skills" rather than the ability to memorize course content [15].

Mann et al. focus on the challenge regarding the lack of collaboration between HEIs and the industry. All but one of their recommendations propose structural industry participation in India's HE sector or the HEIs themselves. These recommendations include to establish an Indian-wide committee with academic and industry experts in order to periodically update the curricula across India according to the industry requirements and the mandatory involvement of the industry in the governance of an HEI to ensure that its infrastructure meets the requirements by the industry. Furthermore, they suggest to hold additional training programs for students and faculty members in cooperation with the industry. At last, they recommend to increase the incentives for faculty members to improve their quality [50].

Similar to Mann et al. and Garg & Varma, Madheswari et al. state that traditional teaching practices are very teacher-centered and that these practices still in use should

be replaced by teaching practices, which are focusing on the students in order to meet the current demands of the industry. They have created 4 recommendations on how to improve and make technical education more robust [49]:

- 1. "Enhance the teaching-learning process" by revising teaching practices in use, shifting to more student-centered practices and outcome based education.
- 2. "Up-skilling the faculty members" by training the respective members, offering certification courses, include the industry via specialization courses. Madheswari et al. state that training and skill development of the faculty members should be the "first priority to enhance the quality of engineering education".
- 3. "Enhancing students' attitude and participatin" by enhancing the already mentioned teaching-learning process (Recommendation 1), using more project-based learning, offering orientation program as well as specialization courses and create awareness of the job market situations.
- 4. "Upgrading the curriculum and facilities" by changing the scope of the curricula to introduce general skills development, to use research and project based learning, to foster innovation and creativity of the students as well as periodically revise and update syllabi and curricula to address changing needs.

2.3 Contributions of this thesis

Although there are various recommendations, the problem of (software) engineering graduates with insufficient skills still exists according to the already mentioned report by Aspiring Minds [16]. In this regard, no related work could be found that addresses the question of why the skill mismatch still exists, even though it has been studied for more than 10 years and recommendations for reducing the skill mismatch are available. From the author's point of view, there are several possible reasons for this, such as that the majority of the HE stakeholders are not aware of the recommendations, that the recommendations are known but are intentionally not taken into account or that the recommendations are not applicable.

Furthermore, no related work was found that evaluates curricula designs or implementations of Indian HEIs or suggest specific solutions to existing problems. It is this shortcoming that this thesis addresses: solutions are proposed that also take the actual realities of those impacted into account, in order to increase their applicability and realizability. In addition, these proposed solutions are summarized in short, concise guidance notes so that results of this thesis can also be easily redistributed and thus be accessible by stakeholders from many HEIs in rural India.



CHAPTER 3

Fundamentals

The fundamentals describe what is generally defined as SE knowledge and which skills are relevant at the level of a beginner in the field of SE. Furthermore, it explains how SE is primarily taught and what the characteristics of this teaching method are. It also introduces the Indian higher education system and explains its structure. In addition, the structures for quality assurance in the Indian higher education system and their quality criteria for HEIs are described.

3.1 Software Engineering Knowledge

The following section outlines a definition about which knowledge is contained in the domain of SE and how it can be divided into different knowledge areas. Furthermore, the individual knowledge areas as well as the respective basic skills are examined.

3.1.1 Knowledge areas

Bourque & Fairly list 15 knowledge areas in their latest Software Engineering Body of Knowledge (SWEBOK) guide. These areas cover different aspects of the SE domain which consider the SDLC such as requirements, design, testing, maintenance and further more. Every knowledge area is composed of "generally accepted knowledge" that every competent software engineer "should be equipped with this knowledge for potential application" [56].

As Sedelmaier & Landes mentioned, the SWEBOK guide defines its knowledge areas on a more abstract level of expertise, but doesn't list clear definitions of how to derive precise teaching goals or targets from the areas [57].

3.1.2 Sufficient skills

According to the Merriam-Webster dictionary, the noun "skill" is defined as "the ability to do something that comes from training, experience, or practice" [58]. The Cambridge Dictionary defines as "skill" as "an ability to do an activity or job well, especially because you have practised it" [59].

Sedelmaier & Landes criticize that there is no concrete definition of the correct way to derive learning objectives from the abstract knowledge areas on which both the SWEBOK guide and the ACM model curriculum are based on. As Sedelmaier & Landes work has been published in 2014, they refer to the second version of the SWEBOK guide by Bourque & Dupius [57]. In the opinion of the author of this thesis, this also applies to the third version of the SWEBOK guide.

Nevertheless, the IEEE society published the Software Engineering Competency Model (SWECOM), which is based on various resources of the IEEE Society such as the third version of the SWEBOK [56] and the standard curriculum of SE [55]. The SWECOM "describes competencies for software engineers who participate in developing and modifying software-intensive systems" [60, p. v] and is composed of five different elements: cognitive skills, behavioral attributes and skills, requisite knowledge, related disciplines and technical skills [60, p. 5]. The latter one of these elements is examined in the following section of this thesis.

Technical Skills according to SWECOM [60]

The technical skills are the main concern of the SWECOM and are categorized into two different skill areas: life cycle skill areas and crosscutting skill areas, whereby each skill area lists concrete skills and each of these concrete skills is broken-down into activities. Furthermore, each activity is specified for five different levels of competency: Technician, Entry level Practitioner, Practitioner, Technical Leader and Senior Software Engineer [60, pp. 8-12]. For example, the technician is able to "follow instructions" and the Entry level Practitioner is able to "assist in performance of an activity or performs an activity with supervision" [60, p. 12]. The technician level is equivalent to a two-year U.S. associate's degree, and the practitioner level is equivalent to an ABET-accredited SE degree program [60, pp. 12-14, 61]. Nevertheless, the SWECOM acknowledges that the skills it lists are not intended to be comprehensive, but rather to highlight the skills and activities they consider most important for a software engineer of the respective level working in this domain [60, p. 74].

Software Requirements Skills (life cycle skill area) This area covers activities that discover, analyse and manage the functional & non-functional attributes as well as the whole requirements engineering process of a software project [60, p. 25]. This area includes skill sets in requirements elicitation, analysis, specification, verification and management [60, pp. 27-30].

Software Design Skills (life cycle skill area) Within this area are activities that "develop and describe the software architecture of a system based on its software requirements" [60, p. 31] and includes skills in software (architectural) design strategies & methods as well as quality analysis & evaluation [60, pp. 32-39].

Software Construction Skills (life cycle skill area) The scope of this area includes activities that are used to transform the specified design into a concrete, functional software according to the requirements [60, p. 41]. Skill sets of this area are managing software construction, coding, debugging & testing as well as integrating & collaborating [60, p. 44-48].

Software Testing Skills (life cycle skill area) This area covers activities that are used to assess the quality of a software product by analyzing the expected and actual behaviour of it [60, p. 49] and the skills which are required to do so are test planning, setting up test infrastructure, use test techniques and test measuring [60, pp. 51-56].

Software Sustainment Skills (life cycle skill area) According to SWECOM, this area includes, but is not limited to, activities related to maintenance, documentation, deployment, operation or retirement of software [60, p. 57]. They lists Software Transition, Support and Maintenance as the 3 skill sets of this area, which include entry-level activities such as assisting in the development of support material for operations, conducting retirement processes or identifying existing problems [60, pp. 59-65].

Software Process and Life Cycle Skills (crosscutting skill area) In this area, the SDLC is the focus, therefore this area includes, among others, those activities that deal with the definition, implementation, measurement or improvement of SDLC processes [60, p. 67]. Accordingly, activities such as the execution of defined SDLC processes, the interpretation of process responsibilities and the implementation of process improvements are listed by them [60, pp. 68-72].

Software Systems Engineering Skills (crosscutting skill area) This area deals with systems where software is one of many components but plays a critical role [60, p. 73]. Since this is where System Life Cycle processes and SDLC processes meet, this area primarily includes activities that address the integration of SE into Systems Engineering development models and enabling collaboration between SD and Systems Engineering teams along the System Life Cycle [62, 60, pp. 75-88].

Software Quality Skills (crosscutting skill area) This area includes both quality assurance and quality control of software [60, p. 89]. Their skills and activities developing "quality products" with the help of appropriate tools according to defined quality standards and models. Participation and collaboration in reviews and audits are also mentioned [60, pp. 92-100].

Software Security Skills (crosscutting skill area) The software security area covers the entire SDLC and includes activities such as developing software based on predefined recommendations and guidelines (similar to software quality skills) to avoid potential vulnerabilities or identifying potential risks through code reviews [60, pp. 101-106].

Software Safety Skills (crosscutting skill area) For safety-critical systems, i.e. those that can have a direct impact on "human life, health, integrity or the environment", the quality of the software is of crucial importance [60, p. 107]. Since this area also covers the entire SDLC, identifying potential risks and hazards and formulating safety requirements based on them are among the relevant skills and activities. The application of appropriate guidelines in the development and testing of such a system is also mentioned [60, pp. 109-114].

Software Configuration Management Skills (crosscutting skill area) SWE-COM describes this according to its definitions in the IEEE Standard 828-2012 and the SWEBOK guide [63]. According to the latter, Software Configuration Management consists of the identification, control, status accounting and auditing of the software configuration as well as the release management and delivery of software [60, p. 115]. Therefore, skills in this area are, for example, the identification of software configuration items, the implementation of changes to the them or the building of software releases [60, pp. 117-122].

Software Measurement Skills (crosscutting skill area) Since quantifiable data obtained through measurement is used to evaluate processes, this area also affects all other areas of the SWECOM [60, p. 123]. Essentially, this area deals with the planning and execution of measurement processes. Thus, the collection of the corresponding data but also the identification of possible improvements of the measurement process in the skills and activities are mentioned. It is noticeable that independent activities in the area of planning are only carried out from the practitioner level onward [60, pp. 124-127].

Human-Computer Interaction Skills (crosscutting skill area) Due to the fact that the user interface often makes the difference between poorly usable or even unusable software and successful software, this area is mentioned separately despite its close similarity to other areas and affects the requirements elicitation, design and evaluation processes [60, p. 129]. Its skills and activities include, but are not limited to, identifying target users and their accessibility requirements, document use cases and interaction dialogues, analyzing a design according to an usability checklist as well as conducting usability tests [60, pp. 131-139].

3.2 Teaching Software Engineering

The following section discusses the requirements of learning environments for SE education and teaching approaches used in SE education, in particular project-based learning.

3.2.1 Learning Environments

Garg & Varma discuss the requirements of a learning environment for SE education as well as pointing out the challenges and limits of teachers and students in SE. According to them, learning objects, systemic- and climate requirements are the 3 types of requirements of a SE learning environment [47]. Learning objectives define the knowledge and skills which should be the outcome of a learning process. The CS Curriculum guideline defines different learning objects for each knowledge area. Systemic requirements are seen as facilitators of the "design, development, operationalization and usage of the learning environment in short and long term" and influence the teaching approaches in use [47]. Systemic requirements interact with or influence each other as well as learning objectives, for example: the ability to scale a learning environment with the number of student enrollments is influenced by the resource efficiency and vice versa. According to Garg & Varma, climatic requirements are reflected in the quality of teaching [47]. For example, the use of practical problems to achieve a learning objective is among these requirements.

Garg & Varma identified 10 systemic requirements. Of those 10 systemic requirements, 4 have been discussed in the context of an Indian SE education scenario:

Need for scalability As a result of the few number of institutes, the lack of experienced faculties and infrastructure, the classes tend to contain up to 200 students, which adds numerous challenges for teachers, for example a bigger administrative overhead or the impossibility of implementing more suitable teaching and learning practices such as active learning or personalized learning. As a result, the typical course is limited to theoretical aspects of SE, but they are neither discussed nor practiced.

Therefore, without any adaptions to handle this effect, a large class size reduces the effectiveness of the courses.

Motivation of students According to Garg & Varma the typical computer science students tend to favor direct programming without using SE processes such as testing or documenting in order to get quick results. Therefore, Indian students do not get motivated enough to learn or practice SE on their own or visiting respective course. As a consequence, teachers also prefer teaching a programming course instead of a SE course. Furthermore, through the need for the ability to work with higher levels of abstraction, especially undergraduates are more difficult to teach because of their lack of computing experience, the ability to deal with ambiguity and interpersonal skills. [64]

Need for for flexibility Because of the fast emergence of new concepts, paradigms and methods in SE, the respective curricula at Indian universities are evolving and therefore, courses and training in SE should be able to adapt to the evolving curricula.

Need for resource efficiency The Indian education system usually depends on limited infrastructure, where there only exist few resources for authentic projects or lab facilities. Furthermore, there is only limited availability of qualified and professionally experienced SE teachers and therefore, many teachers teach multiple subjects in 20+ hours per week, which require significant preparation time and, as a consequence, reduces the available time for unpopular subjects such as SE.

3.2.2 Teaching approaches

Teaching methods can be categorised as student-centred and teacher-centred [65, 66, 67]. Serin summarises teacher-centered approaches in his literature review as follows:

"In teacher-centered classrooms, the teacher is in charge of learning; therefore, he/she transmits knowledge to the students. As the teacher holds the ultimate authority, the students do not collaborate. The content is decided and the learning tasks are structured by the teacher. The instruction is delivered through lecturing and provision of feedback and correct answers are widely used. The teacher is the primary source of information and the textbook is the center of activities" [65].

In contrast to this are student-centred approaches, where students have an active role and work cooperatively with other students to learn together [65]. A teaching method with these characteristics that is most commonly used in the field of SE is project-based learning [68].

Project-based learning

Various teaching approaches for SE or SD education are discussed in the literature [69, 70, 71, 72, 73, 74, 75]. According to Cico et al., who conducted a systematic mapping study of 126 papers regarding SE trends, 82.5% of the used teaching approaches in SE education are Project-based learning (PBL) approaches [68].

Wolpert defines the PBL approach as follows:

"PBL is the ongoing act of learning about different subjects simultaneously. This is achieved by guiding students to identify, through research, a real-world problem, local to global, developing its solution using evidence to support the claim, and offering the solution using a multimedia approach to presentation using skills based in a 21st-century set of tools. Kids show what they learn as they journey through the unit, interact with its lessons, collaborate with each

20

other, and assess themselves and each other. They don't just take a test or produce a product at the end to show their learning" [76].

A shorter definition is given by Valenzuela-Valdés & Fernández, who deal with PBL in the engineering field. They describe PBL as:

"A learner focused educational approach where the student extends previous knowledge to new problems through self-directed reflection, research and practice in solving a problem" [77].

According to them, the PBL approach possesses 8 main characteristics [77]. These are as follows:

- Students are responsible for their own learning.
- Students routinely encounter the problem in their daily lives and can research it in more detail for free.
- Collaboration between students is a fundamental element.
- The self-directed learning of students is applicable to the problem.
- Lessons-learned and the concepts and principles used during the solving of the problem need to be analyzed afterwards.
- Conduct self and peer assessments after solving a problem and at the end of a course.
- The actions performed in PBL must correspond to those valued in the real world.

Project-based learning in SE education

There is already published work in the literature that investigate the application of the PBL method to SE courses [78, 79, 80, 81, 82, 83, 68].

In the experimental study of Souza et al., both the PBL approach and a teacher-centered learning approach were used and compared in introductory courses over the span of 2 years. They studied the students' perception of a practical project as a learning tool, but not which of the two approaches is more suitable as an educational method. According to their results, the group of students taught with the PBL approach positively perceived the elaboration of a project as a learning task. In addition, students of this group recognized the importance of engaging in practical applicability rather than being limited to only theory-based aspects. This was not the case for the group of students who were taught using the traditional teacher-centered learning approach. Students of this group expressed more negative aspects, both about a project as a learning task and about the processes used to document and develop the project [82].

Also applied to introductory courses, Thevathayan's work describes how the application of a PBL approach can address the different levels of prior knowledge of entry-level students, in order to provide a delight- and useful project experience for students with and without limited programming backgrounds. The approach was essentially to allow students to choose the projects so that those with more knowledge were given more complex tasks to achieve better grades. At the same time, it ensured that all students developed the core design skills. [83].

Peréz & Rubio studied small, heterogeneous teams of 9 students, who were in turn divided into 3 teams of equal size, always consisting of 3 roles: Team Leader, Editor, and Proofreader. In this way, the project was also divided into 3 sub-projects, each of which was then worked on by a team. Due to the regular rotation of roles, all students had different tasks to fulfill, which means that they were regularly faced with new challenges. They found, first, that academic performance of students improved and, second, that the students themselves valued their learning experience and the training [81].

Fioravanti et al. analyzed their application of a PBL approach in combination with project management in order to accomplish a more realistic representation of real world software projects. In their experiment, they investigated collaboration between graduate students, who acted as project managers, and a team of 8 to 10 undergraduate students, who acted as software developers. In their experience, the use of a PBL approach achieved positive results. For example, the majority of students found the approach used, the presence of leaders, and the use of real problems with real stakeholders to be positive [79].

Marques et al. examined the effect of "reflexive weekly monitoring" on initial SE project courses, which affected the students in a way that they recognized the importance of coordinated work early in the project and adjusted their behavior for their team's advantage. Furthermore, the students' "coordination, effectiveness, and sense of belonging to a team" improved [80].

A preliminary study by Günay et al. suggests that the PBL approach can increase the critical thinking and problem solving skills of senior-level undergraduate students [78].

3.3 India's Higher Education System at a glance

Universities for higher education already existed in ancient times in Nalanda, Taxila and Vakramsila. The present Indian higher education system emerged in 1823 and replaced the previous education system for the purpose of colonization: in the schools of the new system, "English and European sciences" were to be taught in order to "make the natives of the country thoroughly good English scholars" [84]. In 1948, one year after India became independent from the United Kingdom (UK) [85], the University Education Commission was set up "to report on Indian university education and suggest improvements and extensions". It recommended to form a commission similar to the University Grant Committee of the UK [86]. As a result, the nearly identically-named UGC has been formally established in 1956 by the Indian Government in order to
coordinate, determine and maintain the standards of higher education in India [84]. Today, the Indian higher education system is still administered by the UGC, which is coordinating the educational standards of more than 40000 HEI [21, 22]. Besides the UGC another statutory (but not regulatory) body with similar responsibilities exists. The All India Council for Technical Education (AICTE) is responsible for the "coordinated development of a technical education system throughout the country, the promotion of qualitative improvements of such education in relation to planned quantitative growth, and regulation & proper maintenance of norms and standards in the technical education system and for the matters connected therewith" [87]. However, the Government of India is planning to merge both bodies into a Higher Education Commission of India [88].

3.3.1 Universities

According to the All India Survey on Higher Education (AISHE) Report 2018-2019, 993 universities existed in India in 2019 and 39.67% of the universities were located in rural areas. These universities have been categorized into 7 different types [22]. The following list outlines these different types and also provides the total number of universities of each type:

- Central university (46) These universities are established by a Central Act, which is an act by the Parliament of India [89, p. 4].
- State university (371) These universities are established by a Provincial/State Act, which is an act passed by legislature of a province/state [89, p. 7].
- Open university (16) These universities are specialized on distance education and therefore offer only distance degree programs.
- Private university (304) These universities are established by a Central/State Act with the help of a sponsor.
- Deemed university (124) These institutions are "high-performing" and therefore are declared as "Deemed to be" universities.
- Institute of National Importance (127) These institutions are established by "Act of Parliament" and are declared as such.
- Institute Under State Legislature Act (5)

3.3.2 Colleges

Indian Colleges are institutions "established or maintained by, or admitted to the privileges" of a university. In order to be allowed to provide and run degree programs, such colleges have to be attached to a university or university-level institution, because they are not allowed to award degrees on their own [22]. According to the Annual Report 2018-2019 by the UGC, which is based on data by the AISHE, 41935 such colleges have existed in 2019 in India and 60.53% of the affiliated and constituent collages are located in rural areas [21]. As already mentioned before, these colleges can be divided into 4 different types [21, 22]. The following list describes these different types and also provides the total number of colleges of each type:

- Affiliated colleges (38787) These institutions offer studies and the respective courses of their affiliating university and awarding the respective qualifications, in accordance with the rules and regulations of the affiliating university, i.e.: According to the UGC, state universities are allowed to affiliate colleges [90, 23]. The affiliating university defines the curricula, which are taught at the affiliated college, and conducts the examinations. Although these colleges are eligible to offer additional courses, they are not allowed adapt the given syllabi of the courses, the curricula [23].
- Constituent colleges (1702) In addition to the characteristics of affiliated colleges, constituent colleges are also maintained by a university.
- PG and Off Campus Centres (183) Share the same characteristics with constituent colleges, but in contrast are located outside the main campus of the maintaining university.
- Recognised Centres (1263) These centres are divided further into Study and Regional centres. Study centres are "established and maintained or recognized by the university for the purpose of advising, counseling or for rendering any other assistance required by the students" [22]. Regional centres are coordinating Study Centres in the context of regular/distance education.

Furthermore, 10725 "Stand Alone Institutions" exist, which are mostly located in rural areas (55.9%). These institutions are not affiliated by a university and therefore are not allowed to award university-level degrees. Instead, they run "Diploma level Programmes" [21].

3.3.3 Quality assurance in Higher Education

In India, there are several institutions that are working on quality assurance in higher education. In 1994, the UGC established the National Assessment and Accreditation Council (NAAC), which "conducts assessment and accreditation of HEIs such as colleges, universities or other recognised institutions" [91]. Besides the UGC and the NAAC, the

24

AICTE formed a similar board, the National Board of Accreditation (NBA), "in order to assess the qualitative competence of the programs offered by educational institution from diploma level to post-graduate level in engineering and technology, management, pharmacy, architecture and related disciplines, which are approved by AICTE" [92, p. 1]. In 2013, the NBA became administratively and financially independent from the founding AICTE and specifically evaluates technical HEIS [92, p. 1].

At its 532nd meeting, the UGC adopted the so-called "Quality Mandate", which sets new goals for improving the quality of the HE sector to be achieved by 2022. Among other things, students are to be equipped with essential professional and soft skills and each HEI must be accredited by the NAAC with a certain minimum accreditation score. Various initiatives are intended to help achieve these goals, such as the usage of Information and communications technology (ICT)-based learning tools, a mentoring program for non-accredited HEI, examination reforms and the regular revision of curricula [93].

National Assessment and Accreditation Council (NAAC)

The NAAC assesses and accredits HEIs and departments of HEIs [94]. Every university type, except open universities, as described in Section 3.3.1 and every type of college, except off campus centres, as described in Section 3.3.2 are able to apply for the accreditation & assessment process. In order to be eligible for this process, a HEI must exist for at least 6 years or has recorded two classes of graduates [95].

Furthermore, the NAAC recommends that every HEI should set up an Internal Quality Assurance Cell (IQAC) after its successful accreditation process, in order to improve and maintain efficiency of the HEI, structural conditions as well as equitable access to qualitative study and research programs. The IQAC should be composed of different stakeholders of the HEI such as teachers, management members, locals, students and professionals from the industry. Consequently, they have to dedicate themselves to the tasks of the IQAC [96]. The tasks of the IQAC include:

- Develop and apply benchmarks for the academic and administrative operations of the HEI
- Create and maintain a learner-centered environment that ensures the quality of education and promotes the improvement of faculty members to continuously acquire the knowledge and technology necessary for the participatory teaching and learning processes.
- Gather feedback from students, parents and others stakeholders of the HEI through an institutional, quality-related process.
- Organize workshops regarding quality related topics
- Coordinate quality-related activities and document measures that improved the quality

- Be the focal point at the HEI for quality-related best practices.
- Submit an annual quality assurance report to the NAAC

Accreditation criteria by the NAAC The set of accreditation criteria by the NAAC consists of 7 criteria and each criterion consists of several Key Indicators (KIs) which are weighted differently according to the type of the assessed HEI and the sum of weights of the KIs gives the weightage of the criterion. A complete overview of the criteria and the respective KIs is shown in Table 3.1. In contrast to the NBA, a more elaborate description of the criteria and the KIs by the NAAC has not been found by the author of this thesis.

National Board of Accreditation (NBA)

The major objects of the NBA include the assessment and accreditation of engineering education HEIs and programs as well as the continuous development of standards and parameters used during the respective assessment and accreditation processes [92, pp. 1-2]. The NBA categorizes the HEIs it is responsible for into to two different tiers. Although both tiers "are assessed with a similar set of criteria", the weighting of the criteria differs. Tier I institutions include the different types of universities listed in Section 3.3.1 and Tier II institutions include affiliated colleges as described in Section 3.3.2 [92, pp. 7-8]. Furthermore, only Tier I institutions that are accredited by the NBA are mutually recognized by the other 20 full members of the Washington Accord (WA) such as USA, UK, Japan and others [92, p. 4, 98]. Similar to the Bologna Process, the WA "is a self-governing, autonomous agreement between national organisations (signatories) that provide external accreditation to tertiary educational programs that qualify their graduates for entry into professional engineering practice" [99, 100]. The process of accreditation process used by the NBA is shown in Figure 3.1.

Accreditation criteria by the NBA During the shown accreditation process the respective HEI or study program is assessed with the help of their set of accreditation criteria. The complete set of criteria as well as the weighting at the respective tier is listed in Table 3.3. This set consists of the following criteria, where the criteria no. 1 to 7 are used on the program-level and no. 8 to 10 on the institutional level:

Criterion 1 - Mission, Vision and Program Educational Objectives The first criterion is used to evaluate if the assessing study program has a vision, mission and program educational objectives. The vision and mission do not refer exclusively to the study program, but to the institution in which the study program is implemented. The specific goals of the study program are considered in the Program Educational Objectives (PEOs). The NBA defines visions as "futuristic statement that the institution would like to achieve over a long period of time" and the mission describes the path of development toward the vision. The PEOs of a study program "describe what the graduates are expected to perform and achieve during the first few years after graduation" [101, 102,

		Weight			
Criteria	Key Indicator			Affil./Const.	
		Univer-	Auton. Colleg.	Colleg.	
		sities		UG	PG
	Curriculum Design and Development	50	50	NA	NA
Q · 1	Curricular Planning and Implementa-	NA	NA	20	20
Curricular	tion				
Aspects	Academic Flexibility	50	40	30	30
	Curriculum Enrichment	30	40	30	30
	Feedback System	20	20	20	20
	Student Enrolment and Profile	10	20	40	40
	Catering to Student Diversity	20	30	50	50
Teaching-	Teaching-Learning Process	20	50	50	50
Learning	Teacher Profile and Quality	50	50	60	60
and Evaluation	Evaluation Process and Reforms	40	50	30	30
	Student Performance and Learning	30	50	60	60
	Outcomes				
	Student satisfaction Survey	30	50	60	60
	Promotion of Research and Facilities	20	20	NA	NA
	Resource Mobilization for Research	20	10	15	15
Research	Innovation Ecosystem	30	10	NA	10
Innovations	Research Publications and Awards	100	30	15	25
and Extensions	Consultancy	20	10	NA	NA
	Extension Activities	40	50	60	50
	Collaboration	20	20	20	20
	Physical Facilities	30	30	30	30
Infrastructure	Library as a Learning Resource	20	20	20	20
and Learning	IT Infrastructure	30	30	30	30
nesources	Maintenance of Campus Infrastruc-	20	20	20	20
	ture				
	Student Support	30	30	50	50
Student Support	Student Progression	40	30	30	25
and Progression	Student Participation and Activities	20	30	50	45
Ū	Alumni Engagement	10	10	10	10
	Institutional Vision and Leadership	10	10	10	10
Governance.	Strategy Development and Deploy-	10	10	10	10
Leadership and	ment				
Management	Faculty Empowerment and Strategies	30	30	30	30
	Financial Management and Resource	20	20	20	20
	Mobilization	20	20		
	Internal Quality Assurance System	30	30	30	30
Institutional	Institutional Values and Social Re-	50	50	50	50
Values and	sponsibilities		~~		
Best Practices	Best Practices	30	30	30	30
	Institutional Distinctiveness	20	20	20	20
	Total	1000	1000	1000	1000

27



Figure 3.1: Accreditation process of the NBA [92, p. 13]

TU Bibliotheks Die approbierte gedruckte Originalversion dieser Diplomarbeit ist an der TU Wien Bibliothek verfügbar wien vourknowledge hub The approved original version of this thesis is available in print at TU Wien Bibliothek.

28

		PO 1	PO 2	 PO 12	PSO 1	 PSO 4
(C101.1	1	3	2	-	-
(C101.2	2	-	-	1	3
(C101.5	-	1	3	2	-
(C401.3	1	2	1	-	1

Table 3.2: Course articulation matrix according to an example by the NBA[102, p. 54][101, p. 53]

p. 10]. Furthermore, the NBA defines Program Outcomess (POs), which are necessary objectives for every engineering study program they assess as well as Program Specific Outcomess (PSOs) that are particularly defined for the study program. The POs are described as skills and attributes that a engineering student has to possess after graduation such as the ability to apply the learned mathematics, science and engineering knowledge or the ability to identify, analyse and solve complex problems. [101, 102, pp. 8-9]

Criterion 2 - Program Curriculum and Teaching-Learning Processes According to the second criterion, a study program should describe its structure (course titles, total number of contact hours, etc.), the alignment of the program curriculum along the POs and PSOs and initiatives about possible cooperation between the study program and the industry. Furthermore, the study program should include routines that are used to enhance the students' learning processes as well as the quality of teaching. More specifically, these routines can include ways to use pedagogical methods such as collaborative learning or using real world examples [101, 102, p. 11].

Criterion 3 - Course Outcomes and Program Outcomes For each of the curriculum's courses the study program should outline how the respective course objectives correlate with the POs and PSOs via a course articulation matrix. An example for a course articulation matrix is shown in Table 3.2, where C101.X represents a particular course objective. The number displays the strength of correlation in an interval of $[1,3] \in \mathbb{N}$, where 1 marks a low, 3 a high correlation and "-" no correlation at all.

Criterion 4 - Students' Performance The HEI must ensure that the students' performance is monitored adequately.

Criterion 5 - Faculty Information and Contributions The HEI must ensure to provide a sufficient number of faculty members that teach the students of the study program adequately. Furthermore, the amount of teaching workload of the faculty members should allow them to also participate in activities besides teaching such as professional and curriculum development, mentoring students, training and interaction

No	Criteria	Mark/V	Weightage
INO.	Program Level Criteria	Tier I	Tier II
1	Vision, Mission and Program Educational Objective	50	60
2	Program Curriculum and Teaching - Learning Process	100	120
3	Course Outcomes and Program Outcomes	175	120
4	Students' Performance	100	150
5	Faculty Information and Contributions	200	200
6	Facilities and Technical Support	80	80
7	Continuous Improvement	75	50
	Institution Level Criteria		
8	First Year Academics	50	50
9	Student Support Systems	50	50
10	Governance, Institutional Support and Financial Resources	120	120
	Total	1000	1000

Table 3.3: Set of accreditation criteria and Tier-weighting by the NBA [101, 102, p. 14]

with experts from the industry. Additionally, the study program enables faculty members to participate in Research & Development activities [101, 102, p. 12].

Criterion 6 - Facilities and Technical Support In order to enable the students to achieve the POs and PSOs of the study program, the HEI must provide and maintain sufficient infrastructure such as adequate laboratories, class rooms and meeting rooms [101, 102, p. 12].

Criterion 7 - Continuous Improvement The study program should be improved continuously by analyzing and evaluating the fulfillment of the POs and PSOs and using the resulting insights by revising courses, assessment methods and the curriculum itself. [101, 102, p. 13].

Criterion 8 - First Year Academics The HEI must provide a "first year of graduation study" that consists of courses in "science, mathematics, humanities and general engineering" [101, 102, p. 13].

Criterion 9 - Student Support Systems The HEI establishes systems that provide support for students such as mentoring programs, self-learning facilities, career preparation or offerings of co- or extra-curricular activities [101, 102, p. 13].

Criterion 10 - Governance, Institutional Support and Financial Resources The governance structure of a HEI that includes the distribution of authorities and responsibilities must be defined clearly. Furthermore, a HEI must ensure that the financial resources are sufficient to attain its Mission and PEOs [101, 102, p. 14].

CHAPTER 4

Data Collection & Thematic Analysis

The exploratory part of this thesis uses semi-structured interviews and observations to gather data about a CS curriculum's actual implementation in rural India. The interviews and observations were subsequently analyzed using the thematic analysis method by Braun & Clark [103]. The semi-structured interviews and observations as well as the thematic analysis are discussed in more detail in Section 4.1. The thematic analysis of the gathered data presented in this chapter is then further analyzed using the DRGA in order to answer the research questions of this thesis (see Chapter 5).

4.1 Methodology

In the following, the research methods used to collect the data, which were subsequently used to answer the research questions of this thesis, are described. This part of this thesis has been realized with a partner institution, which is an affiliated college of the Jawaharlal Nehru Technological University Anantapur (JNTUA) and is located in one of the states in the bottom 25 percentile of employability percentage of graduates [16]. Furthermore, the partner institution is 1 of over 80 affiliated colleges that implement the same curriculum designed by a nationally accredited state university. The students are examined by standardized tests developed by the affiliating university.

4.1.1 Semi-structured interviews

The semi-structured interviews were held with teachers working at the partner institution as well as CS students of the partner institution in order to to gather data about the actual implementation of the CS curriculum and the everyday life of teachers or students

Pseudonym	Role	Sex	Experience
Frankie	Teacher	Female	<1 year
Glenn	Teacher	Male	1-3 years
Jordan	Teacher	Female	1-3 years
Morgan	Teacher	Male	>3 years
Riley	Teacher	Male	>3 years
Nuru	Student	Female	4th Semester
Rory	Student	Female	4th Semester
Sam	Student	Male	4th Semester
Taylor	Student	Male	4th Semester

Table 4.1: Overview of the interviewed persons, their role and level of experience

at a HEI located in rural India in general. Therefore, the semi-structured interviews contribute to answering **RQ1** of this thesis.

Because of the significant importance of establishing a personal contact to subjects described by Saunders et al. as well as the exploratory character, the author of this thesis chose semi-structured interviews as an appropriate interview type for this part of the thesis [104, p. 324]. As it is outlined in Table 4.1, five teachers of the partner institution that hold one or more courses as part of the implemented CS curriculum were interviewed. Furthermore, four students of the fourth semester were interviewed to gather data about their everyday life and the challenges students can face in rural India. The interviews took place in a face-to-face setting and were conducted with the help of Zoom [105]. With the consent of the respective interview partner, the semi-structured interviews were recorded and documented through the transcription of the records afterwards. The author of this thesis chose a face-to-face format, since one can adapt the questions, clarifying doubts or make sure, that the questions have been understood correctly [106, p. 124].

Saunders et al. describe different aspects to consider when preparing a semi-structured interview [104, pp. 328-335]:

1. Level of knowledge

"You need to be knowledgeable about the research topic and organizational or situational context in which the interview is to take place" [104, p. 328]. The interviews aim to gather data about the implementation of a prescribed curriculum from the points of view of teachers and students in emerging regions. In order to do so, the interviews have been held with students and teachers from a partner institution, which is located in rural India. Therefore, this thesis examines the partner institution's situational context and its organizational context described in Chapter 2.

2. Level of information supplied to the interviewee In case exploratory interviews are conducted, Saunders et al. describe that a list of interview themes, which is given to a participants before their interview, may promote validity, reliability as well as the credibility by the participants. This would allow participants to prepare themselves for the interviews whereby they could come up with more detailed information or deliver material which helps at the exploratory study [104, pp. 328-329]. Therefore, the participants of the semistructured interviews have been informed that the interview will cover questions about the CS curriculum in general, approaches and tools which are used when teaching classes as well as the everyday life of a CS student or a teacher at the partner institution.

3. Appropriateness of location

According to Saunders et al. interviews should take place at a location in which the participant feels comfortable and is not disturbed easily [104, p. 329]. The partner institution provided a location which met the recommended requirements as best as the capabilities of the partner institution allowed it. Furthermore, it was ensured by the partner institution that the speed and reliability of the internet connection is a good as the region allows it. Despite this support from the partner institution, some interviews were conducted in the interviewee's home, as the partner institution had partially suspended operations due to measures taken against the Corona pandemic [3].

- 4. Nature of the opening comments to be made when the interview commences According to Saunders et al., "the first few minutes of conversation will have a significant impact on the outcome of the interview", especially if the interviewer and the participant haven't met each other before [104, p. 330]. During the opening of an interview, the interviewer should "demonstrate credibility and friendliness" in order to avoid that a participant will become bored or restive. Furthermore, Saunders et al. list recommendations which can be used at the opening of an interview [104, p. 331]. According to these recommendations, the following points have been conducted at the opening of each interview:
 - "The participant was thanked for considering the request for access and for agreeing to the meeting." [104, p. 332s, Box 10.9]
 - A brief introduction to engineering education in India and the existing problems according to the literature. Furthermore, the usage of the interview results as well as a general outline of the expected results of this thesis was presented to the participant.
 - It has been assured that the results of this interview are going to be anonymized so that nothing said by the participant can be related to them without their consent.
 - The participant got informed, that every question can be skipped and that the interview itself can be stopped at any time if the participants would like to do so.

- The participant is asked to give consent about the video- and audio-recording of the interview. If there was no consent, the respective interview has been noted and only handwritten notes were taken. Furthermore, it has been stated that every recording (if consent about recording has been given) is only accessible for the author as well as the co-supervisor and supervisor of this thesis.
- The participant got informed that the results of the whole study will be publicly available.
- At last, the participant was asked if there are questions about the interview procedure in general and has been informed how long the interview is about to take. Furthermore, it was asked again if the participant is still willing to do the interview and if the interviewer is allowed to record it.
- 5. Approach to questioning

Saunders et al. summarized many findings of the literature regarding the questions of an interview. According to their findings, questions need to be phrased as clearly as possible [104, p. 332]. Additionally, the question was repeated if the author of this thesis had the impression that the interviewee did not clearly understand the question or that the answer did not correspond to the question asked. Furthermore, all of the prepared questions are open questions in order to avoid bias [104, p. 332].

Saunders et al. recommend that the questions are about real-life experiences instead of theoretical, abstract concepts. Therefore, the asked questions aim for the experiences of the participants in the respective topics. Specific questions about problems and challenges during the study program experienced by the participants are classified as sensitive questions by the author of this thesis. According to the recommendation of Healy & Rawlinson, sensitive questions are usually best to ask at the end of an interview, because that allows the participant to gain trust into the interviewer [107].

6. Scope to test understanding

In order to avoid "biased or incomplete interpretation" the interviewer should summarize the given answer of the participant and present the summary to the participant. This would allow the participant to correct falsely interpreted answers and add relevant further content [104, p. 334]. The author of this thesis paraphrased an answer during the interviews if the answer was either very long from the author's point of view or if the author did not understand the answer.

7. Approach to recording data

Conforming to Saunders et al., the interviews have been audio-recorded and notes had been taken by the interviewer. Additionally, for each interview its location, the date and time, if the participant is a teacher or a student as well as the immediate impression of the interview performance by interviewer was noted [104, p.334].

34

8. Cultural differences and biases

Saunders et al. recognise that cultural differences may cause biases to influence the whole interview process and its results [104, p.335]. Therefore, the author of this thesis recognized the importance of the performance of the interviewer as well as the understanding of the questions by the participants, in order to avoid misinterpretations of questions and answers as well as biases due to cultural differences.

9. Performance of the interviewer

According to Saunders et al. recommendation about the appropriateness of the researcher's appearance at the interview, the author of this thesis chose a similar style of clothing as the interview participants [104, p. 330]. Regarding the nature and impact of the interviewer's behaviour during the course of the interview, Saunders et al. describe the appropriate behaviour of an interviewer. The interviewer should maintain an open posture and enjoy or at least appear as someone who enjoys the interview, in order to encourage and appear open to the participant. Furthermore, the interviewer should avoid behaviour which could indicate a bias, for example: give a neutral, but not uninterested response to the answer of participants. For the purpose of behaving appropriately during the interviews, the author of this thesis conducted three test interviews with his co-supervisor and other colleagues. Furthermore, the interviews require to give participants enough time for answering so that they can formulate their response well. Furthermore, the interviewer must ensure, that his or her own views are not projected during the interview [104, p. 334].

Questions

The questions of the interviews are listed in Appendix A and were developed by the author of this thesis. They aim to collect qualitative data about the reality of the CS curriculum as perceived by teachers and students. Their focus is to gain insights about the current state of the implementation of the curriculum. Furthermore, the questions aim to collect experiences from the everyday life at the partner institution of students and teachers. The questions are categorized according to the OPTIMISM model of the DRGA, which is described in Section 5.1.1.

4.1.2 Observations

The observations were conducted in order to gain an insight into the actual implementation of a curriculum and to observe the everyday life of teachers and students from the view of an insider as well as to describe observable phenomena very detailed and to gain insights into unexpected aspects [106, p. 131, 108, p. 332]. Therefore, the observations contribute to answering **RQ1** of this thesis.

According to Sekaran & Bougie, observations include the "watching, recording, analysis and interpretation of behavior, actions or events" [106, p. 130]. The conducted obser-

vations are uncontrolled and participant. The uncontrolled setting is meant to ensure that teachers and students are behaving as naturally as possible and therefore, existing problems are not covered by different behavior of teachers and students. The author of this thesis is going to be an "observer as participant" by observing teachers conducting a course or project review sessions between teachers and students as well as working with groups of students on their final project while revealing the role of the author of this thesis as a researcher from his university [104, p. 294].

It is out of the financial and organizational scope of this thesis, to create a non-participant, controlled observation. Furthermore, the observation setting may enable one to learn and observe the existing problems of teachers from the view of an insider, which is seen as one of the main strengths of participant observation [106, p. 131]. Döring & Bortz, which describe this type of observation as qualitative, confirm, that the scope of this observation type are social interactions between the subjects and their environment. Furthermore, it enables one to describe phenomena very detailed and to gain insights into unexpected aspect, which corresponds to the asked research question [108, p. 332].

Döring & Bortz and Saunders et al. are providing guidelines for observations [108, 104]. Both guidelines were taken into account when planning and conducting the observations for this thesis. The observations took place in three different settings between March and June 2021:

- 1. Teachers were observed teaching a class. This was done by recording a video call in which the teacher was visible and audible in front of the whiteboard throughout the lesson. Both the teaching of a regular course and the teaching of a laboratory course were observed.
- 2. Two groups of 4 students each were observed working on their graduation project. The author of this thesis acted as an assistant teacher. He offered to help the group members when problems occurred in the project and to answer questions that arose during the implementation of the project. To this purpose, regular "office" hours were held and a Slack workspace was set up where the group members could ask questions at any time and also exchange information with each other [109].
- 3. Teachers and all final year students were observed in unregular project review sessions, in which the groups of students reported on the progress of their project and received feedback from the faculty on their progress.

An overview of the conducted observations is presented in Table 4.2. Both, the work with the groups and the project review sessions, were not recorded and the author of this thesis took notes during the observations. Therefore, the majority of these observations are secondary observations [104, p. 296].

Observations	Setting	Scope
А	As Observer	Curricular Course Sessions
В	As Observer	Project Review Sessions
С	As Participant	Assisting Final Project Group 1
D	As Participant	Assisting Final Project Group 2

Table 4.2: Overview of the conducted observations, their setting and scope

4.1.3 Thematic analysis

Although there is no common concrete definition of the term thematic analysis, it is widely used [103]. Braun & Clark characterize thematic analysis as a flexible and easy-to-learn method and is therefore suitable for researchers with no or few experiences in qualitative research. Furthermore, it is described as an advantageous method for research, where one is working participants as collaborators [103]. Since these advantages are consistent with the characteristics of this thesis and its author, the thematic analysis was used to analyze the semi-structured interviews and observations.

Braun & Clark outline a guide for conducting a thematic analysis, which has been taken into account when using the thematic analysis method for this thesis. According to them this method consists of 6 phases [103]:

- 1. Get to know the collected data Transcribe and read the data, write down initial ideas.
- 2. Generate initial codes

Systematically create a coding of the interesting data features of the entire data set and collect the data relevant to each code.

- 3. "Searching for themes" Summarize codes on potential themes and create a collection of relevant data for each of them
- 4. "Reviewing themes"

Review and refine the themes using the coded data extracts firstly and the entire data set secondly, generating a thematic map of the analysis.

- 5. "Defining and naming themes" Explain each theme in more detail, identify the essence of each theme and its contents and generate clear definitions and names for them.
- 6. Create conclusion

Describe meaningful extracts from the themes and provide a concluding analysis of those extracts. Show how the analysis relates to the research questions.

The thematic analysis was carried out from two points of view. Thus, statements were analyzed in terms of how the real implementation of the CS curriculum actually takes place. The statements were also analyzed regarding respective problems and challenges a person or a group of persons expressed. The conclusions drawn from the thematic analysis regarding a student's or teacher's reality were then used to answer **RQ1** with the help of the DRGA (see Chapter 5).

4.1.4 Experiment

It is worth mentioning at this point that originally an experiment was planned in order to find a suitable teaching method that considers the local realities of students and teachers. At first, the course which was planned to take place from March to June. Due to the second Corona wave, which also occurred in the district of Anantapur, i.e. the district where the partner institution is located (51 active cases on 15.03.2021, 17.465 on 16.05.2021 and 1389 on 15.06. in the year of 2021), the partner institution was largely closed for students and lessons were held via distance learning [110]. The author of this thesis was also informed that the period in which the selected course was to take place would be extended. So the students would receive their topic already in March, but the results were expected to be presented in September 2021, whereas the students would have time to implement the project until the end of August 2021.

Both participant groups then elaborated the proposal of their project in mid-April 2021. The proposals consisted of a 1 to 2 page document that briefly summarised the basic topic of their work in an abstract and included a comparison between an existing system or problem and a system or solution they were proposing. After receiving the proposals, the author of this thesis set up a Slack workspace to facilitate remote collaboration between the author of this thesis and the groups, although it took 2 weeks for the groups to fully join [109].

At the beginning of May 2021, the partner institution was closed again due to the increasing number of active Corona cases, which meant that the groups could only work from home. The assigned teachers gave the groups the task of working out the architecture of their system, the basic structure and contents of the required modules and so on by the end of May.

During the review of the groups' elaborations at the end of May 2021, the author of this thesis was informed that the projects should be implemented by the end of June instead of August 2021. As a result, the originally planned, different teaching method could no longer be applied and thus caused the experiment to fail.

4.2 Thematic Analysis of Interviews and Observations

In the following section the conducted interviews and observations are thematically analyzed using the method of Braun & Clark, which is outlined in Section 4.1.3. In contrast to their method, the data were already assigned to the OPTIMISM dimensions in a first step and categorized accordingly. If a statement concerned several of these dimensions, it was assigned to both dimensions. After the generation of the codes, these were summarized to a corresponding theme. Subsequently, a thematic map was created for each dimension. With the help of the thematic maps the themes were refined again. Furthermore, the analyses are enriched with observations and statements made during the interviews.

4.2.1 Objectives and Values

The codes listed in Table 4.3 were generated during the thematic analysis of the interviews and observations with regard to the "Objectives and Values" dimension.

Nr.	Code	Interviews
1	want to become a Software Developer	Nuru, Rory, Sam, Taylor
2	interested into coding / programming languages	Nuru, Rory, Sam, Taylor
3	want more practical knowledge	Nuru, Rory
4	want to go to another country / away from home	Rory, Sam, Taylor
	after graduation	
5	college provides additional courses for students and	Morgan, Glenn, Jordan,
	teachers	Rory
6	additional courses are mandatory for students	Glenn, Nuru
7	additional courses teach software development and	Nuru, Rory, Glenn, Jordan,
	soft skills	Morgan
8	(nearly) 100% of the students pass the exams	Morgan, Riley
9	curriculum updates merged courses that should be	Jordan, Morgan
	separated	
10	merged courses combine conceptual content with	Glenn, Jordan, Morgan
	emerging trends	
11	implementing colleges suggest curriculum update to	Glenn, Jordan
	university	

Table 4.3: Extracted codes regarding the "Objectives and Values" dimension

The themes of this dimension have been elaborated using the codes listed in Table 4.3 and further data extracts from the interviews and observations. Additionally, Table 4.4 lists the Themes of this dimension and indicates the codes on which the respective theme is based. The Thematic Map of this dimension, shown in Figure 4.1, presents an overview of the results of the thematic analysis regarding this dimension.

Theme 1.1 (T1.1) - All of the interviewed students stated that they want to move to a major city or another country and work in the field of software development after graduating. This and the interest in programming or programming languages were the reasons for choosing the CS study program.

- **Theme 1.2 (T1.2)** Besides the study program itself, the participation and completion of extra-curricular courses of potential employers (internships) as well as the placement of students with a company at the end of the study program is seen as a crucial part of the study program.
- **Theme 1.3 (T1.3)** It is of great importance to teachers that all students of the partner institution graduate at the end of their studies. As a teacher, you feel proud when 100% of the students pass the exam at the end of the semester.
- **Theme 1.4 (T1.4)** The curriculum changes almost every two years. The reasons for this include feedback from the HEIs implementing the curriculum as well as emerging trends in CS.

Theme	Codes
T1.1	1, 2, 3, 4
T1.2	5, 6, 7
T1.3	8
T1.4	9, 10, 11

Table 4.4: Mapping of themes to codes regarding the "Objectives and Values" dimension



Figure 4.1: Thematic map of the "Objectives and Values" dimension

During the interviews, neither the goals nor the learning outcomes of the curriculum were mentioned in particular by teachers or students. However, it was stated that the curriculum is changed at intervals of a few, usually 2 years. According to Teachers Morgan, Jordan and Glenn, attempts are now often made to combine fundamentals with emerging trends within a course. The teachers have different opinions on this, with two of them describing this tendency as a weakness of the study program. Teacher Jordan states: "They [affiliating university] have introduced C and Data Structures as one subject [...], which is like they have dumped everything to them [students]". Additionally, when asked about desired changes in the curriculum, Teacher Morgan answers: "Now they [affiliating university] suddenly introduce Python with Data Science. Without knowing Python, how they [students] can understand Python with Data Science?". However, Teacher Glenn wishes for exactly that: "[...] topics like Data Science, Data Architecture, Machine Learning, so these topics I'd like to suggest to add on for the courses [of the curriculum]".

Regarding the values associated with the implementation of the curriculum, it has been noted that some teachers seem to be very proud if 100% of the students pass the exams at the end of the semester. However, students have the feeling that they are mainly studying to pass the exams and not to really understand the acquired knowledge and to be able to apply it practically. For example, when asked what s_he would like to change about the topics of the courses, Student Nuru answered as follows: "We [students] are learning [...] for the sake of exams only. Not learning the subject".

Students stated they are very focused on their own goal, which was to become a software developer or engineer, which is why all of the students interviewed started the CS study program. In this regard, both groups emphasised the importance of internships and placements and that the combination of them with the study program is essential. For example, Teacher Glenn described the strengths of the curriculum as follows: "Here the main advantage for the students was we are providing the additional courses [extra-curricular courses held by companies]".

4.2.2 Processes

The codes listed in Table 4.5 were generated during the thematic analysis of the interviews and observations with regard to the "Processes" dimension.

The themes of this dimension have been elaborated using the codes listed in Table 4.5 and further data extracts from the interviews and observations. Additionally, Table 4.6 lists the Themes of this dimension and indicates the codes on which the respective theme is based. The Thematic Map of this dimension, shown in Figure 4.2, presents an overview of the results of the thematic analysis regarding this dimension.

- **Theme 2.1 (T2.1)** Students are evaluated on the basis of questions defined by the affiliating university. In addition to the main exams by the affiliating university in the middle and at the end of the semester, teachers conduct smaller, additional assignments using the main exam questions on the most recently discussed topic on a weekly basis.
- Theme 2.2 (T2.2) Teachers present the course content and students listen or take notes. In addition, there is one project per semester and extra-curricular courses. Two of the interviewed teachers of curricular courses stated that they try to include real-world examples in order to make their topics more tangible. For example, one of the two teachers tested the Instagram registration page with her students using Selenium [111, 112].

Nr.	Code	Interviews / Observations
1	additional tests are conducted in curricular courses	Frankie, Glenn, Jordan,
		Morgan, Nuru, Riley
2	additional tests do not influence the grade	Glenn, Morgan
3	mandatory projects are conducted each semester	Glenn, Nuru, Rory, Sam
4	teachers mostly stand in front of class and present	Nuru, Sam
	course contents	
5	teachers use blackboard/whiteboard for teaching	Glenn, Jordan, Morgan,
		Riley
6	teachers use PowerPoint slides for teaching	Glenn, Morgan, Riley
7	teachers use videos for teaching	Glenn, Jordan, Morgan,
		Riley
8	teachers use code assignments in practical courses	Sam, Taylor
9	teachers use real life examples in practical courses	Frankie, Jordan
10	students use ready-made solutions to solve assign-	A, B, C, D
	ments/project	
11	teachers provide solutions to assignments	A, C
12	feedback of students is gathered every semester	Morgan, Rory
13	implementing colleges suggest curriculum update to	Glenn, Jordan
	university	

Table 4.5: Extracted codes regarding the "Processes" dimension

- **Theme 2.3 (T2.3)** In both the practical courses and the mandatory projects, students are provided with solutions by their teachers, which students need to edit only slightly, if at all, in order to finish the respective assignment.
- Theme 2.4 (T2.4) A standard procedure for obtaining feedback from students about the teachers of the courses they visit exists. In this process, students rate the respective teacher on several aspects, for example, the "knowledge of the teacher on the subject" or the "Teacher's ability to organize lectures".
- **Theme 2.5 (T2.5)** Teachers can summarize their feedback into a collective complaint and forward it to the affiliating university, which can then be used by the committee responsible for the curriculum to change it.

The actual teaching methodology to be used for one or more courses is not specified in the curriculum; a distinction is only made between regular courses and laboratory courses. The actual teaching methods they use to teach the courses are left up to the teachers themselves. It was found that teachers mostly gave a lecture standing in front of the class to which students listened and asked questions when needed. For example, Teacher Glenn answered the questions about his_her teaching method as follows: "Most of the time we'll [teachers] use blackboard, whiteboard with markers and sometimes we'll

Theme	Codes
T2.1	1, 2
T2.2	3, 4, 5, 6, 7, 8, 9
T2.3	10, 11
T2.4	12
T2.5	13

Table 4.6: Mapping of themes to codes regarding the "Processes" dimension



Figure 4.2: Thematic map of the "Processes" dimension

be switching back with ppts [PowerPoint presentations] also". Sometimes the teacher also involved students with follow-up questions. In Observation A, students wrote down the solutions to assignments of the curriculum presented by the teacher in advance and then entered them into the computer. In addition, students have to work on several projects regularly during their studies. Student Sam confirmed that they have to do a smaller project each semester: "Yes, each semester we [students] have one project. Yearly, we have two minimum projects". In their last semester, students also have to do a graduation project. As described in Section 4.1.2, two groups of students were accompanied by the author of this thesis.

One group of students had to work out a "Hostel Management System" (Observation C), the other a "Payroll Management System" (Observation D). A Google search on 11 January 2022 for "hostel management system project" returned 20.3 million, for "payroll management system" 98.2 million results. These results included links to commercial and non-commercial websites that offer already completed projects in different programming languages and project reports matching the task description of the observed groups [113, 114, 115, 116, 117]. Additionally, on the web platform "scribd", a search on 11 January 2022 for the two project titles yielded around 61,000 documents for the hostel project

and around 232,000 documents for the payroll project [118, 119]. Considering the first 50 results for each project, it is noted that they were very similar to the objectives of the project reports of the groups observed, both in terms of content and structure. Furthermore, almost all of the project reports from this set of results were written by students of Indian HEIs. In Observation D, the students confirmed that the teacher who officially supervises their project has provided them a link to a freely available project that they can take and work on to complete the assignment. As a result, this project has been copied almost completely by the group. They have only tried to create the impression that the project was developed by them specifically for the partner institution by, for example, replacing images on the project's websites.

Regarding the evaluation of students, it was found that students are evaluated solely on the basis of their scores on the mid and end exams. During a semester, two mid exams and one end exam are held. The mid exams account for 30%, the end exams account for 70% of the grade of the respective course. The questions of the end exam are prescribed by the affiliating university. The questions of the mid exams are prepared by the faculty themselves, but in reality they are mostly composed of questions from old end exams. In order to better track the learning progress, the faculty conducts smaller assignments on the individual course units at regular intervals. The questions asked in these assignments are also largely drawn from old end exams. If a student makes too many mistakes on theses assignments, Teacher Morgan stated that they offer remedial classes: "So after class those who are weak students come to faculty and conduct remedial classes to improve the weak students like that".

Furthermore, Teacher Morgan confirms in the interview that they collect feedback about themselves from students of their courses throughout the partner institution: "Every semester we [teachers] are conducting, we are taking feedback from the students on all faculty on all subjects.". This procedure usually takes place before the exams at the end of a semester. A form is used on which questions are asked about the teacher of the respective course (see Appendix C). For example, the students should rate the "knowledge of the teacher on the subject", the "Clarity and understandability of Teacher's explanation" and the "Overall Teaching effectiveness of the teacher" either with "Excellent", "Good" "Fair" or "Poor". In total, the form consists of 13 different questions about the teacher: 4 questions based on the 4-point scale just mentioned, 6 questions based on different 3-point scales, 1 Yes/No question and 2 open questions.

Also, teachers collect their feedback on the curriculum and on individual courses. If problems are identified, they create a collective complaint, which is then sent to the affiliating university that is able to revise the curriculum. According to Teacher Jordan: "We [teachers] can rise it as a collective complaint. Not from only [the partner institution itself, from the collective colleges we can rise this complaint after completion of the semester".

4.2.3 Technology

The codes listed in Table 4.7 were generated during the thematic analysis of the interviews and observations with regard to the "Technology" dimension.

Nr.	Code	Interviews / Observations
1	teachers use blackboard/whiteboard for teaching	Glenn, Jordan, Morgan,
		Riley
2	teachers use PowerPoint slides for teaching	Glenn, Morgan, Riley
3	teachers use videos for teaching	Glenn, Jordan, Morgan,
		Riley
4	practice-oriented courses are conducted in computer-	А
	labs	
5	infrastructure is suitable for teaching	Glenn, Morgan, Riley
6	most students have a smartphone	Nuru, Rory, Sam, Taylor
7	most students do not have a laptop/PC	Nuru, Sam, Taylor
8	infrastructure is insufficient for self-learning	Rory, Taylor

Table 4.7: Extracted codes regarding the "Technology" dimension

The themes of this dimension have been elaborated using the codes listed in Table 4.7 and further data extracts from the interviews and observations. Additionally, Table 4.8 lists the Themes of this dimension and indicates the codes on which the respective theme is based. The Thematic Map of this dimension, shown in Figure 4.3, presents an overview of the results of the thematic analysis regarding this dimension

- **Theme 3.1 (T3.1)** Teachers describe the infrastructure and technologies provided as sufficient: the classrooms are equipped with blackboards or whiteboards and usually also projectors. The partner institution provides one laboratory per study program, which includes computer workstations.
- **Theme 3.2 (T3.2)** The majority of students do not own a laptop or computer, only a smartphone. Furthermore, students state that the infrastructure is insufficient for adequate self-learning.

Theme	Codes
T3.1	1, 2, 3, 4, 5
T3.2	6, 7, 8

Table 4.8: Mapping of themes to codes regarding the "Technology" dimension

For the teachers, it can be stated that in their opinion they have adequate infrastructure for teaching using their teaching methods. For example, Teacher Glenn acknowledged



Figure 4.3: Thematic map of the "Technology" dimension

the suitability of the existing infrastructure for his_her class as follows: "Yeah, we have good infrastructure at college which we can use for our courses".

Students agree that the infrastructure for teaching is sufficient, but they note that there is a big problem regarding the availability of adequate infrastructure for learning outside the classroom. While they have access to the rooms of the partner institution that are not in use and to the hostel, students in the hostel share a room with at least two, sometimes with eleven other students, which often makes concentrated studying impossible (see Section 4.2.6). Furthermore, Student Rory states that the computers available outside of class time are almost always already occupied by others and that only "15% is having laptop and rest of them doesn't have a laptop". According to a survey conducted with students of an extra-curricular course, while 96% own a smartphone only 46% own a computer [120]. According to Student Rory, students without a laptop often ask students with such a device if they can borrow it for a PowerPoint presentation: "when there is any seminar our friends will be asking".

4.2.4 Information

The codes listed in Table 4.9 were generated during the thematic analysis of the interviews and observations with regard to the "Information" dimension.

The themes of this dimension have been elaborated using the codes listed in Table 4.9 and further data extracts from the interviews and observations. Additionally, Table 4.10

Nr.	Code	Interviews / Observations
1	teachers use referenced books of curriculum to pre-	Frankie, Glenn, Jordan, Ri-
	pare course	ley
2	teachers use different internet platforms to pre-	Frankie, Glenn
	pare/conduct courses	
3	students use YouTube for self-learning	Nuru, Sam
4	students use ready-made solutions from internet	B, C, D
	platforms for assignments	

Table 4.9: Extracted codes regarding the "Information" dimension

lists the Themes of this dimension and indicates the codes on which the respective theme is based. The Thematic Map of this dimension, shown in Figure 4.4, presents an overview of the results of the thematic analysis regarding this dimension

- Theme 4.1 (T4.1) Teachers use referenced textbooks that are listed in the curriculum for each course. However, these are often limited or not available at all. Furthermore, they often use other resources than those listed in the curriculum to prepare or conduct their courses. YouTube, NPTEL and EduSkills were mentioned as examples [121, 122, 123].
- Theme 4.2 (T4.2) Students use platforms with freely available resources for self-learning or solving project assignments.

Theme	Codes
T4.1	1, 2
T4.2	3, 4

Table 4.10: Mapping of themes to codes regarding the "Information" dimension

In terms of information, it can be noted that in addition to the objectives and learning outcomes already mentioned above, the curriculum also contains a breakdown of the course content into individual topics, for which the curriculum also refers to specific books. Besides these referenced books, teachers also use other, free sources to prepare the lessons and for the lessons themselves like YouTube, NPTEL and EduSkills [121, 122, 123]. For example, Teacher Glenn answers the question about his method of preparing a course as follows: "They [affiliating university] are giving some prescribed textbooks and reference books. Along with that we also go with the internet sources to get the present updated information about the content". However, according to Teacher Glenn, these books are usually hardly accessible, especially for students. Therefore, students almost exclusively use YouTube and the resources provided by teachers for learning and also to solve assignments.



Figure 4.4: Thematic map of the "Information" dimension

4.2.5 Management Systems and Structures

The codes listed in Table 4.11 were generated during the thematic analysis of the interviews and observations with regard to the "Management Systems and Structures" dimension.

Nr.	Code	Interviews
1	no IQAC implemented at the college	Jordan, Morgan
2	assigned courses of teachers change frequently	Glenn, Morgan
3	taught different courses over the last semesters	Glenn, Morgan
4	teachers use different on- and offline structures for	Frankie, Glenn
	coordination	

Table 4.11: Extracted codes regarding the "Management Systems and Structures" dimension

The themes of this dimension have been elaborated using the codes listed in Table 4.11 and further data extracts from the interviews and observations. Additionally, Table 4.12 lists the Themes of this dimension and indicates the codes on which the respective theme is based. The Thematic Map of this dimension, shown in Figure 4.5, presents an overview of the results of the thematic analysis regarding this dimension

Theme 5.1 (T5.1) - At the time of the interviews, there was no IQAC implemented at the partner institution.

- **Theme 5.2 (T5.2)** The courses are assigned to teachers by the management. Which teacher is responsible for which courses changes every few years, sometimes even annually or every semester.
- **Theme 5.3 (T5.3)** The teachers hold a weekly meeting in which, among other things, the organizational implementation, for example of laboratory units, is discussed and coordinated. In addition, the progress of students is also discussed.

Theme	Codes
T5.1	1
T5.2	2, 3
T5.3	4

Table 4.12: Mapping of themes to codes regarding the "Management Systems and Structures" dimension



Figure 4.5: Thematic map of the "Management Systems and Structures" dimension

In order to implement the study program, the management assigns teachers the courses they have to offer according to the curriculum. When asked about the subjects taught by the respective teacher, most also gave a history of all the different subjects they had taught in previous semesters. Also, the explicit question of whether the courses were changed frequently was also answered affirmatively.

Furthermore, the teachers involved in the implemented CS curriculum hold regular meetings to make the best use of the limited laboratory capacity and to ensure that the courses are running smoothly, as well as to coordinate the scheduling of internships and placements. They also discuss the progress of individual classes at these meetings. An IQAC (see Section 3.3.3) or a structure with similar objectives did not exist at the time of the interviews.

4.2.6 Investment Resources

The codes listed in Table 4.13 were generated during the thematic analysis of the interviews and observations with regard to the "Investment Resources" dimension.

Nr.	Code	Interviews
1	teachers work 18 to 36 hours per week in teaching	Frankie, Glenn, Jordan, Ri-
		ley
2	teachers work 36 to 48 hours per week in total for	Glenn, Riley
	all responsibilities	
3	teachers need 2 to 3 hours to prepare a topic of a	Frankie, Glenn, Jordan
	course	
4	teachers have additional responsibilities besides	Frankie, Glenn, Jordan, Ri-
	teaching	ley
5	college provides additional courses for students and	Glenn, Jordan, Morgan
	teachers	
6	additional courses for students and teachers are for	Jordan, Morgan
	free	
7	curricular courses are scheduled from 8:30am to	Nuru, Rory, Sam, Taylor
	5pm, Monday to Saturday	
8	extra-curricular courses are scheduled from 6pm to	Rory, Sam, Taylor
	7pm, Monday to Saturday	
9	occasionally special classes are scheduled for Sun-	Rory, Taylor
	days	
10	students have other college responsibilities in addi-	Rory, Taylor
	tion to their courses	
11	students do not have laptop due to lack of financial	Rory, Sam
	resources	

Table 4.13: Extracted codes regarding the "Investment Resources" dimension

The themes of this dimension have been elaborated using the codes listed in Table 4.13 and further data extracts from the interviews and observations. Additionally, Table 4.14 lists the Themes of this dimension and indicates the codes on which the respective theme is based. The Thematic Map of this dimension, shown in Figure 4.6, presents an overview of the results of the thematic analysis regarding this dimension

- Theme 6.1 (T6.1) According to the interviews, teachers work up to 48 hours per week over 6 working days. Of these working hours, 18 to 36 hours are spent on teaching responsibilities.
- **Theme 6.2 (T6.2)** Teachers also have other responsibilities than teaching such as the organization and supervision of additional courses and examinations, conducting charity projects for the local community or administrating the IT infrastructure

of the partner institution. Additionally, teachers also attend additional courses provided by the management.

- Theme 6.3 (T6.3) -During the semester students have classes from 8:30 a.m. to 5:00 p.m., including breaks. After an hour break, extra-curricular courses are usually taken from 6:00 to 7:00 p.m., after which there should be an additional 3 hours of study once or twice a week. This is all from Monday to Saturday, Sundays are usually free, although it sometimes happens that a course is held on Sunday as well or that students need this day to prepare for an exam.
- **Theme 6.4 (T6.4)** -Students are assigned by faculty to help with various events, not always taking into account their curricular and other extra-curricular responsibilities.
- **Theme 6.5 (T6.5)** -According to one of the interviewed students, most of the students in general do not have enough financial resources to buy a laptop.

Theme	Codes
T6.1	1, 2, 3
T6.2	4, 5, 6
T6.3	6, 7, 8
T6.4	10
T6.5	11

Table 4.14: Mapping of themes to codes regarding the "Investment Resources" dimension



Figure 4.6: Thematic map of the "Investment Resources" dimension

In terms of the time investments of both groups, it was found that both students and teachers have a work week that consists of 6 days. Teachers say they work up to 48 hours per week, with 18 to 36 hours spent on teaching, depending on their assigned courses. Other responsibilities, such as IT administration, supervising exams, or charity work for the local community as part of their work for the partner institution, then take up the rest of their time. Research activities were not mentioned by teachers during the interviews.

Students have very intense days from Monday to Saturday, from 8:30 a.m. to 7:00 p.m. (including breaks) and sometimes even on Sunday mornings. In addition, there are other tasks assigned to them by teachers or the management, which they cannot refuse. These sometimes lead to a missed course session, which is not taken into account subsequently. According to some students, the psychological pressure on students is very hard and the workload often seems unmanageable. For example, Student Rory stated: "I think here students' life is very tough situation. That means [...] students are not only permitted to only study. They'll be having many other works". All but one of the students interviewed confirmed that they have other responsibilities at the partner institution besides studying.

Regarding financial resources, for some students financial constraints make it very difficult to buy a laptop, especially due to the COVID-19 pandemic [3]. Student Sam stated: "[...] we [students] cannot ask the parents I want to buy laptop because Corona times there is no work". Furthermore, the space available for students to study or work at the hostel also depends on a student's financial resources. According to one student, some students share a room with one or two fellow students, while others share a room with up to 11 fellow students, depending on the amount of rent they can pay.

4.2.7 Staffing and Skills

The codes listed in Table 4.15 were generated during the thematic analysis of the interviews and observations with regard to the "Staffing and Skills" dimension.

The themes of this dimension have been elaborated using the codes listed in Table 4.15 and further data extracts from the interviews and observations. Additionally, Table 4.16 lists the Themes of this dimension and indicates the codes on which the respective theme is based. The Thematic Map of this dimension, shown in Figure 4.7, presents an overview of the results of the thematic analysis regarding this dimension

- Theme 7.1 (T7.1) Most teachers did not gain professional experience before entering the higher education sector as a teacher. Typically, teachers came to the partner institution directly after finishing their postgraduate studies.
- Theme 7.2 (T7.2) According to teachers, the frequent change of assigned courses often leads to the situation that teachers have to teach subjects in which they have little or no previous experience or knowledge. As result, teachers were not able to evaluate students' projects properly.

52

Nr.	Code	Interviews
1	teachers usually do not have professional experience	Frankie, Glenn, Riley
	in the subjects they teach	
2	teachers came directly from college into HE sector	Frankie, Glenn, Riley
3	teachers often do not have teaching experience be-	Frankie, Jordan
	fore joining the college	
4	assigned courses of teachers change frequently	Glenn, Morgan
5	taught different courses over the last semesters	Glenn, Morgan
6	teachers are not able to evaluate projects of students	В
	correctly	
7	occasionally teachers seem to teach subjects without	Nuru, Rory
	deeper knowledge	
8	teachers are good in general	Rory, Sam, Taylor
9	students are not able to work independently on their	B, C, D
	final project	
10	students use ready-made solutions from internet	B, C, D
	platforms for assignments	
11	teachers provide solutions to assignments	A, C

Table 4.15: Extracted codes regarding the "Staffing and Skills" dimension

- Theme 7.3 (T7.3) Students also state that it happens that teachers teach a course without having deeper knowledge in the respective subject. Nevertheless, the majority of the interviewed students think that their teachers are good in general.
- Theme 7.4 (T7.4) Students of the final semester were not able to elaborate a small software project on their own, but instead relied on ready-made solutions, which they slightly modified.

Theme	Codes
T7.1	1, 2, 3
T7.2	4, 5, 6
T7.3	7, 8
T7.4	9, 10, 11

Table 4.16: Mapping of themes to codes regarding the "Staffing and Skills" dimension

Regarding the skills of teachers, it was found that most teachers interviewed did not have any practical experience in the industry. Those who decide to pursue a teaching career usually come to a HEI directly after completing a post-graduate degree program and take over teaching certain courses there as needed, regardless of how experienced they are in the respective subject areas, which affects the quality of teaching according to some of the students. For example, Student Nuru confirmed that teachers often give the



Figure 4.7: Thematic map of the "Staffing and Skills" dimension

impression that they do not know the subject they are teaching. Therefore, especially experienced teachers who have been able to teach their subjects for a long time are more popular among students, although young teachers also stand out positively, especially when they work with practical examples that tie in with the reality of students as it is mentioned in Section 4.2.2.

During Observation B it was noted that teachers were not able to evaluate graduation projects sufficiently. For example, one group presented the configuration file of a development environment for several minutes as source code of the website they have developed for their project. Since the configuration file was written in XML, it was structured similarly to the source code of a website. However, from the point of view of the author of this thesis, an experienced person from the field of CS would have noticed very quickly that the tags used in HTML were not present and that it could therefore not be a website. Furthermore, during the presentation it was recognizable which file was displayed, including its path. Therefore, from the point of view of the author of this thesis, it was easily recognizable for an experienced person from the SE field that a configuration file of the development environment was shown. However, this was not noticed by any of the teachers.

Regarding the skills of students, as described in Section 4.2.2, it was noted through Observations C & D that students in the eighth and final semester were not able to independently analyse problems that arose in the context of their graduation project and work out possible solutions on their own. Instead, students limited themselves to the procedure of trying to modify already existing projects that were found on the internet or brought in by colleagues, as if they had been developed by themselves. However, it was still claimed that it was entirely their own work and not a plagiarised one. For example, one of the supervised groups only admitted that it was a copy after asking several times and being confronted with the source. It turned out that in the beginning they have received the link to the project from their supervising teacher.

4.2.8 Milieu

The codes listed in Table 4.17 were generated during the thematic analysis of the interviews and observations with regard to the "Milieu" dimension.

Code	Interviews
college provides additional courses for students and	Glenn, Jordan, Morgan
teachers	
students move to major cities for jobs after gradua-	Glenn, Jordan
tion	
no local industry for CS graduates in the area of	Glenn, Jordan
the college exists	
teachers have additional responsibilities besides	Frankie, Glenn, Jordan, Ri-
teaching	ley
students have other responsibilities to the college in	Rory, Taylor
addition to their courses	
	Code college provides additional courses for students and teachers students move to major cities for jobs after gradua- tion no local industry for CS graduates in the area of the college exists teachers have additional responsibilities besides teaching students have other responsibilities to the college in addition to their courses

Table 4.17: Extracted codes regarding the "Milieu" dimension

The themes of this dimension have been elaborated using the codes listed in Table 4.17 and further data extracts from the interviews and observations. Additionally, Table 4.18 lists the Themes of this dimension and indicates the codes on which the respective theme is based. The Thematic Map of this dimension, shown in Figure 4.8, presents an overview of the results of the thematic analysis regarding this dimension

- Theme 8.1 (T8.1) Additional courses are provided for both teachers and students. Company-related courses offered on the EduSkills platform, for example on AWS and Cisco, are attended by both students and teachers [124, 125]. Individual companies hold extra-curricular courses for students on certain topics, such as programming with Python or Java, so-called internships. Those students who do well in their course from the company's point of view then have the chance of getting a job after their graduation.
- Theme 8.2 (T8.2) Both students and teachers state that the local industry near the partner institution has no need for graduates of the CS curriculum. In this respect, Bangalore and Hyderabad are mentioned as the cities where students are most likely to live and work after completing their studies.
- Theme 8.3 (T8.3) The partner institution interacts with the local community through charity work on the one hand and events open to everyone in the area on the other.

The charity work focuses on the support of villages or slums by students and teachers in various activities. The events are organised on various occasions, such as International Women's Day or Independence Day [126].

Theme	Codes
T8.1	1
T8.2	2, 3
T8.3	4, 5



Table 4.18: Mapping of themes to codes regarding the "Milieu" dimension

Figure 4.8: Thematic map of the "Milieu" dimension

Concerning the milieu of the curriculum's implementation, it was found that the external stakeholders consist of the JNTUA as the affiliating university, the companies that are interested in graduates of the study program and the local community in or near the area of the partner institution.

Several companies cooperate with the partner institution and offer additional courses in Java or Python programming, for example, as internships that take place independently from the curriculum. Furthermore, students of the last semester have the chance to get a job at a company by passing additional exams, which are called placements. Teachers and students have confirmed that taking part in internships and placements is mandatory. For example, when asked whether students work near the partner institution or move away after graduation, Teacher Glenn explained: "After they [students] graduated we'll [teachers] bring the companies for the recruitment to the college itself and whatever the placement examinations are, we'll be organizing that they take an exam in college and wherever they get selected they'll moving to those remote locations". When asked where they would like to work in the future, all the students answered that they would like

to live and work somewhere other than their home town, for example in Bangalore or Hyderabad, but also outside India.

Additionally, the partner institution offers company-related courses from the EduSkills platform. Teachers Glenn and Morgan stated that both, students and teachers, attend courses on this platform, for example on AWS or Cisco [124, 125] services. Furthermore, Teacher Jordan confirmed that in addition to their teaching responsibilities, s_he and the partner institution also support children and their parents who live in small, "backward" villages.

According to a report by the management of the partner institution, the faculty carries out a social responsibility projects with students between the end of the fourth semester and the beginning of the fifth semester. Students develop projects on topics such as "Applications for drones in agriculture", "Applications of solar technologies for rural purpose" or "Smart irrigation for saving water". The social responsibility project became known to the author of this thesis only after the completion of data collection through observations and interviews, as it was introduced within a curriculum revision that has been developed in parallel with this work. As a result, it was not possible to collect more detailed data on it.

CHAPTER 5

Design-Reality Gap Analysis

This chapter presents the analysis of the thematically analyzed gathered data presented in Chapter 4 with the help of the DRGA, which is is described in Section 5.1.1. Then, solutions for bridging the identified design-reality gaps are proposed. As it is previously described in Section 4.1, this thesis has been realized with a partner institution, which is 1 of over 80 affiliated colleges that implement the same curriculum designed by a nationally accredited state university.

5.1 Methodology

The thematic analysis of the gathered data is further analyzed using the DRGA. The DRGA, which is explained in more detail in Section 5.1.1, was used in order to determine the differences between the design and the actual implementation of the CS curriculum and thus to be able to answer **RQ1**. As in the Chapter 3, the DRGA also analyzes the thematically analyzed data along the OPTIMISM dimension that are based on the DRGA by Bass & Heeks [127].

While conducting the DRGA, the author of this thesis noticed that there is thematically analyzed data which, strictly speaking, would not be taken into account by the DRGA, since no opposing design part existed to certain realities. These data, which therefore do not represent gaps between design and reality in the strict sense, are summarized in Section 5.3.2 in further detail after the DRGA.

The identified gaps and challenges were each summarized in a separate list for the purpose of an overview, followed by more detailed description of each gap/challenge and the findings from the DRGA they are based on. If findings of two or more different dimensions were related, they were combined into one gap, such as G1.



Figure 5.1: Adapted OPTIMISM model of Bass & Heeks [127, p. 6]

5.1.1 Design-Reality Gap Analysis

The DRGA framework was originally designed by Heeks and is used as a tool to analyze the deviation as well as the extent of deviation of the resulting reality from its planned design, especially for Information and Communications Technology for Development (ICT4D) projects [128]. This can be applied to the evaluation of an implemented IT/IS system, for example, to e-government projects, but also to a curriculum or its modified version [127, 129, 130, 131]. Bass & Heeks adapted the DRGA and used the OPTIMISM model, which is shown in Figure 5.1, to consider the dimensions of a curriculum or the modification of it. Along these dimensions, the gap between the design of the curriculum and the realized implementation of it were examined.

With the help of the DRGA, it was examined whether and to what extent the design of a CS curriculum deviates from the implementation in reality. The existing curriculum as well as publicly available information about the curriculum and its environment are used to conclude the design part of the DRGA analysis. The reality part of the DRGA is based on the perceptions and experiences of teachers and students of the partner institution, which were gathered by semi-structured interviews and observations.

5.2 Design-Reality Gap Analysis of the Curriculum and its Implementation

In the following section, the differences between the curriculum developed by the affiliating university and its implementation at the partner institution, among others, are outlined using the DRGA. The following DRGA is structured according to the OPTIMISM dimensions by Bass & Heeks, which are examined in Section 5.1.1. As it is described in Section 5.1.1, for each of these dimensions, the Design part is based on the CS curriculum itself and information that the affiliating university communicates through its websites. As the affiliating university of the partner institution is accredited, the Design part is complemented with the NBA and NAAC guidelines and criteria discussed in Section 3.3.3 The Reality part consists of the thematically analyzed interviews and observations of teachers and students, which are described in Section 4.2, as well as information that the partner institution communicates through its website.

5.2.1 Curriculum

The curriculum consists of several documents, each of them containing the regulations for up to two academic years [132, 133, 134]. Each academic year consists of 2 semesters. At beginning of the document the courses that are held in the respective semester are listed for each semester (see Table 5.1). Each course is categorized, for example, into Basic Science (BS), Engineering Science (ES) or Professional Core (PC). Furthermore, for each course it is defined how many hours per week the course is taught in the respective semester and by which method. Thus, a distinction is made between Lecture (L), Tutorial (T), and Practical (P) hours. The amount of hours then determines the credits, which are used as a unit for the course workload.

No	Course Code	Course Name	Category	Hours/Week			Credits
110.			Category	L	Т	Р	
1.	20A54304	Discrete Mathematics &	BS	3			3
		Graph Theory					
2.	20A05301T	Advanced Data Structures	PC	3			3
		& Algorithms					
3.	20A05301P	Advanced Data Structures	PC			3	1.5
		& Algorithms Lab					

Table 5.1: Example of course overview in the curriculum [133, p. 3]

The majority of the curriculum consists of the course descriptions. The course description includes the "Course Objectives", "Course Outcomes", and the contents of each course unit. In addition, text books, reference books and online learning resources are listed (see Appendix B).

5.2.2 Objectives and values

This dimension considers the objectives of the curriculum and the values it wants to transport through its implementation.

Design

The objectives and values of a curriculum are considered by both bodies, the NBA and the NAAC. As described more specifically in Section 3.3.3, the NBA recognises this dimension with their first criteria, which describes that an institution must have a vision and a mission and a curriculum possesses multiple POs and PSOs. The NAAC lists "Institutional Vision and Leadership" and "Institutional Values and Social Responsibilites" in its set of KIs (see Table 3.1).

The affiliating university and the partner institution are both communicating their individual vision and mission as well as the curriculum's POs and PSOs with the help of their websites. The vision and mission of the affiliating university are both listed on their website's frontpage [135]. Regarding the POs and PSOs, all of the curricula offered on the affiliating university website do not contain general or specific objectives, but they do explain within their Methodology page which teaching methods they use to achieve certain objectives. Theses POs include initiating leadership in their students and prepare them "to meet project requirements of the industry" [136].

The curriculum includes a wide variety of CS courses, such as "Object Oriented Programming", "Software Engineering", "Web and Internet Technologies" or "Operating Systems" [132, p. 3, 133, p. 3, 134, p. 1-3]. As it is described in Section 5.2.1, the CS curriculum defines objectives and learning outcomes for each course of the curriculum. In addition, each course is divided into individual units whose learning outcomes are also listed. In the laboratory courses, there is usually only one unit, which also has objectives and outcomes [132, p. 47]. Furthermore, it was noted that the course articulation matrix recommended by the NBA to outline the relationship between POs/PSOs and courses is not indicated in the latest version of the curriculum reviewed [132, 133].

Reality

The vision and mission of the partner institution are outlined at their website [137]. Furthermore, they also published a vision statement for their curriculum implementation ("Excellence in creating globally competent professionals and leaders in the field of computer science & engineering") and mission statements for their curriculum implementation such as "Developing Communication Skill, Teamwork and Leadership Qualities for Continuing Education Among the Students, Through Project-Based and Team-Based Learning" or "Empowering Students for Employability, Aspiring Higher Studies and to become an entrepreneur" [138].

Additionally, they list PSOs for their curriculum implementation such as "The ability to understand, analyze and develop computer programs in the areas related to system software, multimedia, web design, big data analytics, and networking for efficient design of computer-based systems of varying complexity" or "The ability to apply standard practices and strategies in software project development using open-ended programming environments to deliver a quality product for business success" [138].

Since the mission statements of the curriculum implementation and the PSOs were published after the interviews, students and teachers could not be interviewed about them explicitly. Regarding the courses and their objectives, it was found that they change almost every two years due to a revision by the JNTUA (see T1.4). According to teachers, there is a tendency for courses that combine the basics with new topics to replace the existing courses.

Teachers seemed to be very concerned that as many students as possible pass the final exams (see T1.3). It can be stated that it is particularly important for teachers that 100% of the students pass the exams at the end of the semester, regardless of the skills that students actually possess afterwards. From the point of view of the author of this thesis, teachers fully trust the end exams by the affiliating university to be able to check the actual learning success and those students who have passed all the exams are then also expected to achieve the goals and learning outcomes of the curriculum and therefore have mastered the skills listed in the curriculum.

However, some students feel that in this curriculum implementation they mainly study for the exams, but not to acquire the knowledge that a software developer should have from their point of view. Students are very focused on their own goal, which is to become software developers or engineers (see T1.1).

Regarding the role of extra-curricular courses, it is noted that both students and teachers confirmed that the curriculum implementation only seems to be useful through the combination of the curriculum and internships. Furthermore, teachers and students have confirmed that the placements at the end of their studies, whose role was described in more detail in Section 4.2.8, are seen as a crucial part of the curriculum implementation as they help students to enter the respective labour market in the first place (see T1.2).

Gap Analysis

As required by both accreditation organisations, the affiliating university formulates objectives and learning outcomes for the individual courses in its curriculum that the partner institution implements. The POs and PSOs are not listed in the curriculum, but the program-specific visions, missions and objectives are described analogously on the websites of the partner institution. The students' goal to become a software engineer aligns with the program-specific vision statement of the partner institution in terms of becoming a "competent professional". The frequent changes to the curriculum and its courses are in line with the recommendations of the accreditation bodies regarding the continuous improvement of a curriculum. However, it was outside the scope of this thesis to determine whether the curriculum actually improves as a result of the constant changes.

5. Design-Reality Gap Analysis

Dimension Finding 1 (DF1): One of the two findings of this gap concerns the objectives of the curriculum and whether students actually achieve these objectives by completing the implemented curriculum in the end: The subject areas listed in the PSOs, in which students are supposed to be proficient at the end of their studies, correspond to the contents and objectives of the courses in the curriculum. However, through the observations of the author of this thesis regarding the teachers' confidence in the JNTUA examinations and the students' statements about their studies, it is found that it is never examined whether the objectives of the courses are actually achieved and whether students have actually acquired the respective skills at the end of their studies.

Dimension Finding 2 (DF2): The other finding in this dimension concerns the employability of students after graduation. Although this topic is not explicitly considered in the curriculum and is therefore not a part of the curriculum in the proper sense, it is mentioned in the teaching methodology of the affiliating university. According to their methodology, they designed the curricula that will equip the students with employability [136]. Additionally, the partner institution's program-specific missions include to make students employable [138]. The partner institution pursues this mission exclusively with courses that are not part of the curriculum but are offered by companies. According to students and teachers, this combination of curricular and extra-curricular courses is what makes it possible for students to enter the job market later on. Furthermore, from the point of view of the author of this thesis, the existing combination of the implemented curriculum with internships and placements only determines whether the skills and knowledge that students have acquired through their studies fit into the respective companies involved, but not whether the objectives of the curriculum have actually been achieved. Therefore, it is determined whether the respective students are employable at a corresponding small subset of companies, but not their employability in a broader sense. Therefore, it is noted that students do not achieve the goal of general employability through the implemented curriculum alone. However, it was out of the scope of this thesis to determine whether the curriculum itself already enables general employability.

5.2.3 Processes

The "Processes" dimension includes teaching approaches, assessment methods and evaluation processes that are used for conducting and revising a curriculum.

Design

Both the NBA and the NAAC take this dimension into account, each with its own criterion. In addition to the Teaching-Learning process, the NAAC criterion also covers the evaluation and revision of the processes themselves and includes student satisfaction. As described in Section 3.3.3, the NBA recommends that teaching approaches and assessment methods in use are described in the curriculum. Nevertheless, no teaching methods are listed in the curriculum examined, but the courses are divided into theory

courses and practice-oriented laboratory courses. Furthermore, no assessment methods or evaluation processes in use are described in the curriculum.

The affiliating university states its teaching methodology on the institutional level on its website. According to this methodology, they take advantage of student-centered teaching approaches, using interactive teaching-learning methods such as group discussions, seminars and project-based learning, which are also to included in the assessment alongside the examinations at the end of the semester [136]. The curriculum lists objectives and learning outcomes for each course, but teaching methods are neither prescribed nor described in it.

Reality

The partner institution specifies through one of their program-specific mission statements that they use project-based and team-based learning for the implementation of the curriculum [138]. According to teachers and students, the teaching methods used depend on the respective teacher, but are overall highly teacher-centred since teachers mainly present the contents of a course and students participate by listening, taking notes or asking questions (see T2.2). Although the laboratory classes aim to be practice-oriented, in some laboratories the actual teaching method is nevertheless teacher-centered, as students type the previously presented solution for an assignment into the computer, which means that students do not work out the solutions to the laboratory assignments themselves (see T2.3). Therefore it can be noted that the partner institution uses teacher-centered teaching approaches almost exclusively. However, the partner institution conducts project-based learning with projects that students have to work on each semester in small groups of 3 to 4 students.

Furthermore, during their last semester, students are supposed to work on a concrete project and write a report about it. However, the author of this thesis observed that both students and teachers tend to use ready-made solutions instead of trying to develop their own solution (see T2.3). Under the corresponding project titles, tens of thousands of these reports and ready-made solutions for these projects, which show great similarities to each other, can be found on various commercial and non-commercial web platforms. From the point of view of the author of this thesis, this indicates that copying instead of actually creating a solution to a certain task seems to be a widespread phenomenon, at least in the context of the Indian higher education system.

Students are evaluated solely on the basis of their scores on the examinations in the middle and at the end of the semester and are prescribed and designed by the affiliating university of the partner institution. Apart from these examinations, teachers also use smaller tests, which do not seem to be relevant for the evaluation of students, in order to regularly check their learning progress. These tests are largely based on written tests whose questions are given by the affiliating university or have already been given in the past for the same tests (see T2.1.

Regarding feedback processes, students and teachers confirmed that student feedback is gathered at least once per semester (see T2.4). Students rate their teachers on different criteria on a 3 or 4 point scale (see Appendix C). These criteria include the teacher's knowledge of the course content, the ability to organize the course, or the teacher's effectiveness in teaching the course. There is also a feedback process between the partner institution and the affiliating university through which teachers can share their feedback about the curriculum and its courses with the affiliating university in order to initiate changes to the curriculum (see T2.5).

Gap Analysis

In contrast to the recommendations of the accreditation bodies, the curriculum does not contain specifications on the teaching approaches or assessment methods to be used, nor do internal guidelines of the partner institution or anything similar exist. Similarly, there are no routines to improve both the teaching and learning process. Furthermore, it is noted that the curriculum includes the observed graduation project, but not the smaller ones in the previous semesters mentioned by students during the interviews [134, p. 3]. However, the graduation project is only listed in the course overview of the last semester, but unlike the other courses, it is not explained and described in more detail in the curriculum.

Dimension Finding 3 (DF3): A gap between PBL in theory and its actual implementation at the partner institution was found, for example in the observed graduation project course. This course used both project-based and team-based learning, as outlined in one of the program-specific mission statements mentioned above. One difference in this regard is that the implementation of PBL at the partner institution has a different understanding of this teaching methodology: It has been observed that both students and teachers tend to copy a ready-made solution from somewhere else instead of developing their own solution to a problem. A similar approach was also observed in the laboratory courses where the teacher provided a solution instead of letting students work out the solution to an assignment by themselves.

Dimension Finding 4 (**DF4**): Even though the teachers themselves regularly communicate with each other and with their supervisors in weekly meetings, no quality assurance or enhancement processes have been found that evaluate and revise courses in a regular and institutional way. However, feedback is solicited from students in which they evaluate their teachers and their abilities at the end of each semester. Nevertheless, contrary to the NAAC criteria, student satisfaction is not sought in an institutional manner.

5.2.4 Technology

The technology dimension sums up the infrastructure that is needed to implement a curriculum.

TU Bibliothek, Die approbierte gedruckte Originalversion dieser Diplomarbeit ist an der TU Wien Bibliothek verfügbar WIEN Your knowledge hub. The approved original version of this thesis is available in print at TU Wien Bibliothek.

This dimension is also represented by a separate criterion by both the NAAC and the NBA. According to both bodies, a HEI must ensure that access to adequate infrastructure is provided so that students can complete the curriculum and the faculty can teach it. The NBA adds that a HEI has to offer self-learning facilities. The curriculum itself does not contain any specifications as to which infrastructure and technologies are to be used in order to be able to implement the curriculum.

The affiliating university specifies the infrastructural requirements in the affiliation regulations. In the case of a permanent affiliation, for example, the computer-student-ratio, the available internet bandwidth and the electricity and water supply are considered. In addition, it is also checked whether adequate equipment is available for each curriculum implemented at the college, whereby the adequacy is oriented towards AICTE standards [139, pp. 7-11].

Reality

The partner institution has a comprehensive description of various physical facilities that are located on their respective campuses [140]. According to the thematic analysis of this dimension, the facilities are overall adequate for conducting the respective curriculum when it comes to its individual courses (see T3.1). However, the available facilities for self-learning are very limited, especially in terms of existing computers available. Furthermore, the computers available outside of class time are almost always already occupied by others, and since most students only have a smartphone and not a laptop, they cannot, for example, continue working on projects or deepen practical knowledge (see T3.2).

Gap Analysis

Overall, the partner institution provides an adequate infrastructure for the delivery of curricular and extra-curricular courses with no differences between the design and reality.

Dimension Finding 5 (DF5): In terms of the infrastructural needs of students, it is noted that there are very few learning spaces available for all students at the partner institution and the availability of computer workstations outside of courses is very low, which in turn would be of great importance for CS students to work independently on software development skills, for example.

5.2.5 Information

This dimension considers the contents of the curriculum that are necessary when implementing it.

The NAAC addresses this dimension using the "Curricular aspect" criterion. As it is shown in the Table 3.1, depending on the type of HEI, the NAAC assesses either the design and development or the planning and implementation of the curriculum. Furthermore, as described above in Section 3.3.3, according to the NBA, each curriculum should both reflect its structure and show which courses contribute to which goals and to what extent. In addition, the NBA specifies more concrete contents of a curriculum through a criterion specifically described for this purpose.

In addition to the objectives and outcomes, the curriculum describes the contents of each course, divided into topics and refers to those books on which the contents are based. Furthermore, it lists online learning resources for further studying and specific assignments for laboratory courses. In some of the courses, the assignments are also divided into weeks, so that it is indicated in which week of the semester which assignments of a course are to be discussed.

Reality

The referenced books for each course in the curriculum are often used by teachers to prepare their courses. Furthermore, teachers use resources from other platforms such as YouTube or NPTEL to prepare and conduct their courses (see T4.1). Similar to the teachers, students use such platforms for self-learning. However, students also tend to use ready-made solutions from different internet platforms to solve assignments (see T4.2).

Gap Analysis

Apart from the limited availability of the books listed in the curriculum, the thematic analysis of the gathered data do not reveal any major gap between design and reality. A deviation of the contents listed in the curriculum from those actually taught in the corresponding course could not be found. The existing online learning resources were added in the latest version of the curriculum and therefore were unknown by the interview partners [141, 133].

5.2.6 Management Systems and Structures

This dimension describes management systems and structures that are used for the implementation, conduction, evaluation and revision of a curriculum. The curriculum itself does not describe the structures and systems needed to implement it. However, as it is described in Section 3.3.3, the "Quality mandate" of the UGC aims to have all HEI accredited to NAAC standards by 2022. Furthermore, the NAAC recommends alike systems and structures at HEI after they have been accredited. Therefore, for this dimension, the Design part is composed by the concepts of the NBA and NAAC regarding this dimension.

68

The NAAC and the NBA use criteria to verify that systems and clearly defined structures are in place to govern an HEI, enabling the implementation and conduction of a curriculum in the first place (see Section 3.3.3), but they take different approaches regarding the systems and structures for evaluating and revising a curriculum. As described in Section 3.3.3, the NBA in its seventh criterion, requires that a curriculum needs to be regularly analyzed, evaluated, and revised, but does not specify a concrete system for doing so. Similarly, the NBA states in its fourth criterion that the HEI must ensure that it can adequately monitor the students' performance. The NAAC, in turn, calls for the establishment of a dedicated committee at each institution for the purpose of quality assurance, the already mentioned IQAC. As it is shown in Table 3.1, the NAAC uses several key indicators, such as "Evaluation Process and Reforms", "Student Performance and Learning Outcomes", or "Student satisfaction survey", to verify the existence of management systems for quality assurance. Both the NAAC and the NBA take into account the support of students through their own systems. According to the NBA, the respective institution must set up a system that supports students on the one hand and offers various programs and activities on the other. In addition, the NAAC also takes into account the commitment of alumni with the help of its own indicator. The curriculum itself does not list any systems or structures necessary for its implementation, conduction, evaluation or revision.

Reality

The examined partner institution has a separate management position for each department, which in turn is headed by a director and a chairperson. It was found that the faculty members have informal systems, such as a weekly meeting of all teachers that are involved in the implementation of the curriculum (see T5.2). They use these for content-related but also organizational coordination in order to be able to implement the respective program. However, no IQAC or comparable organizational structure was found to exist (see T5.1).

According to the interviews and the perceptions of the author of this thesis, the partner institution has a functioning management structure. However, during the interviews, teachers and students confirmed that the allocation of teachers to courses changes quite frequently due to managerial decisions (see T5.3). As a result, teachers sometimes have little or no previous experience in the courses they are assigned to teach and therefore often have to familiarise themselves with the subject matter first.

Gap Analysis

In comparing the reality and design according to concepts of the NBA and NAAC, a gap was identified in the systems for evaluating or revising the curriculum. **Dimension Finding 6** (DF6): Although an affiliated college cannot revise the curricula it implements, there should still be a structure for evaluating them. Moreover, the partner institution conducts not only courses of the curriculum, but also extra-curricular courses that are offered by companies and other stakeholders. Since these are outside the direct control of the affiliating university, they can be directly controlled and revised by the management of the partner institution.

5.2.7 Investment Resources

The "Investment Resources" dimension displays the resource needs of a curriculum in terms of financial resources of the HEI and time resources of the respective faculty members. However, it was outside the scope of this thesis if labor policies such as employment protection, paid sick leave or unemployment benefits exist.

Design

As explained above in Section 3.3.3, according to the NBA, teachers' time resources must be sufficient to actually implement the curriculum. Furthermore, the NBA emphasizes that there must be time resources for other activities in addition to teaching responsibilities of the faculty members. The NAAC, on the other hand, does not have a criterion nor a key indicator that considers faculty resources. In contrast, both bodies consider financial resources. The NAAC lists "Financial management and resource mobilization" as a key indicator (see Table 3.1) and the NBA, through its tenth criterion, requires that an institution's financial resources must be sufficient for implementing the curriculum in order to meet its objectives and those of the HEI. As it is described in Section 5.2.1, the curriculum specifies the weekly hours for each course and divides them into hours for lectures, tutoring and practicals. On average, the latest version of the curriculum specifies 28 hours per semester [132, p. 3, 133, p. 3].

Reality

When collecting the data, financial resources of the partner institution were not considered, as the focus was on the everyday life of teachers and students and their resources. It was found that students with fewer financial resources have a hostel room with more people per square meter, which means that there is less potential learning space available for those students. Additionally, the lack of a computer due to their financial constraints makes it difficult for most students to independently learn or deepen their knowledge outside of courses (see T6.5).

Regarding the time resources, it was found that teachers work approximately 48 hours on 6 days per week, whereby 18 to 36 hours are necessary for their teaching responsibilities (see T6.1). Additionally, teachers have other responsibilities such as administrating the IT infrastructure or conducting charity projects and events for the local community (see T6.2).

Students also work for 9.5 hours per day, including breaks, for 6 days per week minimum. Their responsibilities include curricular and extra-curricular courses as well as helping to conduct events of the college (see T6.3 and T6.4). From the point of view of the author of this thesis, the students' intense schedule barely leaves them time to deal with subjects more intensively or even to recover completely. Students expressed the existence of very high pressure and a partial overtaxing of their time resources even without a corresponding question from the interviewer.

Teachers did not make any statements of their own accord that indicated that they felt overburdened with their workload or that their time resources may be insufficient for conducting their courses.

Gap Analysis

Based on teachers' statements, it is determined that their time resources are sufficient to implement the curriculum in reality, so no difference between design and reality in this dimension could be determined with regard to the teachers.

Dimension Finding 7 (DF7): Regarding the students, it was found that their situation causes some of them considerable stress and is very burdensome for some of them, although strictly speaking no difference can be found between design and reality. Therefore, from the point of view of the author of this thesis, it is important to note that the time resources of students are not taken into account by the accreditation authorities, the university or the partner institution. Although the curriculum specifies the number of hours students spend in lectures or laboratories, it does not take into account the time resources required to learn the material. In addition, the extra-curricular activities, including additional courses offered by companies, are also not taken into account.

5.2.8 Staffing and Skills

This dimension describes the skills that students learn during their studies according to the curriculum as well as the skills that faculty members should possess to teach parts of the curriculum.

Design

The skills of faculty members are taken into account by the NAAC in the accreditation process with the help of the key indicator "teacher profile and quality" (see Table 3.1). The NBA devotes a separate criterion to this dimension, which requires the respective institution to ensure that a sufficiently large number of faculty members are available in order to be able to adequately implement the curriculum. The NBA also considers the workload of faculty members and their ability to work besides of their teaching duties such as their own development or mentoring students (see Section 3.3.3). The curriculum describes for each course and partly also for each unit of a course the skills that a student

should have after completing the unit or the course. The curriculum does not contain any information about the skills teachers need to conduct their courses.

Reality

According to the interviewed teachers, they usually have no practical experience at the beginning of their careers; many started their teaching career at the partner institution itself and therefore also had no teaching experience at first (see T7.1). They acquire the skills and knowledge required for the individual courses on their own. Furthermore, due to the above-mentioned demand-oriented assignment of courses to teachers and the resulting frequent changes in these assignments, teachers sometimes have very different skill levels in the courses they are supposed to conduct, whereby it often happens that teachers have to teach a course in which they have no previous experience or skills (see T7.2 and T7.3).

When working with students on their graduation project, it has been observed by the author of this thesis that students were not able analyse or solve even small problems by themselves at the end of their studies (see T7.4). In the case observed, students copied a ready-made solution for their project assignment provided by their supervising teacher and tried to modify it. Although they had a ready-made solution, they were not able to change the source code of it in such a way that a different image was displayed than that of the copied solution.

With regard to the teachers, observations of the regular review sessions of this course, as it has been described in Section 4.2.7, have shown that they are mostly not capable of carrying out an actual evaluation of a software project (see T7.1). Regarding the concrete observed situation, from the point of view of the author of this thesis, an experienced person from the field of CS or SE would have noticed very quickly that shown source code is very different to a common website and that a configuration file of the development environment was shown so it could therefore not be the website created by the respective group members. Additionally, as already stated in Section 5.2.3, some teachers tended to use ready-made solutions instead of helping students to develop their own solution to their project assignment (see T2.3).

Gap Analysis

Empirically evaluating whether the skills of the teachers are sufficient for the actual implementation of the curriculum or whether students actually acquire the skills specified in the curriculum was outside the scope of this thesis. Also, it was not empirically examined whether both students and teachers have the technical skills according to SWECOM that have been described in the Fundamentals of this thesis (see Section 3.1.2) [60].

Dimension Finding 8 (DF8): However, from the point of view of the author of this thesis, there are indications that there is a big gap between the skills that students should

have by design at the end of their studies and those that they actually have in reality. One of these indications was found through the interaction of the author of this thesis with student groups in the context of their graduation project. In this specific case, it was found that students did not possess certain skills that they would have acquired through several courses of the curriculum. According to the course descriptions in the curriculum, at least one course, Web and Internet Technologies, should have enabled them to solve at least small problems in the observed context, especially since they are students in their final semester in a 4-year CS degree program [134, pp. 44-45].

Dimension Finding 9 (DF9): Most teachers have no professional experience before they start teaching, and some of them have no teaching experience either. In addition, teachers are not informed by the curriculum or the partner institution about potential teaching methods and the teaching methods applied. Therefore, it is not assured that teachers are able to implement PBL in such a way that it can help students to actually achieve the learning objectives. Furthermore, when observing teachers in a situation where they were reviewing students' projects, it was found that they were not able to detect errors and inconsistencies in the presented projects. However, from the point of view of the author of this thesis, obvious errors or inconsistencies of the type observed must be noticed by a person who teaches in this subject area and is supposed to supervise such projects. As described in the Design-part of this dimension, while both accreditation bodies, NBA and NAAC, address teachers and their quality, the curriculum itself and the affiliating university as well as the partner institution do not seem to have sufficient quality requirements for teachers. However, the observed skills of teachers indicate to the author of this thesis that teachers do not have the skills required by the accreditation bodies' design for the adequate implementation of the entire curriculum in reality.

Dimension Finding 10 (**DF10**): The curriculum design does not specify the skills and experiences that a teacher must have to be able to teach a certain course. Thus, no gap between design and reality in the strict sense can be determined. However, it is noted that the implementation of the curriculum requires constant reassignment of courses based solely on demand and not on the existing skills and experiences of the respective teachers. Teachers report that it is always very challenging when they have to teach subjects in which they have no previous experiences or skills. Students confirm that it often happens that teachers give the impression that their skills and experiences in the respective topics of the course are not sufficient to be able to conduct the course adequately.

5.2.9 Milieu

The Milieu dimension includes the key factors of the institution's environment and the way the environment and the institution mutually influence each other.

Both, the NAAC and NBA consider this dimension in considering the collaborations between faculty members of the HEI and the industry. The NAAC also takes the social responsibilities of a HEI into account.

The curriculum considers this dimension through a mandatory 'Community Service Internship/Project' for 6 weeks during the students' summer vacation after the fourth semester [133, pp. 50-56]. Like the other courses in the curriculum, the description for it contains a detailed plan for its implementation and the objectives of the project.

Reality

The companies are not directly involved in the implementation of the curriculum itself and no direct or indirect influence of the companies on the implementation of the curriculum itself could be determined. Nevertheless, companies from the cities of Hyderabad and Bangalore, which are interested in graduates of the partner institution, are cooperating with the partner institution via additional courses and job offerings for graduating students (see T8.1 and T8.2).

From the point of view of the author of this thesis, the fact that the additional courses besides the curricular courses as well as the placements at the end of the last semester are mandatory for students means that the combination of the implemented curriculum with extracurricular courses offered by companies from the partner institution's environment is understood as a coherent "study program" by both students and teachers.

The local environment of the institution is taken into account on the one hand through various events such as the Independence Day and on the other hand through social responsibility projects, through which the students deal with topics that affect the local environment (see T8.3).

Gap Analysis

The examined institution integrates its environment through curricular and extracurricular activities of students and teachers. Likewise, through collaborations with different companies, the partner institution provides extra-curricular courses that the students would have to attend to. Due to the numerous found links between the institution and its environment, no gap between the design and the reality is found in this dimension. However, the very high importance of these collaborations between companies and the partner institution should be noted, as this contributes significantly to the success and also, according to students, to the enrolments for the curriculum implementation.

5.3 Identified Gaps and Challenges

This section provides an overview of the identified gaps and challenges, followed by a more detailed description of each gap/challenge and which findings of the DRGA they



are based on. Figure 5.2 presents an overview of the findings of each dimension and on which of these findings the identified gaps and challenges are based on.

Figure 5.2: Mapping of dimension findings to gaps and challenges

As it is already mentioned in Section 5.1, the author of this thesis noticed that the DRGA would ignore challenges and problems mentioned by students and teachers unless they are taken into account by the design of the curriculum. However, from the point of view of the author of this thesis, these challenges and problems should also be taken into account in order to find realizable solutions for the gaps. Therefore, they are explained in Section 5.3.2. Similar to the identified gaps of Section 5.3.1, an overview of the identified challenges is provided, followed by a more detailed description of each challenge and the findings of the DRGA the respective challenge is based on. Again, as already mentioned in Section 5.1, if two or more findings from different dimensions were related to each other, they were combined into one gap, such as G1.

5.3.1 Identified Gaps

The gaps that were identified with the help of the DRGA are summarized as follows:

- Gap 1 (G1) Incomplete Student Skills Students do not have the skills they should have according to the curriculum at the end of their studies. Based on DF1 & DF8
- Gap 2 (G2) Unmet Employability Goal Students do not achieve the goal of the designed curriculum that students are equipped with general employability through the implemented curriculum.
 Based on DF1 & DF2
- Gap 3 (G3) Improperly Implemented Teaching Methods The teaching methods broadly formulated by the affiliating university and the affiliated college are not properly implemented. Based on DF3
- Gap 4 (G4) Missing Curriculum Evaluation There is no management structure dedicated solely to the evaluation of the curriculum and to quality assurance as well as quality enhancement of the curriculum implementation. Furthermore, there are no institutional processes in place for quality assurance and quality enhancement. Based on DF4 & DF6
- Gap 5 (G5) Inadequate Teacher Skills Teachers do not have the relevant skills and professional experience to implement the curriculum adequately. Based on DF9

Gap 1 - Incomplete Student Skills

This gap results from the analysis of the differences between design and reality in the dimensions "Objectives and Values" and "Staffing and Skills" (see Sections 5.2.2 & 5.2.8).

In the "Objectives and Values" dimension, it was found that it is of foremost importance to teachers that as many students as possible pass the standardized final examinations and not whether the students actually achieve the course objectives and acquire the respective skills, which was also confirmed by students (see DF1). Additionally, the circumstance that according to teachers (nearly) 100% of the students pass the examinations is contrary to the findings of the "Staffing and Skills" dimension, namely that students do not have the skills that they should have at the end of their studies according to the curriculum (see DF8).

Therefore, based on the findings of the "Objectives and Values" (DF1) as well as the "Staffing and Skills" (DF8) dimensions, students do not have the skills they should have according to the curriculum at the end of their studies.

76

Gap 2 - Unmet Employability Goal

This gap results from the analysis of the differences between design and reality in the "Objectives and Values" dimension.

Although in the "Objectives and Values" dimension it was found that the PSOs are aligned with the objectives of the individual courses, according to the findings in the "Staffing and Skills" dimension, these course objectives are not being met (see DF8). Consequently, the PSOs of the curriculum are not met either (see DF1).

Furthermore, it was also found that the partner institution attempts to achieve the employability goal of the curriculum specifically through extra curricular courses (see DF2. Therefore, the employability goal is not achieved through the curriculum. As described in DF2, it is also noted that the resulting employability of the students is, in the view of the author of this thesis, limited to those companies that carry out the corresponding extra curricular course.

Gap 3 - Improperly Implemented Teaching Methods

This gap is based on findings of the "Processes" dimension 5.2.3. From the point of view of the author of this thesis, the lack of experience of teachers described in Gap 5 (see Section 5.3.1) is one reason for this gap.

Although both the curriculum design and its implementation consider teaching methods such as PBL, they are not properly implemented. As described in the fundamentals in Section 3.2.2, the PBL methodology is particularly committed to analysing and solving a problem independently to a large extent, either alone or in a group. However, it was found that, the implemented interpretation of this teaching method at the partner institution is contrary to the PBL methodology, as students do not work out solutions to the assignment independently and both teachers and students tend to use ready-made solutions (see DF3). In the view of the author of this thesis, this implemented interpretation of PBL is not sufficient for students to actually achieve the intended learning objectives and acquire the respective skills that they should have according to the curriculum. Therefore, the actual implementation of the teaching methodologies constitutes a large gap between design and reality.

Gap 4 - Missing Curriculum Evaluation

The "Processes" as well as the "Management Systems and Structures" dimensions of the curriculum were examined with regard to the evaluation of the curriculum (see Sections 5.2.3 & 5.2.6). It was found that no processes exist to evaluate or revise the implementation of the curriculum (see DF4). Furthermore, no management systems or structures were found that deal specifically with quality assurance and thus the evaluation and revision of curricula (see DF6).

Gap 5 - Inadequate Teacher Skills

In the "Staffing and Skills" dimension it was found that teachers lack both teaching knowledge and professional experience (see Section 5.2.8).

Regarding professional experience, it was found that most of the teachers did not have any experience in a professional setting before becoming a teacher at the partner institution. This lack of professional experience was particularly evident in those situations where the teachers had to review students' projects. It was found that none of the teachers were able to recognize even minor errors or inconsistencies in presumably elaborated small software applications (see DF9).

Furthermore, teachers were not able to properly implement other teaching methods that were not teacher-centered. For example, it was found that teachers in different courses provide students with solutions in advance instead of letting students work out the solutions themselves (see Section 5.3.1.

5.3.2 Identified Challenges of Teachers and Students

The challenges that were identified during the DRGA, which do not fit into a corresponding gap due to a missing design part, but should nevertheless be considered from the author's point of view, are summarized as follows:

- Challenge 1 (C1) Difficulties in Self-Learning The self-learning spaces and computer workstations available are not adequate to allow the majority of students sufficient self-study opportunities outside of course rooms. Based on DF5
- Challenge 2 (C2) Excessive Student Workload Students feel overburdened by their perceived workload. Additionally, their time resources and their satisfaction are not sufficiently taken into account. Based on DF7
- Challenge 3 (C3) Ever-changing Course Assignments Teachers face the challenge of teaching courses about subject areas that are often new to them due to the frequent changes in their teaching assignments. Based on DF10

Challenge 1 - Difficulties in Self-Learning

This challenge results from DF5 found in the "Technology" dimension (see Section 5.2.4): Students find it very difficult to work independently on curriculum-related topics, to study for them or to educate themselves. One reason for this is that only very few students have their own computer or laptop available. Another reason is the lack of space available for students. In addition, students are only allowed to stay outside the campus until a certain time in the evening, which means that potential places outside the campus are also only available to a limited extent.

Challenge 2 - Excessive Student Workload

According the findings of the "Investment Resources" dimension described in Section 5.2.7, another very significant challenge is the workload that students have (see DF7). Due to the combination of curricular and extra-curricular courses, students are usually busy from early in the morning until 7:00 in the evening, with the exception of some smaller breaks and a longer lunch break, continuously for 6 days per week. This also leaves students relatively little time for self-study of content, even if a computer and a suitable study space were available to them.

Challenge 3 - Ever-changing Course Assignments

This challenge concerns the constant reassignment of teachers to courses that was found in the "Staffing and Skills" dimension (see Section 5.2.8). It was found, that teachers are get courses assigned according to the staffing needs of the faculty. However, it frequently happens that teachers with no previous experience in the respective subject areas of a course have to teach this course by themselves. From the point of view of both the teachers and the students, this situation constitutes a new challenge every semester (see DF10).



CHAPTER 6

Closing Gaps & Overcoming Challenges

In this chapter, solutions to reduce or bridge the gaps and challenges that have been identified in Sections 5.3.1 and 5.3.2. Finally, to provide an overview, all the solutions found are listed in Table 6.1, including the gaps and challenges the respective solution is intended to solve and the literature the corresponding solution proposed by the author of this thesis is based on.

6.1 Methodology

For each gap identified in Section 5.3.1, solutions for bridging the design-reality gaps from the author's point of view are developed and categorized into smaller, short-term and easier to implement solutions as well as sustainable, long-term solutions. These solutions are based on literature that deal with the problematic characteristics of the corresponding gap and how to address them. When selecting the literature that the solutions are based on, the author of this thesis aimed for work that has studied the respective problems as closely as possible to the context of this thesis. Since the evaluation of each of these solutions is beyond the scope of this thesis, the solutions are to be considered as solutions proposed by the author of this thesis based on literature and his impressions and experiences, which the author of this thesis gained in the course of the examination of the partner institution. As described in Section 8.1, the proposed solutions shall be evaluated in future work.

6.2 Bridging Gap 1: Incomplete Student Skills

One of the gaps found concerns the difference between the skills that students should have at the end of their studies according to the curriculum and those that they actually have (see **G1**). The skills that students are supposed to learn in the curriculum are assessed at the partner institution almost exclusively with the help of predefined written tests. In their systematic literature review, Sokhanavar et al. found that authentic assessment helps to improve the students' satisfaction and performance and thus to really achieve the teaching goals. It also improves students' reflectiveness and self-awareness as well as soft skills such as critical-thinking and problem-solving skills. In authentic assessment, students engage in several tasks designed to connect their understanding of the knowledge they have learned to real-life situations. Authentic assessment uses tools such as presentations, projects, exhibitions, portfolios, case studies or reflection journals [142].

6.2.1 Equip Students with Hands-On Experience (S1.1, short-term)

As a first step, teachers of laboratory courses are advised to stop presenting the solutions of the practical assignments in advance, so that students can also gain hands-on experiences. Furthermore, in order to help the students to actually achieve the educational goals of the curriculum, the currently periodic assignments should also consider the previously mentioned authentic assessment tools instead of using only tests composed of questions taken directly from the curriculum or previous mid and end exams.

6.2.2 Utilize Authentic Assessment (S1.2, long-term)

The guidelines that are described in Section 6.4 should also include information on methods and tools that enable the authentic assessment just described. This would enable teachers who have no experience in this method to introduce it in their courses. Furthermore, teachers need to be supported in its implementation by the partner institution.

6.3 Bridging Gap 2: Unmet Employability Goal

Among the gaps found, one describes the shortcomings in achieving the general employability goal of the designed curriculum (see G2). With the help of extra-curricular courses, the affiliated college tries to provide students with employability. However, this has the consequence that these are not necessarily in line with the curriculum and the students' workload is increased.

Borah et al. confirm that the form of industry collaboration implemented at the partner institution, which they call course-based collaboration, is widely used in India. Another possibility they mentioned is a project-based collaboration, in which students work out a real-world ICT project as a group of 3 to 4 students either within a semester or an academic year [143]. Projects of this kind are conducted by students at the partner institution, but without industry collaboration. In addition, Borah et al. describe factors that can hinder collaboration, such as the rigidity of a curriculum, although they do not take into account the students' time available in addition to their studies [143]. However, an excessive workload can lead to students studying only superficially or even dropping out of their studies. They may also lose their self-confidence or suffer from anxiety and depression [144]. A distinction can be made between objective and perceived workload. It is important to mention that a reduction of the objective workload does not necessarily lead to a reduction of the perceived workload, but too much perceived workload that causes too much stress has a negative impact on students' academic performance [145, 146]. For example, project-based learning can lead to more hours being spent objectively, but students do not perceive this as an increased workload due to the additional motivation provided by this teaching approach [147].

6.3.1 Involve Reviewers with Professional Experience (S2.1, short-term)

A first, realistically implementable possibility within the existing structures would be that external partners from companies participate in the project review sessions of the ongoing annual and graduation projects described in Section 4.2.2. In this way, they could evaluate the projects together with teachers and provide feedback that also includes a professional point of view. From the point of view of the author of this thesis, this could compensate the lack of professional experience of teachers and thus lead to students receiving a more realistic feedback.

6.3.2 Establish Experts with Professional Experiences as Teachers (S2.2, long-term)

Another and, from the author's point of view, more sustainable possibility would be to combine both forms of collaboration and the mandatory semester projects. As it also described by Borah et al., instead of offering courses, companies could offer real-world projects and work them out with student groups [143]. In a short introductory phase, teachers from the college could teach the theoretical fundamentals, which, however, would also have to be applied by the projects of the external partners. Subsequently, experts from the industry would supervise students in the development of their project. In this way, the theory remains with the college itself and the practical content is taught by external partners. This ensures, on the one hand, that the projects remain within the scope of the curriculum, and, on the other hand, that the projects are supervised by persons who also have professional experience in the relevant subject area of the project. This would free up teaching capacities, but also reduce the overall workload of students, as the extra-curricular courses would no longer be required and the existing annual projects could be used to make students a little more industry-ready, which in turn is in the interest of the companies. Borah et al. also describe additional possibilities for designing a course based on collaboration between companies and an HEI [143, pp. 8-9]

The possibility of combining curricular and extra-curricular courses and the subsequent elimination of the extra-curricular courses currently taking place, would thus already reduce the objective workload of students and teachers. If the goals of the students are taken into account, the possibility of working with a company as part of the already mandatory project would also increase the motivation of the students, which as a consequence would also reduce the perceived workload by students of the partner institution and therefore also tackle the challenge regarding the students' workload (see **C2**).

6.4 Bridging Gap 3: Improperly Implemented Teaching Methods

One of the gaps found is that teaching methods are not implemented as specified by the curriculum and the college (see **G3**). Teachers sometimes come to the partner institution without any prior pedagogical knowledge and teach courses in which they have little or no prior knowledge of the subject areas. However, those who teach certain subjects should already have expert knowledge in the subjects they teach and also have knowledge about pedagogy in general [148]. Teachers should know about the principles of learning and understand how their teaching can help students learn in the best possible way. In particular, teachers should know how students learn and process information in their own individual way. Knowledge of teaching techniques in turn helps to teach students more effectively [149]. Furthermore, assessment methods are important for the learning process to be shaped, as they strongly influence it [150].

6.4.1 Equip Teachers with Pedagogical Basics (S3.1, short-term)

In order to promote the proper usage of adequate teaching and assessment methods and thus swiftly narrow this gap, new teachers would need to be taught the necessary basics of teaching and learning as well as the assessment of the knowledge imparted. Therefore, the college should support teachers by offering pedagogical (online) courses similar to the EduSkills courses mentioned in Section 4.2.4 and encourage as well as providing the time for teachers to continue their education not only in technical subjects.

6.4.2 Develop Teaching Guidelines (S3.2, long-term)

In order to provide teachers with the necessary teaching fundamentals in a manageable, scalable manner, it would be necessary, in the view of the author of this thesis, to create guidelines for teaching at the partner institution. These guidelines should explain the above topics briefly and in a way that is understandable for new teachers and include examples of possible teaching approaches and assessment methods. Nevertheless, this should not replace the above-mentioned courses, because even in the long term, teachers' pedagogical knowledge should be continuously enhanced. This would enable teachers to broaden and deepen their knowledge of teaching, learning and assessment methods. In this way, teachers would be able to continuously improve their teaching guidelines based on the latest academic insights in pedagogy.

Furthermore, the implementation of the curriculum should consider whether and how certain courses can be taught using other teaching methods to further reduce the perceived workload of students and therefore, the respective challenge **C2**. However, the change of teaching method must be accompanied and monitored to prevent students from being overloaded by the change [144]. Additionally, as students are evaluated solely on examinations prescribed and designed by the affiliating university, the change of teaching method should not worsen the student's grades.

6.4.3 Establish a Buddy System for Teachers (S3.3, long-term)

The college should implement a "buddy system" whereby experienced teachers from the college accompany new teachers at the beginning of their employment at the college [151]. In this way, especially teachers who are still at the beginning of their teaching career could be equipped relatively quickly with the necessary basics of teaching and the corresponding teaching methods. According to Eisner, who studied a similar system but in a business school of a public college in New Jersey, both the new and the experienced teachers benefit from such a system [152]. The work of Ghani et al, which was conducted with secondary education science teachers in Malaysia, advises the implementation of accompanied training when establishing such a system to ensure that teachers implement it correctly [153].

6.5 Bridging Gap 4: Missing Curriculum Evaluation

Another of the gaps found describes the lack of structures or systems in place for quality assurance and quality enhancement of the curriculum (see **G4**). It has been found that no appropriate structures at the partner institution are implemented to evaluate curricular and extra-curricular courses as well as teachers. One approach to narrowing the gap would be to establish the aforementioned IQAC. According to Sawant, a highly autonomous IQAC would result in quality standards being met in teaching, learning and evaluation of a curriculum [154]. In particular, students should be involved in the implementation and ongoing development of the curriculum, as their feedback is one of the most effective tools for quality improvement in higher education [155].

Furthermore, the satisfaction of students depends strongly on the places where they can study. Providing a quiet environment where they can work alone or in groups is crucial to increasing student satisfaction [156]. Both Dhaqane & Afrah as well as Mansoor & Ali, who surveyed students from universities in Sri Lanka and Somalia, found a strong positive correlation between student satisfaction and performance [157, 158]. Therefore, it would be essential for the partner institution to provide learning spaces that students are satisfied with in order to bridge this gap and also improve the students' academic performance. However, from the point of view of the author of this thesis, it is important to consider that students need to have the time resources in order to make use of these learning spaces.

Enhance Existing Evaluation Processes (S4.1, short-term) 6.5.1

As a first step, the management should document the current evaluation processes and ensure that they continue to be carried out. Furthermore, the existing feedback forms for students should be extended so that they also take into account the students' satisfaction and perceived stress and workload. This allows the management to take this parameters into account when planning and implementing the curriculum, both in the short and long term. For example, to discover and consequently solve existing problems in courses or to optimize the course of the corresponding semester in the following academic year. In this way, the challenge of the excessive workload of students (see C2) can also be taken into consideration during the implementation of the curriculum, in order to avoid the consequences of an excessive workload that have previously been described in Section 5.3.2.

Additionally, anonymous feedback should also be gathered from teachers in order to consider their resources and needs. For example, they could be asked whether they can manage the time available to them, whether they need further education, or what the college can do to improve the teachers' situation.

Establish Quality Assurance & Enhancement (S4.2, long-term) 6.5.2

In the long term, an IQAC or a similar structure with the same responsibilities should be established that specifically takes care of the topics of quality assurance and quality improvement. On the one hand, this structure can take over the collection of the feedback from students and teachers mentioned above, and on the other hand, it can evaluate the feedback, summarize the findings and formulate concrete, realizable improvements for the management. Additionally, to learn more about a curriculum being implemented at the college, the college should meet regularly with other colleges that are implementing the same curriculum. In this way, the implementing colleges could exchange ideas and experiences about the implementation of the curriculum or compare the results of different measures in order to learn from each other.

In order to also tackle the infrastructural challenge of sparsely available self-learning spaces (see C1), the IQAC should monitor which rooms are unused at which times. This makes it possible, when creating the schedule for the actual implementation of a semester, to plan the courses in such a way that individual rooms can be declared entirely as learning spaces for students. Therefore, by utilizing unused rooms, the supply of learning spaces could be increased by at the latest the next time a semester is planned, without having to make infrastructural changes or build new rooms. In the longer term, there should be spaces that are explicitly declared as learning spaces and are also designed accordingly, in order to sustainably increase the students' satisfaction and thus the students' academic performance. Therefore, these spaces should address the different needs of students, for example individual, small booths to enable quiet learning or larger rooms with sufficient capacity and equipment to enable work in groups.

86

6.6 Bridging Gap 5: Inadequate Teacher Skills

The last of the gaps found describes the teachers' lack of skills that they would actually need in order to be able to teach the curriculum adequately (see **G5**). Chaubey et al. have studied critical factors for good teaching in engineering education in India. According to them, a good teacher must have, among other things, pedagogical skills suitable for teaching the subjects they are supposed to teach. They must also have a basic, in-depth understanding of the subject matter in those very subjects [159]. However, the current practice at the college, where teachers regularly have to teach courses for which they do not have the appropriate pedagogical skills and for which the existing subject knowledge is insufficient, is diametrically opposed to the findings of Chaubey et al.

6.6.1 Maintain Teacher-Course-Assignments (S5.1, short-term)

Keeping the assignment of teachers to courses longer and also to the subject areas of their courses, both teachers and students would experience a higher continuity. Thus, the challenge of frequent changes of courses, which was expressed by both teachers and students, would at least be reduced (see C3). This would also give teachers the chance to build and deepen their knowledge in the subject areas of their courses. Reassignment should then only happen if the particular teacher leaves the college or their wishes change and a reassignment would be possible.

6.6.2 Consider Needs and Interest of Teachers (S5.2, long-term)

New teachers should be recruited considering both their subject areas and interests as well as the courses to be implemented at the college, in order to increase the chance that teachers are willing to teach their assigned courses longer. In addition, the colleges' management must also continuously address the needs of teachers for this purpose. Thus, by addressing teachers' interests, the challenge of frequent changes of their assigned courses would likewise be positively impacted (see **C3**). Rao's work on the needs of teachers working in a college in Andra Pradesh should be considered for this purpose, as the college studied in this thesis is located in the same state and thus examines the interests of teachers in a very similar context [160].

6.6.3 Establish a Handover Protocol (S5.3, long-term)

If frequent changes of assigned courses continue to be necessary, one possibility would be to create a general handover protocol applicable to every course in the curriculum. This could include, for example, possible procedures on how teachers can support their potential successors in taking over the respective course. Additionally, a part of this protocol could be basic material of a course from the curriculum that the affiliating university provides, for example ready-made PowerPoint slides for the course units. This would avoid that every new and possibly inexperienced teacher first has to collect the literature available for him_her in order to be able to prepare his_her courses accordingly. This in turn could, from the point of view of the author of this thesis, not only help to close the gap between the required and actual knowledge of the teachers, but also at least reduce the challenge of frequent changes of assigned courses, if this challenge remains (see C3).

Solution	Description	Time	Gap	Challenge	Based On
01.1		Horizon	01		[1.40]
S1.1	Equip Students with	Short-	GI		[142]
	Hands-On Experience	term			
S1.2	Utilize Authentic Assess-	Long-	G1		[142]
	ment	term			
S2.1	Involve Reviewers with	Short-	G2		[143]
	Professional Experience	term			
S2.2	Establish Experts with	Long-	G2	C2	[143]
	Professional Experiences	term			
	as Teachers				
S3.1	Equip Teachers with Peda-	Short-	G3		[148, 149, 150]
	gogical Basics	term			
S3.2	Develop Teaching Guide-	Long-	G3	C2	[148, 149, 150]
	lines	term			
S3.3	Establish a Buddy System	Long-	G3		[152, 153]
	for Teachers	term			
S4.1	Enhance Existing Evalua-	Short-	G4		[155]
	tion Processes	term			
S4.2	Establish Quality Assur-	Long-	G4	C1	[154]
	ance & Enhancement	term			
S5.1	Maintain Teacher-Course-	Short-	G5	C3	[159]
	Assignments	term			
S5.2	Consider Needs and Inter-	Long-	G5	C3	[159, 160]
	est of Teachers	term			
S5.3	Establish a Handover Pro-	Long-	G5	C3	[159]
	tocol	term			_

Table 6.1: List of proposed solutions including the gaps and challenges the respective solution is intended to solve and the literature it is based on

6.7 Guidance Notes

In this section, guidance notes are presented that have been developed based on identified gaps and the proposed solutions described the previous Sections 5.3.1 and 6. The guidance notes are modelled on those published by EdTech Hub in collaboration with the UNESCO Chair in ICT4D, which aim to provide targeted advice to decision-makers

on their most important specific issues [161]. The guidance notes consist of a short contextual description and then contain highly adaptable recommendations that can be used as goals for decision-makers, combined with examples of how these have already been achieved elsewhere and further readings [162].

The guidance notes are provided for both, curriculum designers and curriculum managers. Each of those contain a description of its target audience that are based on the identified gaps and challenges of **RQ1**, the guidance that is based on the proposed solutions of **RQ2** as well as further readings to enable the target audience to familiarize themselves even more closely with the solutions. As the guidance notes are intended to be presented as independent documents, they are also included as appendices to this thesis (see Appendix D).

As it is described in Section 1.4, the evaluation of the guidance notes' efficacy was outside the scope of this thesis and therefore must be examined by future work (see Section 8.1).

6.7.1 Target Audience

These guidance notes are aimed at HEIs in rural India that are implementing a CS curriculum. However, to indicate possible applicability to other implementing HEIs in a similar context or with similar characteristics, the target audience is defined on the basis of broadly formulated problem statements. If one or more of these problem statements apply to an implementing HEI, it is within the target audience of these guidance notes. The following problem statements were formulated by the author of this thesis and are based on the gaps identified in the course of this thesis (see Section 5.3.1):

- "Students do not meet the goals of the curriculum at the end of their studies", based on **G1**.
- "The implementing institution attempts to ensure student employability exclusively through extra-curricular courses", based on G2.
- "The intended teaching methods and assessment approaches are not executed sufficiently", based on G3.
- "The implementing institution has no dedicated structures to evaluate or improve their implementation of the curriculum", based on ${\bf G4}$.
- "(Potential) Teachers do not have relevant professional experiences or pedagogical skills to implement courses of the curriculum adequately", based on G5.

For curriculum managers, it is described that the respective guidance note is intended for them if their own HEI faces one or more of these problems. For curriculum designers, the guidance note is intended for them if one or more HEIs implementing their CS curricula face one or more of these problems.

6.7.2 Guidance for Curriculum Designers

The guidance note for curriculum designers is intended to help them design the curriculum in a way that enables implementing institutions, despite their challenges, to provide a high-quality study program in which students achieve the goals of the curriculum. In this context, curriculum designers are persons who are responsible for creating or revising a CS curriculum. For example, they define the objectives of a curriculum, the contents of the individual courses, the learning outcomes or the skills that students should have at the end of their studies of the respective curriculum. In the context of the partner institution, these would be the persons at the affiliating university who specify the curriculum taught at the partner institution.

The following guidance is based on the proposed solutions described in Section 6:

- 1. "The designing HEIs should include information about suitable teaching approaches for the courses of the curriculum. Particularly for laboratory courses, the curriculum should include instructions on teaching them: First present the task, then let the students work it out by themselves and answer further questions, if any, and only afterwards present the respective solution", based on **S1.1**
- 2. "The designing HEIs should determine topics for the final project work in a way that the respective topics are not too ambitious, but feasible for the students and assessable for the teachers of the implementing institutions", based on **S1.1**
- 3. "The designing HEIs should include authentic assessment methods and tools for the examinations they prescribe to affiliated HEIs and the end of each semester", based on **S1.2**.
- 4. "The designing HEIs should offer courses for (new) teachers of the partner institutions to equip them with basic pedagogical knowledge. Subsequently, teaching guidelines with pedagogical fundamentals as well as teaching and assessment approaches that are suitable for the courses of the curriculum should be developed and provided to the partner institutions", based on **S3.1** and **S3.2**
- 5. "The designing HEIs should supply curriculum managers with information about possible quality assurance and quality enhancement measurements. Among other things, suitable methods for evaluating the implemented curriculum should be provided on the basis of concrete instructions, for example, ways of evaluating the workload of students and teachers and the possibilities of using these results afterwards", based on **S4.2**.
- 6. "The designing HEIs should provide its implementing institutions with a platform to help them exchange ideas, learn from each other and share Good Practices in a similar context", based on **S4.2**.

7. "The designing HEIs should provide a handover protocol for teachers taking on one or more courses in the curriculum, along with the curriculum. In this regard, basic materials tailored to teachers should be provided for each course", based on **S5.3**.

6.7.3 Guidance for Curriculum Managers

The guidance note for curriculum managers is intended to enable the his_her implementing HEI, despite its challenges, to provide a qualitative study program in which students achieve the goals of the curriculum. In this context, curriculum managers are people who are responsible for implementing a CS curriculum at a HEI. This includes providing the necessary infrastructure, employing and managing teaching staff or ensuring and improving the quality of courses within the scope of their possibilities. In the context of the partner institution, the person ultimately responsible for this would be its director, who can delegate the responsibilities.

- 1. "The implementing HEIs should provide guidelines for teaching that contain basic information about how students learn and possible ways of teaching them. Particularly for laboratory courses, the curriculum management should instruct teachers to first present the task, then let the students work it out by themselves and answer further questions, if any, and only afterwards present the respective solution", based on **S1.1**
- 2. "The implementing HEIs should develop guidelines for the assessment of students beyond the mandatory mid-term and final examinations, including information on authentic assessment and the tools and methods appropriate for it, but also taking into account the workload perceived by students", based on S1.2
- 3. "The implementing HEIs should try to collaborate with industry experts within the curricular courses. The industry experts should bring practical experience to the study program, for example by supervising project reviews or working with teachers in the development and supervision of project work", based on S2.1 and S2.2
- 4. "The implementing HEIs have to equip new teachers with pedagogical basics and provider guidelines for suitable teaching approaches and assessment methods. This could be achieved by offering courses for (new) teachers of the respective institution to equip them with with the contents just mentioned. Additionally, teaching guidelines with pedagogical fundamentals as well as teaching and assessment approaches that are suitable for the courses of the curriculum should be developed", based on S3.1 and S3.2
- 5. A "buddy system" for teachers should be established so that newly hired teachers who have little or no teaching experience are mentored by experienced teachers. Through this system, new teachers are supposed to be integrated into the HEI

from the very beginning and learn from experiences already acquired by their more senior colleagues", based on ${\bf S3.3}$

- 6. "The implementing HEIs should establish a internal structure or body that regularly evaluates the curriculum and reports possible improvements to the curriculum management based on the evaluation results. This evaluation should at least consider feedback on the study program by both students and teachers. Additionally, the perceived and objective workload as well as the satisfaction of students and teachers should be monitored and taken into account", based on S4.1 and S4.2
- 7. "The implementing HEIs should avoid changing teachers' assigned courses frequently. This will allow them to become more proficient in the subject area of their courses and deepen their respective knowledge in it. Thus, both students and teachers could benefit from the higher continuity", based on **S5.1**
- 8. "The implementing HEIs have to ensure that the teachers' interests and needs are addressed in order to prevent the necessity of reassigning courses due to a frequent loss of teachers. Nevertheless, the implementing HEIs should provide a handover protocol to assist newly assigned teachers in taking over a course in a way that minimizes knowledge loss through the handover process", based on S5.2 and S5.3

92

CHAPTER

Discussion of Results

In this chapter, the results of this thesis from Chapter 5 are compared with the objectives of this thesis from Section 1.3. Furthermore, they are compared with the findings of related work described in Chapter 2.

As mentioned in Section 1.3, this thesis aims to provide insights that help to increase the number of SE graduates with sufficient skills. As it is outlined in Chapter 2, the reasons for the insufficient skills of (software) engineering graduates have been studied repeatedly in recent years through similar work. Likewise, numerous suggestions exist in the literature on how to overcome this problem. However, as it is described in Section 1.1, the problem still exists. The reasons why the problem still exists, although it has been known for at least 10 years, are unclear.

Dubey et al. attribute the reasons for the mismatch between the skills of engineering graduates and the needs of the industry to the high privatization of the higher engineering education sector [14]. As described in Section 3.3, the private sector of the higher education system in India largely consists of affiliated colleges, which in turn implement the curriculum of a state university at their own. Furthermore, the related work in the literature examining the existing skill mismatch among engineering graduates did not examine the design of the existing curricula or address the question of whether and to what extent students' actual skills differ from those they should have according to the curriculum.

Therefore, this thesis examines the implementations of a given curriculum using the implementation of 1 of 80 affiliated colleges located in rural India that are implementing the same curriculum. This thesis determines whether and which differences exist between the design of the prescribed curriculum and its actual implementation through answering **RQ1**. Furthermore, solutions to overcome these differences are proposed in the course of this thesis by answering **RQ2** (see Section 1.3). Based on the answers to **RQ1** and **RQ2**, guidance notes (see Appendix D) were developed that summarize the differences

and the proposed solutions for decision makers of the context that was examined by this thesis. However, as described in Section 6.7.1, broadly formulated problem statements were used in describing the target audience for the guidance notes. This attempts to generalize the proposed solutions in order to make them applicable in other contexts, for example, outside of rural India or other curricula.

As already mentioned in the Limitations, neither social nor cultural problems were taken into account in this thesis. Likewise, potential labor policies were not examined (see Section 1.5). From the point of view of the author of this thesis, these factors might have a direct or indirect influence on the implementation of the curriculum. Therefore a root cause analysis of the identified gaps and challenges is noted as possible future work in Section 8.1.

Furthermore, this thesis was developed from an external, European point of view (see Section 1.5). The influence of the local, Indian view is reflected by the answers, which were given in the course of the interviews, whereby these were also analyzed thematically by the author of this thesis. Therefore, this thesis provides a starting point for future work, which shall be carried out in cooperation with local researches on site (see Section 8.1).

From the point of view of the author of this thesis, it is important to note for the evaluation of the proposed solutions that the solutions were not proposed together with stakeholders of the partner institution (see Section 1.5). Therefore, from the point of view of an insider, these are all proposals coming entirely from external sources. In order to include the local perspective in the solutions, one way to evaluate the solutions proposed in this work together with local people would be to use the so-called PDIAtoolkit [163]. This toolkit was developed at the Center for International Development at the Harvard University and is used to analyze problems and develop solutions for them in an iterative process. In future work, the gaps identified in this thesis can be used as the already previously mentioned starting point in a way, that the gaps are the problems to analyze. Through the toolkit, the gaps' root causes shall be analyzed and solutions shall be designed together with local people, which in turn can be compared to the proposed solutions of this thesis.

7.1 Comparing the Identified Gaps & Challenges

The identified gaps listed in Section 5.3.1 between the fixed curriculum and its implementation at a HEI answer **RQ1**. Concurrently, existing challenges of students and teachers were identified that are relevant for solving the underlying problems from the point of view of the author of this thesis, although they are not gaps in the actual sense (see Section 5.3.2). By comparing the identified gaps and challenges found in this thesis with the reasons given in the related work for the skill mismatch in higher engineering education (see Section 2.2), it is found that the reasons largely correspond to findings of this thesis. Table 7.1 provides an overview of the identified gaps and challenges that correspond those aforementioned reasons.
Reason for skill mismatch	Related	Gap /
	Work	Challenge
The almost exclusive use of teacher-centered teaching meth-	[45, 49, 51,	G3
ods and the lack of teaching on practical applicability	50]	
Insufficient structures for quality assurance and quality	[49, 50]	G4
improvement		
Insufficient qualification of the teachers, both from a tech-	[14, 49, 50]	G5
nical and pedagogical perspective		
The partial lack of sufficient infrastructure	[45, 49, 50]	C1

Table 7.1: Related work that correspond to identified gaps / challenges

As mentioned previously, it was beyond the scope of this thesis to evaluate whether the design of the curriculum is capable of equipping students with sufficient SE skills (see Section 1.5). However, as noted in Section 5.2.2, the designed curriculum specifies both theoretical and practical courses on, for example, object-oriented programming or SE. Nonetheless, the objectives of these courses are not met in the implementation of the curriculum (see **G1**). Furthermore, it was found that the affiliating university aims to make the students employable through their curriculum design. As it is also mentioned in Section 1.5, it is not within the scope of this thesis to determine whether the design of the curriculum is sufficient to make students employable. However, it was found that only the extra-curricular courses offered by the partner institution in collaboration with companies make students employable, at least for the companies involved (see **G2**).

7.2 Comparing the Proposed Solutions

The fact that the skill mismatch still exists in spite of the recommendations known from the related work described in Section 2.2.1 can have many reasons from the point of view of the author of this thesis, such as that these recommendations are not known, they are intentionally not taken into account or they are not applicable. The actual reason for this was not found in related work, but to address at least the last of these potential reasons, it was determined for the purpose of this thesis that solutions for solving this problem should be as specific as possible. This, and the involvement of those people who either have to implement these solutions or are directly affected by them, are intended to ensure that the solutions are actually applicable. Therefore, the problems and challenges of the students and teachers that implements a CS curriculum and to whom, consequently, the solutions are directed, were gathered when answering **RQ1**.

The solutions elaborated in Chapter 6 answer **RQ2** and are addressed to those who are responsible for the creation and revision of a curriculum as well as to those who have to implement the curriculum. This includes both the teachers and the management that is responsible for the teachers and ultimately the implementation of the curriculum. From the point of view of the author of this thesis, the recommendations of the related

7. Discussion of Results

work are rather abstract and do not seem to take local realities into account. For this reason, more concrete and realizable solutions were proposed within this thesis, which shall be evaluated by future work (see Section 8.1). Table 7.2 provides an overview of the proposed solutions that correspond to recommendations that have already been proposed in related literature.

Recommendations for skill mismatch	Related	Proposed
	Work	Solution
Shift to student-centered teaching approaches and teach	[15, 46, 49,	S1.1
applicability of theoretical knowledge	51]	
Use methods to also assess skills and not only memorized	[15, 46, 49]	S1.2
knowledge		
Include industry experts with the help of (extra-)curricular	[46, 49, 50]	S2.1, S2.2
courses		
Additional training for teachers	[46, 49, 50]	S3.1
Periodically evaluate the curriculum and its courses	[49, 51]	S4.1, S4.2
Consider incentives for teachers	[50]	S5.2

Table 7.2: Related work that correspond to proposed solutions

In the related work, the continuing education of teachers only addresses professional experiences in their respective fields, but not pedagogical knowledge nor experiences. Likewise, the related work recommends to improve the teaching-learning process by other teaching methods and other examination methods, whereby the focus should be on student-centered approaches. However, they do not mention any concrete examples except for PBL.

Other recommendations of related work were already implemented at the partner institution. As described ins Section 2, Mann et al. recommend additional training for students and teachers. This additional training is realized through extra-curricular courses at the examined partner institution, but according to the students, the mandatory nature of these courses leads to an excessive perceived workload (see C2)

CHAPTER 8

Conclusion and Future Work

One of the major challenges of the Indian higher education system, which is also observed in other countries of the global south, was explained in more detail in the problem statement of this thesis: For years, the skills of the students at the end of their studies do not match the needs of the industry. As it is described in the context of this thesis, the skills in question, as well as reasons and recommendations for this skill mismatch, have been discussed in the past and explained in the literature. One of the reasons for this skill mismatch is the high degree of privatization in the higher education system in India.

In the fundamentals of this thesis, it was described that there are over 38,000 private colleges in India that offer study programs based entirely on the curricula set by their affiliating universities. The colleges themselves are not allowed to make any changes to those curricula and their students are evaluated solely on the basis of standardized tests by that same university. Based on this, this thesis investigated through a Design-Reality Gap Analysis (DRGA) if and which differences exist between a curriculum designed by an affiliating university and its implementations at affiliated colleges. This was done by examining 1 of 80 affiliated colleges implementing the same curriculum of an affiliating university.

The examination of the affiliated college involved the gathering of data about the reality of the students and teachers. Thus, semi-structured interviews were conducted with students and teachers, through which data was collected about their everyday life such as personal objectives, teaching from the points of view of students and teachers or the time resources they need to invest. Additionally, data was collected by observing both students and teachers. For example, teachers were observed during their teaching responsibilities and students were observed during the elaboration of assignments within one of their curricular courses. Afterwards, the gathered data was thematically analyzed along the OPTIMISM dimension used in the DRGA. The results of this thematic analysis then served as the reality part of the DRGA. In addition to the curriculum itself, the design part of the DRGA included information such as the vision and goals of the affiliating university, in order to provide a more complete picture of the curriculum and specifically its goals.

The design was compared with reality for each OPTIMISM dimension of the curriculum and the findings of the gap analysis of each dimension were described. The identified gaps were elaborated based on the dimension findings in the results of this thesis and answer **RQ1** of this thesis. The summarized gaps are as follows:

- 1. Incomplete Student Skills (G1)
- 2. Unmet Employability Goal (G2)
- 3. Improperly Implemented Teaching Methods (G3)
- 4. Missing Curriculum Evaluation (G4)
- 5. Inadequate Teacher Skills (G5)

This thesis aspires to reduce the mismatch between the skills of the graduates and the needs of the industry. For this purpose, solutions for the affiliating universities (curriculum designers) on the one hand and solutions for the affiliated colleges (curriculum managers) on the other hand were developed in order to bridge the identified gaps.

While conducting the DRGA, the author of this thesis found that students and teachers face challenges when implementing the curriculum that, strictly speaking, are not to be considered as identified gaps, because the curriculum does not contain design counterparts for those expressed realities. Nevertheless, from the point of view of the author of this thesis, these should be also taken into account in the development of possible solutions, in order to be able to develop solutions that can actually be realized by the responsible persons. The identified challenges are elaborated in the results of this thesis and are summarized as follows:

- 1. Difficulties in Self-Learning (C1)
- 2. Excessive Student Workload (C2)
- 3. Ever-changing Course Assignments (C3)

For each identified gap, solutions were developed by the author of this thesis using existing literature. The proposed solutions are elaborated in the results of this thesis and answer $\mathbf{RQ2}$ of this thesis. The proposed solutions are summarized as follows:

- 1. Equip Students with Hands-On Experience (S1.1)
- 2. Utilize Authentic Assessment (S1.2)

- 3. Involve Reviewers with Professional Experience (S2.1)
- 4. Establish Experts with Professional Experiences as Teachers (S2.2)
- 5. Equip Teachers with Pedagogical Basics (S3.1)
- 6. Develop Teaching Guidelines (S3.2)
- 7. Establish a Buddy System for Teachers (S3.3)
- 8. Enhance Existing Evaluation Processes (S4.1)
- 9. Establish Quality Assurance & Enhancement (S4.2)
- 10. Maintain Teacher-Course-Assignments (S5.1)
- 11. Consider Needs and Interests of Teachers (S5.2)
- 12. Establish a Handover Protocol (S5.3)

Guidance notes for curriculum designers and curriculum managers were then created on the basis of these proposed solutions. In addition to the details of the proposed solutions, these also contain a description of the target audience for which the solutions are intended, as well as further readings to enable the target audience to familiarize themselves even more closely with the respective topic. The guidance note for the curriculum designers aim to support them in designing a CS curriculum in a way that they support the adequate implementation of the CS curriculum at, for example, affiliated colleges. The guidance note for curriculum managers contains solutions that can be implemented by affiliated colleges. Thus, affiliated colleges do not have to wait for suggestions of improvement by the affiliating university, but they can use the proposed solutions to implement improvements within the scope of their possibilities and responsibilities.

The proposed solutions and the guidance notes were developed by the author of this thesis, but were not evaluated subsequently, because evaluating each of the proposed solutions would have exceeded the scope of this thesis. However, as mentioned previously, literature was used in the development of the proposed solutions that included recommendations for the problematic characteristics of the respective gap for which the proposed solution is intended to be applied. In order to determine whether the proposed solutions are actually suitable for this purpose, future work shall therefore evaluate these proposed solutions.

8.1 Future Work

This section explains what opportunities arise from the results this thesis and what steps can be taken next to move closer to the larger goal behind this thesis: increasing the number of CS graduates with sufficient skills.

8.1.1 Design of the curriculum

This thesis examines the gaps between the implementation of a curriculum and its design. So, in order to contribute to the reduction of the skill mismatch , this thesis assumes that the curriculum is suitable to teach engineering graduates skills that fulfill the needs of the industry in India, since the curriculum examined was designed by a nationally accredited state university. However, from the point of view of the author of this thesis, the curriculum itself should be evaluated in further work to determine whether the structure and content are fundamentally suitable for equipping students with sufficient skills at the end of their studies. The ACM & IEEE guidelines for CS curricula shall be used in the evaluation of the curriculum's design and the skills to be acquired through the curriculum can be compared, for example, with the SWEBOK skills [54, 56]. In this way, the curriculum would be studied in relation to international concepts.

Furthermore, design changes shall be studied in future work. For example, students could be taught self-learning skills at the beginning of their studies, so that they can use online resources to improve their software development skills outside of curricular and extra-curricular courses. Additionally, it shall be examined how an increase in the share of practical laboratory courses affects the students' skills, given that the gap of insufficiently implemented teaching methods has been bridged.

8.1.2 Analyze root-causes of the design-reality gaps

Through the DRGA, gaps between the design and implementations of a curriculum were identified, but also problems and challenges of the students and teachers were found. However, it was beyond the scope of this work to also investigate the actual causes of the particular gaps or challenges. For example, it is not known whether the teaching methods are used incorrectly due to insufficient knowledge or against better knowledge. Therefore, future work shall investigate on the basis of which problems the respective design-reality gap occurs. Additionally, it shall be examined whether and how underlying problems influence each other in order to be able to counter them more effectively.

8.1.3 Evaluate the proposed solutions

The proposed solutions for overcoming gaps between the real implementations of a curriculum and its design are based on literature, but they were nevertheless developed by the author of this thesis.Each of the proposed solutions shall be examined in future work to evaluate whether the respective solution is actually suitable to close its design-reality gap. From the point of view of the author of this thesis, the proposed solutions shall be evaluated at the partner institution of this thesis or at 1 of the other 79 HEIs that are implementing the same CS curriculum.

Create and evaluate teaching guidelines

From the point of view of the author of this thesis, the "Teaching guidelines" mentioned in **S3.2** should be formulated in cooperation with one or more HEIs that are situated in a similar context. These "Teaching Guidelines" may already take into account the challenges of teachers identified in order to make them as applicable as possible for the teachers. Likewise, when developing these teaching guidelines, it should be taken into account that they are intended to help equipping teachers with more basic pedagogical knowledge and to make them familiar with the advantages of student-centered approaches in a way that they actually use them in suitable situations.

Establish a Buddy System for Teachers

One solution proposed in this thesis is the establishment of a buddy system to assist in the onboarding of new teachers, so that they learn sufficient teaching skills as quickly as possible. This proposal is based on two works, each of which investigates a buddy system for teachers, but in different contexts. Therefore a buddy system shall be developed and evaluated at HEIs in rural India.

8.1.4 Generalizability of the identified Design-Reality Gaps

The design-reality gaps were identified by examining the curriculum implementation at 1 of more than 80 HEIs. It is therefore yet to be evaluated whether the same or similar design-reality gaps also occur in the more than 79 other HEIs that implement the same curriculum design. Further work shall therefore examine whether similar gaps are identified in other implementations of the same curriculum. In addition, further work shall examine whether the identified design-reality gaps also occur in implementations of other CS curriculum designs.

8.1.5 Root-causes and impact of computer science brain drain

Both teachers and students confirmed during the interviews that all students strive to move to a major city after graduation, for example, to work as software developers. From the point of view of the author of this thesis, it shall be investigated what the causes and effects of the constant move of computer science graduates from rural India to its major cities are.

8.1.6 Impact of changes in student workload and their necessary time investments

The excessive stress experienced by students and the amount of time resources students devote to curricular and extra-curricular activities is one of the identified challenges. Although this is not part of the DRGA in the strict sense, the author of this thesis considers this challenge as necessary to take a closer look on. Therefore, future work shall examine the effects of reducing the necessary time resources at a HEI that is embedded in

the same or a similar context and whose students require a similarly high time investment. From the author's point of view, both the objective and the perceived workload should be evaluated. Likewise, it shall be examined how alterations of the workload affect the same students.

List of Figures

1.1	Design of this thesis' case study	6
3.1	Accreditation process of the NBA [92, p. 13]	28
4.1	Thematic map of the "Objectives and Values" dimension	40
4.2	Thematic map of the "Processes" dimension	43
4.3	Thematic map of the "Technology" dimension	46
4.4	Thematic map of the "Information" dimension	48
4.5	Thematic map of the "Management Systems and Structures" dimension .	49
4.6	Thematic map of the "Investment Resources" dimension	51
4.7	Thematic map of the "Staffing and Skills" dimension	54
4.8	Thematic map of the "Milieu" dimension	56
5.1	Adapted OPTIMISM model of Bass & Heeks [127, p. 6]	60
5.2	Mapping of dimension findings to gaps and challenges	75

List of Tables

3.1	Accreditation Criteria and Key Indicators (KIs) of the NAAC [97]	27
3.2	Course articulation matrix according to an example by the NBA[102, p. 54][101,	20
3.3	Set of accreditation criteria and Tier-weighting by the NBA [101, 102, p. 14]	29 30
<i>A</i> 1	Overview of the interviewed persons, their role and level of experience	39
4.2	Overview of the conducted observations, their role and level of experience	37
4.3	Extracted codes regarding the "Objectives and Values" dimension	39
4.4	Mapping of themes to codes regarding the "Objectives and Values" dimension	40
4.5	Extracted codes regarding the "Processes" dimension	42
4.6	Mapping of themes to codes regarding the "Processes" dimension	43
4.7	Extracted codes regarding the "Technology" dimension	45
4.8	Mapping of themes to codes regarding the "Technology" dimension	45
4.9	Extracted codes regarding the "Information" dimension	47
4.10	Mapping of themes to codes regarding the "Information" dimension	47
4.11	Extracted codes regarding the "Management Systems and Structures" dimen-	
	sion	48
4.12	Mapping of themes to codes regarding the "Management Systems and Struc-	
	tures" dimension	49
4.13	Extracted codes regarding the "Investment Resources" dimension	50
4.14	Mapping of themes to codes regarding the "Investment Resources" dimension	51
4.15	Extracted codes regarding the "Staffing and Skills" dimension	53
4.16	Mapping of themes to codes regarding the "Staffing and Skills" dimension	53
4.17	Extracted codes regarding the "Milieu" dimension	55
4.18	Mapping of themes to codes regarding the "Milieu" dimension	56
5.1	Example of course overview in the curriculum [133, p. 3]	61
6.1	List of proposed solutions including the gaps and challenges the respective solution is intended to solve and the literature it is based on	88
7.1	Related work that correspond to identified gaps / challenges	95
7.2	Related work that correspond to proposed solutions	96

Acronyms

ACM Association for Computing Machinery. 12, 16, 100

AICTE All India Council for Technical Education. 23, 25, 67

AISHE All India Survey on Higher Education. 23, 24

AQ Austria Agentur für Qualitätssicherung und Akkreditierung Austria. 4

BE Bachelor of Engineering. 10

BTech Bachelor of Technology. 10

- **CS** Computer Science. 4–7, 10, 12, 19, 31–33, 35, 38, 40, 41, 49, 54, 55, 59–62, 67, 72, 73, 89–91, 95, 99, 100
- DRGA Design-Reality Gap Analysis. 6, 31, 35, 38, 59–61, 74–76, 78, 97, 98, 100, 101

HE Higher Education. 2, 9, 10, 12, 13, 25, 53

HEI Higher Education Institution. 2–5, 10–13, 15, 23–26, 29, 30, 32, 40, 44, 53, 67–70, 74, 83, 89–92, 94, 100, 101

ICT Information and communications technology. 25, 82

ICT4D Information and Communications Technology for Development. 60, 88

IEEE Institute of Electrical and Electronic Engineers. 12, 100

IQAC Internal Quality Assurance Cell. 25, 48, 49, 69, 85, 86

JNTUA Jawaharlal Nehru Technological University Anantapur. 31, 56, 63, 64

KI Key Indicator. 26, 27, 62, 104

MPI Multidimensional Poverty Index. 1, 2

- NAAC National Assessment and Accreditation Council. 24–27, 61, 62, 64, 66–71, 73, 74, 104
- NBA National Board of Accreditation. 25, 26, 29, 30, 61, 62, 64, 67–71, 73, 74, 104
- **PBL** Project-based learning. 20–22, 66, 73, 77, 96
- PEO Program Educational Objective. 26, 30
- **PO** Program Outcomes. 29, 30, 62, 63
- PSO Program Specific Outcomes. 29, 30, 62–64, 77
- **SD** Software Development. 10, 17, 20, 125, 126
- **SDG** Sustainable Development Goal. 1, 4
- SDLC Software Development Life Cycle. 11, 15, 17, 18
- SE Software Engineering. 3-5, 9-12, 15-17, 19-22, 54, 72, 93, 95, 125, 126
- SWEBOK Software Engineering Body of Knowledge. 15, 16
- SWECOM Software Engineering Competency Model. 16
- UGC University Grant Commission. 3, 22–25, 68
- **UN** United Nations. 1, 4
- WA Washington Accord. 26

Bibliography

- Francisco H.G. Ferreira et al. "A global count of the extreme poor in 2012: data issues, methodology and initial results". In: *Journal of Economic Inequality* (2016). ISSN: 15738701. DOI: 10.1007/s10888-016-9326-6.
- [4] World Bank Group. Poverty and Shared Prosperity 2020 : Reversals of Fortune. The World Bank, 2020. DOI: 10.1596/978-1-4648-1602-4.
- [7] "Education, development and poverty reduction: A literature critique". In: International Journal of Educational Development (2012). ISSN: 07380593. DOI: 10.1016/j.ijedudev.2012.02.015.
- [8] World Bank. "Learning for All: Investing in People's Knowledge and Skills to Promote Development". In: Investing in People's Knowledge and Skills to Promote Development (2011). ISSN: 0010-4086, 1545-701X.
- [9] Abdul Abdullah, Hristos Doucouliagos, and Elizabeth Manning. "Does education reduce income inequality? A meta-regression analysis". In: *Journal of Economic Surveys* (2015). ISSN: 14676419. DOI: 10.1111/joes.12056.
- [12] Himanshu. "Inequality in India". In: WIDER Working Paper 2019.42 (May 2019).
 DOI: 10.35188/UNU-WIDER/2019/676-0.
- Ganesh Anant Hegde. "Privatization in higher education in India: A reflection of issues". In: A Global Perspective on Private Higher Education. 2016. ISBN: 9780081008980. DOI: 10.1016/B978-0-08-100872-0.00009-4.
- [14] Amlendu Dubey et al. "Reforms in technical education sector: evidence from World Bank-assisted Technical Education Quality Improvement Programme in India". In: *Higher Education* (2019). ISSN: 1573174X. DOI: 10.1007/s10734-018-0343-1.
- [15] Andreas Blom and Hiroshi Saeki. "Employability and Skill Sets of Newly Graduated Engineers in India: Policy Research Paper, The World Bank". In: *IUP Journal of Soft Skills* (2012).
- [17] Government of India. "Concepts and Definitions Used in National Sample Survey (NSS)". In: Golden jubilee publication (2001).
- [18] Vani K. Borooah. "Measuring Inequality of Access to Higher Education in India". In: Journal of Quantitative Economics (2017). ISSN: 0971-1554. DOI: 10.1007/ s40953-016-0053-4.

- [19] A Abdul Salim. "Economic Reforms and Inclusive Growth of Higher Education in India". In: *Productivity (New Delhi)* 55.3 (2014), p. 309. ISSN: 0032-9924.
- [20] Dr. Neena Aneja. "Higher Education in Rural India: Issues and Problems". In: Indian Journal Of Applied Research 5.9 (September 2015). ISSN: 2249-555X.
- [33] Robert K Yin. Case study research and applications : design and methods. Sixth edition. Los Angeles: SAGE, 2018. ISBN: 1506336167.
- [35] Nicola Duell. "Skills shortages and mismatch in the ICT labour market: Challenges for skill development in India, Indonesia and Thailand". In: *Digital Skills Insights* 2019. Ed. by Suella Hansen. CH-1211, Geneva, Switzerland, 2020, pp. 51–59. ISBN: 978-92-61-29331-4.
- [36] Jonathan Lee and Yu Chin Cheng. "Change the face of software engineering education: A field report from Taiwan". In: Information and Software Technology (2011). ISSN: 09505849. DOI: 10.1016/j.infsof.2010.08.005.
- [37] Prantosh Paul et al. "Computing & Allied Engineering Domain in India with Reference to Private Universities: A Case Study of Bachelors Programs". In: International Journal on Recent Researches in Science, Engineering & Technology 5.11 (Nov. 2017), pp. 51–63. ISSN: 2347-6729.
- [45] Kirgi Garg and Vasudeva Varma. "Software Engineering Education in India: Issues and Challenges". In: Conference on Software Engineering Education and Training 21 (Apr. 2008), pp. 110–117. DOI: 10.1109/CSEET.2008.36.
- [46] Kirti Garg and Vasudeva Varma. "People issues relating to software engineering education and training in India". In: Proceedings of the 2008 1st India Software Engineering Conference, ISEC'08. 2008. ISBN: 9781595939173. DOI: 10.1145/ 1342211.1342235.
- [47] Kirti Garg and Vasudeva Varma. "Systemic requirements of a software engineering learning environment". In: ACM International Conference Proceeding Series. 2015. ISBN: 9781450334327. DOI: 10.1145/2723742.2723757.
- [48] K. Kamar Jahan and Christy Selvarani. "Higher Education in India: Issues and Challenges". In: International Conference on Humanities, Literature and Management 15 (18 2015), pp. 81–86. DOI: 10.15242/icehm.ed0115098.
- [49] S. Pavai Madheswari and S. D. Uma Mageswari. "Changing paradigms of engineering education - An Indian perspective". In: *Procedia Computer Science*. 2020. DOI: 10.1016/j.procs.2020.05.034.
- [50] K. S. Mann, M. Narendra Kumar, and Harvinder Singh Saini. "The major challenges for Engineering institutions to fulfill industry demands in India". In: Proceedings of the 2015 IEEE 3rd International Conference on MOOCs, Innovation and Technology in Education, MITE 2015. 2016. ISBN: 9781467367462. DOI: 10.1109/MITE.2015.7375282.

- [51] Rupa Mahanti and P. K. Mahanti. "Software engineering education from indian perspective". In: Proceedings - 18th Conference on Software Engineering Education and Training, CSEE and T 2005. 2005. ISBN: 9780769523248. DOI: 10.1109/ CSEET.2005.30.
- [52] Jeevamol Joy and V. G. Renumol. "Activity oriented teaching strategy for software engineering course: An experience report". In: Journal of Information Technology Education: Innovations in Practice (2018). ISSN: 2165316X. DOI: 10.28945/4116.
- [53] Nurul Ezza Asyikin Mohamed Almi et al. "Software engineering education: The gap between industry's requirements and graduates' readiness". In: ISCI 2011 2011 IEEE Symposium on Computers and Informatics. 2011. ISBN: 9781612846903. DOI: 10.1109/ISCI.2011.5958974.
- [56] Pierre Bourque and Richard E. Fairley. *Guide to the Software Engineering Body* of Knowledge (SWEBOK 2014). 2014. ISBN: 0-7695-2330-7.
- [57] Yvonne Sedelmaier and Dieter Landes. "Software engineering body of skills (SWE-BOS)". In: *IEEE Global Engineering Education Conference*, *EDUCON*. 2014. ISBN: 9781479931910. DOI: 10.1109/EDUCON.2014.6826125.
- [60] IEEE Society. Software Engineering Competency Model 1.0. IEEE, 2014. ISBN: 978-0-7695-5373-3.
- [62] Kevin Forsberg, Rick Adcock, and Alan Faisandier. "Life Cycle Models". In: The Guide to the Systems Engineering Body of Knowledge (SEBoK). Ed. by R.J. Cloutier (Editor in Chief). v 2.4. 2021.
- [64] William E. Richardson. "Undergraduate software engineering education". In: Software Engineering Education. Ed. by Gary A. Ford. New York, NY: Springer New York, 1988, pp. 121–144. ISBN: 978-0-387-34779-0.
- [65] Hamdi Serin. "A Comparison of Teacher-Centered and Student-Centered Approaches in Educational Settings". In: International Journal of Social Sciences & Educational Studies 5.1 (2018), pp. 164–167. ISSN: 24091294. DOI: 10.23918/ijsses.v5i1p164.
- [66] Elvis Munyaradzi Ganyaupfu. "Teaching Methods and Students' Academic Performance". In: International Journal of Humanities and Social Science Invention ISSN (Online) 2.9 (2013), pp. 2319–7722.
- [67] George D Catalano and Karen Catalano. "Transformation: From Teacher-Centered to Student-Centered Engineering Education". In: Journal of Engineering Education 88.1 (1999), pp. 59–64. DOI: 10.1002/j.2168-9830.1999.tb00412.x.
- [68] Orges Cico et al. "Exploring the intersection between software industry and Software Engineering education - A systematic mapping of Software Engineering Trends". In: Journal of Systems and Software (2020). ISSN: 01641212. DOI: 10. 1016/j.jss.2020.110736.

- [69] Philip Johnson. "Design and evaluation of an "athletic" approach to software engineering education". In: ACM Transactions on Computing Education (2019). ISSN: 19466226. DOI: 10.1145/3344273.
- [70] Maíra R. Marques, Alcides Quispe, and Sergio F. Ochoa. "A systematic mapping study on practical approaches to teaching software engineering". In: *Proceedings Frontiers in Education Conference, FIE.* 2015. DOI: 10.1109/FIE.2014.7044277.
- [71] Mauricio R.de A. Souza et al. A systematic mapping study on game-related methods for software engineering education. 2018. DOI: 10.1016/j.infsof.2017.09. 014.
- [72] Lyn Brodie, Hong Zhou, and Anthony Gibbons. "Steps in developing an advanced software engineering course using problem based learning". In: *Engineering Education* (2008). ISSN: 1750-0052. DOI: 10.11120/ened.2008.03010002.
- [73] Paul Gestwicki. "Design and evaluation of an undergraduate course on software development practices". In: SIGCSE 2018 Proceedings of the 49th ACM Technical Symposium on Computer Science Education. 2018. ISBN: 9781450351034. DOI: 10.1145/3159450.3159542.
- [74] Nicolás Martín Paez. "A flipped classroom experience teaching software engineering". In: Proceedings 2017 IEEE/ACM 1st International Workshop on Software Engineering Curricula for Millennials, SECM 2017. 2017. ISBN: 9781538627952. DOI: 10.1109/SECM.2017.6.
- [75] I. V. Semushin et al. "The WHATs and HOWs of maturing computational and software engineering skills in Russian higher education institutions". In: *European Journal of Engineering Education* (2018). ISSN: 14695898. DOI: 10.1080/ 03043797.2017.1385594.
- [76] Heather Wolpert-Gawron. DIY project based learning for math and science. eng. New York: Routledge, 2016. ISBN: 1317486412.
- [77] Juan Valenzuela-Valdes and Pedro Pardo Fernandez. *Project based learning on engineering : foundations, applications and challenges.* eng. Education in a competitive and globalizing world. New York: Nova Publishers, 2015. ISBN: 163482234X.
- [78] Cengiz Günay et al. "Project-Based Learning Improves Critical Thinking for Software Development Students". In: Proceedings of the 20th Annual SIG Conference on Information Technology Education. SIGITE '19. Tacoma, WA, USA: Association for Computing Machinery, 2019, p. 105. ISBN: 9781450369213. DOI: 10.1145/3349266.3351362.
- [79] Maria Lydia Fioravanti et al. "Integrating project based learning and project management for software engineering teaching: An experience report". In: SIGCSE 2018 Proceedings of the 49th ACM Technical Symposium on Computer Science Education. 2018. ISBN: 9781450351034. DOI: 10.1145/3159450.3159599.

TU Bibliothek, Die approbierte gedruckte Originalversion dieser Diplomarbeit ist an der TU Wien Bibliothek verfügbar WIEN Vourknowledge hub. The approved original version of this thesis is available in print at TU Wien Bibliothek.

- [80] Maira Marques et al. "Enhancing the Student Learning Experience in Software Engineering Project Courses". In: *IEEE Transactions on Education* (2018). ISSN: 00189359. DOI: 10.1109/TE.2017.2742989.
- [81] Beatriz Pérez and Ángel L. Rubio. "A Project-Based Learning Approach for Enhancing Learning Skills and Motivation in Software Engineering". In: Proceedings of the 51st ACM Technical Symposium on Computer Science Education. SIGCSE '20. Portland, OR, USA: Association for Computing Machinery, 2020, 309–315. ISBN: 9781450367936. DOI: 10.1145/3328778.3366891.
- [82] Maurício Souza, Renata Moreira, and Eduardo Figueiredo. "Students Perception on the Use of Project-Based Learning in Software Engineering Education". In: *Proceedings of the XXXIII Brazilian Symposium on Software Engineering*. SBES 2019. Association for Computing Machinery, 2019, 537–546. ISBN: 9781450376518. DOI: 10.1145/3350768.3352457.
- [83] Charles Thevathayan. "Evolving Project Based Learning to Suit Diverse Student Cohorts". In: Proceedings of the 22nd International Conference on Evaluation and Assessment in Software Engineering 2018. EASE'18. Association for Computing Machinery, 2018, 133–138. ISBN: 9781450364034. DOI: 10.1145/3210459. 3210472.
- [86] Michael Shattock. The UGC and the Management of British Universities. Society for Research into Higher Education and Open University Press, 1994. ISBN: 0-335-19161-4.
- [103] Virginia Braun and Victoria Clarke. "Using thematic analysis in psychology". In: Qualitative Research in Psychology (2006). ISSN: 14780887. DOI: 10.1191/ 1478088706qp0630a.
- [104] Mark Saunders, Philip Lewis, and Adrian Thornhill. Research Methods for Business Students 5th Ed. 2008. ISBN: 978-0-273-71686-0. DOI: 10.1007/s13398-014-0173-7.2. arXiv: arXiv:1011.1669v3.
- [106] Uma Sekaran and Roger Bougie. Research method for business. 2013. ISBN: 9788578110796. DOI: 10.1017/CB09781107415324.004. arXiv: arXiv: 1011.1669v3.
- [107] Michael J. Healey and Michael B. Rawlinson. "Interviewing business owners and managers: a review of methods and techniques". In: *Geoforum* (1993), pp. 339–355.
 ISSN: 00167185. DOI: 10.1016/0016-7185(93)90026-E.
- Jürgen Bortz and Nicola Döring. Forschungsmethoden und Evaluation in den Sozial- und Humanwissenschaften. Vol. 3. Springer, 2016, pp. 157–165. ISBN: 978-3-642-41088-8. DOI: 10.1007/978-3-642-41089-5. arXiv: 3540419403.
- [120] Raimund Rittnauer. "Enhancing the Resilience of Software Engineering Courses by Implementing Continuous Education using Smartphones exemplified by rural India". Unpublished Bachelor Thesis. TU Wien, 2021.

- Julian M. Bass and Richard Heeks. "Changing Computing Curricula in African Universities: Evaluating Progress and Challenges via Design-Reality Gap Analysis".
 In: The Electronic Journal of Information Systems in Developing Countries (2011).
 ISSN: 1681-4835. DOI: 10.1002/j.1681-4835.2011.tb00341.x.
- [128] Richard Heeks. "Information systems and developing countries: Failure, success, and local improvisations". In: *Information Society* (2002). ISSN: 01972243. DOI: 10.1080/01972240290075039.
- [129] S. I. Dasuki et al. "Evaluating the implementation of international computing curricular in African universities: A design-reality gap approach". In: International Journal of Education and Development using Information and Communication Technology (IJEDICT) (2015).
- [130] Lemma Lessa, Solomon Negash, and Mesfin Belachew. "Steering e-government projects from failure to success: Using design-reality gap analysis as a midimplementation assessment tool". In: International Business: Concepts, Methodologies, Tools, and Applications. 2016. ISBN: 9781466698161. DOI: 10.4018/978-1-4666-9814-7.ch086.
- [131] Hyeri Choi et al. "Rethinking the assessment of e-government implementation in developing countries from the perspective of the design-reality gap: Applications in the Indonesian e-procurement system". In: *Telecommunications Policy* (2016). ISSN: 03085961. DOI: 10.1016/j.telpol.2016.03.002.
- [142] Zahra Sokhanvar, Keyvan Salehi, and Fatemeh Sokhanvar. "Advantages of authentic assessment for improving the learning experience and employability skills of higher education students: A systematic literature review". In: *Studies in Educational Evaluation* 70 (2021), p. 101030. ISSN: 0191-491X. DOI: 10.1016/j. stueduc.2021.101030.
- [143] Dhruba Borah, Khaleel Malik, and Silvia Massini. "Are engineering graduates ready for R&D jobs in emerging countries? Teaching-focused industry-academia collaboration strategies". In: *Research Policy* 48.9 (2019), p. 103837. ISSN: 0048-7333. DOI: 10.1016/j.respol.2019.103837.
- [144] José-Reyes Ruiz-Gallardo, José L González-Geraldo, and Santiago Castaño. "What are our students doing? Workload, time allocation and time management in PBL instruction. A case study in Science Education". In: *Teaching and Teacher Education* 53 (2016), pp. 51–62. ISSN: 0742-051X. DOI: 10.1016/j.tate.2015. 10.005.
- [145] Nudrat Sohail. "Stress and academic performance among medical students." In: Journal of the College of Physicians and Surgeons–Pakistan : JCPSP 23.1 (Jan. 2013), pp. 67–71. ISSN: 1681-7168 (Electronic).
- [146] Christina A Spivey et al. "Longitudinal examination of perceived stress and academic performance of first-year student pharmacists". In: *Currents in Pharmacy Teaching and Learning* 12.9 (2020), pp. 1116–1122. ISSN: 1877-1297. DOI: 10.1016/ j.cptl.2020.05.002.

TU Bibliothek, Die approbierte gedruckte Originalversion dieser Diplomarbeit ist an der TU Wien Bibliothek verfügbar WIEN Vourknowledge hub. The approved original version of this thesis is available in print at TU Wien Bibliothek.

- [147] Eva Kyndt et al. "'Time is not enough.' Workload in higher education: a student perspective". In: Higher Education Research & Development 33.4 (2014), pp. 684–698. DOI: 10.1080/07294360.2013.863839.
- [148] National Research Council. Investigating the Influence of Standards: A Framework for Research in Mathematics, Science, and Technology Education. Ed. by Iris R Weiss et al. Washington, DC: The National Academies Press, 2002. ISBN: 978-0-309-07276-2. DOI: 10.17226/10023.
- [149] L Dee Fink, Susan Ambrose, and Daniel Wheeler. "Becoming a Professional Engineering Educator: A New Role for a New Era". In: *Journal of engineering* education (Washington, D.C.) 94.1 (2005), pp. 185–194. ISSN: 1069-4730.
- [150] Kiomi Matsumoto-Royo and María Soledad Ramírez-Montoya. "Core practices in practice-based teacher education: A systematic literature review of its teaching and assessment process". In: *Studies in Educational Evaluation* 70 (2021), p. 101047. ISSN: 0191-491X. DOI: 10.1016/j.stueduc.2021.101047.
- [152] Susan Eisner. "Onboarding The Faculty: A Model For Win-Win Mentoring". In: *American Journal of Business Education (AJBE)* 8.1 (Dec. 2015), pp. 7–22. ISSN: 1942-2512. DOI: 10.19030/AJBE.V8I1.9013.
- [153] Abdul Ghani et al. "The Effectiveness of Buddy Support System Implementation among Science Teachers: The Case of Malaysia". In: Asian Social Science 6.4 (Mar. 2010), p122. ISSN: 1911-2017. DOI: 10.5539/ASS.V6N4P122.
- [154] D G Sawant. "Role of IQAC in maintaining quality standards in teaching, learning and evaluation". In: *Pacific Science Review B: Humanities and Social Sciences* 2.2 (2016), pp. 66–69. ISSN: 2405-8831. DOI: 10.1016/j.psrb.2016.09.016.
- [155] Lee Harvey. "1 The nexus of feedback and improvement". In: Student Feedback. Ed. by Chenicheri Sid Nair and Patricie Mertova. Chandos Learning and Teaching Series. Chandos Publishing, 2011, pp. 3–26. ISBN: 978-1-84334-573-2. DOI: 10. 1016/B978-1-84334-573-2.50001-X.
- [156] Ronald Beckers, Theo van der Voordt, and Geert Dewulf. "Learning space preferences of higher education students". In: *Building and Environment* 104 (2016), pp. 243-252. ISSN: 0360-1323. DOI: 10.1016/j.buildenv.2016.05.013.
- [157] Uthuma Mansoor and Mohamed Hussain ali Ali. "Impact of Hostel Student's Satisfaction on their Academic Performance in Sri Lankan Universities". In: Proceedings of 5th International Symposium 2015 on Emerging Trends and Challenges in Multidisciplinary Research. South Eastern University of Sri- Lanka, Oluvil, Sri-Lanka, 2015, pp. 43–46.
- [158] Khalif Dhaqane and Nor Afrah. "Satisfaction of Students and Academic Performance in Benadir University". In: *Journal of Education and Practice* 7.24 (2016).
 ISSN: 2222-1735 (Paper), 2222-288X (Online).

- [159] Aabha Chaubey, Bani Bhattacharya, and Shyamal Kumar Das Mandal. "Attributes of good teaching in engineering education in Indian subcontinent". In: Sādhanā 43.11 (2018), p. 188. ISSN: 0973-7677. DOI: 10.1007/s12046-018-0946-3.
- [160] Madasu Bhaskara Rao. "Motivation of teachers in higher education". In: Journal of Applied Research in Higher Education 8.4 (Jan. 2016), pp. 469–488. ISSN: 2050-7003. DOI: 10.1108/JARHE-08-2015-0066.

Web References

- [2] United Nations. Ending Povery. Web. URL: https://www.un.org/en/ sections/issues-depth/poverty/.
- [3] World Health Organization. Coronavirus. URL: https://www.who.int/ health-topics/coronavirus (visited on 01/28/2022).
- [5] United Nations. Transforming our world: The 2030 Agenda for Sustainable Development. A/RES/70/1. 2015. URL: sustainabledevelopment.un.org (visited on 02/04/2020).
- [6] Sabina Alkire et al. Global Multidimensional Poverty Index 2020. 2020. URL: http://hdr.undp.org/en/2020-MPI (visited on 10/14/2020).
- [10] World Bank Group. World Bank Education Overview : Higher Education. Washington, D.C. URL: http://documents.worldbank.org/curated/en/ 610121541079963484/World-Bank-Education-Overview-Higher-Education (visited on 10/20/2020).
- [11] Sabina Alkire et al. Global Multidimensional Poverty Index 2019. 2019. URL: http://hdr.undp.org/en/2019-MPI (visited on 10/14/2020).
- [16] Varun Aggarwal, Siddharth Nithyanand, and Malvika Sharma. "National Employability Report - Engineers". In: (2019). URL: https://www.aspiringminds. com/research-reports/national-employability-report-forengineers-2019/ (visited on 05/30/2020).
- [21] University Grants Commission. Annual Report 2018-2019. New Delhi, India. URL: https://www.ugc.ac.in/page/Annual-Report.aspx (visited on 11/08/2020).
- [22] Bureau of Planning, Monitoring and Statistics, Department of Higher Education, Ministry of Education (formerly Human Resource Development), Government of India. All Indian Survey on Higher Education Instruction Manual. 2020. URL: http://aishe.nic.in/aishe/viewDocument.action?documentId= 271 (visited on 12/06/2020).
- [23] Centre for Civil Society. Policy Brief: Regulatory Structure of Higher Education in India. URL: https://ccs.in/sites/default/files/research/ policy-brief-regulatory-structure-higher-education.pdf (visited on 11/08/2020).

- [24] TU Wien Informatics. MSc Business Informatics. URL: https://informatics. tuwien.ac.at/master/business-informatics/ (visited on 02/14/2022).
- [25] Rechtsinformation des Bundes. Gesamte Rechtsvorschrift für Universitätsgesetz 2002, Fassung vom 09.02.2022. translated by the author of this thesis. URL: https://www.ris.bka.gv.at/GeltendeFassung.wxe?Abfrage= Bundesnormen&Gesetzesnummer=20002128 (visited on 02/14/2022).
- [26] Technische Universität Wien. Studienplan (Curriculum) für das Masterstudium Business Informatics. 2019. URL: https://www.tuwien.at/fileadmin/ Assets/dienstleister/studienabteilung/MSc_Studienplaene_ 2019/Masterstudium_Business_Informatics.pdf (visited on 03/15/2020).
- [27] AQ Austria. About AQ Austria. URL: https://www.aq.ac.at/de/ueberuns/ (visited on 02/14/2022).
- [28] European Association for Quality Assurance in Higher Education. About EQAF. URL: https://www.enqa.eu/events/2021-online-european-qualityassurance-forum-eqaf/ (visited on 02/14/2022).
- [29] AQ Austria. Gutachten zum Verfahren zur Akkreditierung des FH-Masterstudiengangs "Data and Information Science", StgKz 0807, am Standort Graz des Erhalters FH Joanneum GmbH. URL: https://www.aq.ac.at/de/akkreditiertehochschulen-studien/dokumente-entscheidungen-fh/Gutachten_ A0807_Data-and-Information-Science_Veroeffentlichen.pdf (visited on 02/14/2022).
- [30] AQ Austria. Gutachten zum Verfahren zur Akkreditierung des FH-Bachelorstudiengangs "Wirtschaftsinformatik & Digitale Transformation", StgKz 0842, am Standort Urstein der Fachhochschule Salzburg GmbH. URL: https://www.aq.ac.at/de/ akkreditierte-hochschulen-studien/dokumente-entscheidungenfh/Erh011_FHS_A0842_GA_Veroeff_20180920.pdf (visited on 02/14/2022).
- [31] AQ Austria. Gutachten zum Verfahren zur Akkreditierung des Bachelorstudiengangs "Digital Business Management" der Privatuniversität Schloss Seeburg. Expert review is not published because application was withdrawn after rejection. URL: https://www.aq.ac.at/de/akkreditierte-hochschulenstudien/entscheidungen_pu.php (visited on 02/14/2022).
- [32] AQ Austria. Gutachten zum Verfahren zur Akkreditierung des Universitätslehrgangs "Master of Advanced Studies in Digitalisierung" der Privatuniversität Schloss Seeburg. Expert review is not published because application was withdrawn after rejection. URL: https://www.aq.ac.at/de/akkreditierte-hochschulenstudien/entscheidungen_pu.php (visited on 02/14/2022).
- [38] CollegeDekho. Colleges in India. URL: https://www.collegedekho.com/ colleges-in-india/ (visited on 03/01/2022).
- [39] CollegeDekho. B. Tech Computer Science Colleges in India. URL: https://www. collegedekho.com/btech-computer_engineering-colleges-inindia/ (visited on 03/01/2022).

- [40] CollegeDekho. B. Tech Software Engineering Colleges in India. URL: https:// www.collegedekho.com/btech-software_engineering-collegesin-india/ (visited on 03/01/2022).
- [41] CollegeDekho. B. Tech Software Development Colleges in India. URL: https:// www.collegedekho.com/btech-software_development-collegesin-india/ (visited on 03/01/2022).
- [42] collegedunia. List of Top Colleges in India Based on 2022 Ranking. URL: https: //collegedunia.com/india-colleges (visited on 03/01/2022).
- [43] collegedunia. List of Top BE/B. Tech in Computer Science Colleges in India Based on 2022 Ranking. URL: https://collegedunia.com/btech/computerscience-colleges (visited on 03/01/2022).
- [44] collegedunia. List of Top BE/B. Tech in Software Engineering Colleges in India Based on 2022 Ranking. URL: https://collegedunia.com/btech/ software-engineering-colleges (visited on 03/01/2022).
- [54] Joint Task Force on Computing Curricula of ACM & IEEE Computer Society. Computer Science Curricula 2013. Curriculum Guidelines for Undergraduate Degree Programs in Computer Science. Feb. 2015. URL: https://www.acm. org/binaries/content/assets/education/cs2013_web_final.pdf (visited on 04/06/2022).
- [55] Joint Task Force on Computing Curricula of ACM & IEEE Computer Society. Software Engineering 2014. Curriculm Guidelines for Undergraduate Degree Programs in Software Engineering. 2015. URL: https://www.acm.org/binaries/content/assets/education/se2014.pdf (visited on 02/05/2020).
- [58] Merriam-Webster Dictionary. *Essential Meaning of skill*. URL: https://www.merriam-webster.com/dictionary/skill (visited on 02/11/2022).
- [59] Cambridge Dictionary. *Meaning of skill in English*. URL: https://dictionary.cambridge.org/dictionary/english/skill (visited on 02/11/2022).
- [61] Accreditation Board for Engineering and Inc. Technology. *Accreditation*. URL: https://www.abet.org/accreditation/ (visited on 10/08/2021).
- [63] IEEE Standards Association. IEEE Standard 828-2012. URL: https://standards. ieee.org/standard/828-2012.html (visited on 12/08/2020).
- [84] University Grants Commission. Genesis. URL: https://www.ugc.ac.in/ page/Genesis.aspx (visited on 12/06/2020).
- [85] UK Public General Acts. Indian Independence Act. 1947. URL: https://www. legislation.gov.uk/ukpga/Geo6/10-11/30/contents (visited on 12/06/2020).
- [87] All India Council for Technical Education (AICTE). *About AICTE*. URL: https: //www.aicte-india.org/about-us/history (visited on 12/08/2020).

- [88] Amandeep Shukla. Bill to merge AICTE, UGC in final stages. URL: https:// www.hindustantimes.com/india-news/bill-to-merge-aicte-ugcin-final-stages/story-30Q8YPFcgzLBq40oUQgTsK.html (visited on 12/08/2020).
- [89] Legislative Department. The General Clauses Act. 1897. URL: https://www. indiacode.nic.in/bitstream/123456789/2328/1/189710.pdf (visited on 08/18/2021).
- [90] University Grants Commission. UGC (Affiliation of Colleges by Universities) Regulations 2009. URL: https://www.ugc.ac.in/oldpdf/regulations/ AffiliationofCollegesbyUniversitiesregulation.pdf (visited on 11/08/2020).
- [91] National Assessment and Accreditation Council. *About NAAC*. 2020. URL: http: //naac.gov.in/index.php/about-us (visited on 02/22/2021).
- [92] National Board of Accreditation (NBA). General Manual for Accreditation. New Delhi, 2019. URL: https://www.nbaind.org/publications (visited on 02/22/2021).
- [93] University Grants Commission. UGC Quality Mandate. URL: https://www.ugc. ac.in/pdfnews/9470862_letter-to-VCs-reg-Quality-Mandate. pdf (visited on 03/21/2021).
- [94] NAAC. Units of assessment. URL: http://naac.gov.in/index.php/ assessment-accreditation#units (visited on 03/01/2021).
- [95] NAAC. Eligibility criteria. URL: http://naac.gov.in/index.php/ assessment-accreditation#eligibility (visited on 03/02/2021).
- [96] NAAC. Internal Quality Assurance Cell. URL: http://naac.gov.in/index. php/info-for-institutions#iqac (visited on 03/16/2021).
- [97] NAAC. Annexure 1: Distribution of weightage across Key Indicators (KIs). URL: http://naac.gov.in/images/docs/Key-Indicators-and-Weightages. pdf (visited on 03/01/2021).
- [98] Internationl Engineering Alliance. Signatories of the Washington Accord (WA). URL: https://www.ieagreements.org/accords/washington/signatories/ (visited on 12/08/2020).
- [99] European commission. The Bologna Process and the European Higher Education Area. URL: https://ec.europa.eu/education/policies/highereducation/bologna-process-and-european-higher-educationarea_en (visited on 02/22/2021).
- [100] International Engineering Alliance. 25 Years Washington Accord. New Zealand, 2014. URL: https://www.ieagreements.org/accords/washington/ (visited on 02/22/2021).

TU **Bibliothek**, Die approbierte gedruckte Originalversion dieser Diplomarbeit ist an der TU Wien Bibliothek verfügbar wien vour knowledge hub. The approved original version of this thesis is available in print at TU Wien Bibliothek.

- [101] National Board of Accreditation (NBA). Manual for Accreditation of Undergraduate Engineering Programs (Tier I Institutions). New Delhi, 2019. URL: https: //www.nbaind.org/publications (visited on 02/22/2021).
- [102] National Board of Accreditation (NBA). Manual for Accreditation of Undergraduate Engineering Programs (Tier II Institutions). New Delhi, 2019. URL: https://www.nbaind.org/publications (visited on 02/22/2021).
- [105] Zoom Software. URL: https://zoom.us/ (visited on 09/09/2021).
- [109] Slack Software. URL: https://slack.com/ (visited on 09/09/2021).
- [110] Covid19 Statistics of Andra Pradesh. URL: https://www.covid19india. org/state/AP/ (visited on 10/08/2021).
- [111] Instragram Signup. URL: https://www.instagram.com/accounts/emailsignup/ (visited on 12/08/2021).
- [112] Selenium Framework. URL: https://www.selenium.dev (visited on 12/08/2021).
- [113] Marcus Atish D Rozario. Project report Employee Database and Payroll Management System. URL: https://www.rcciit.org/students_projects/ projects/mca/2018/GR5.pdf (visited on 01/12/2022).
- [114] FreeProjectzw. Payroll Management System Project Download Project Source Code And Database. URL: https://www.freeprojectz.com/paid-projects/ php-mysql/payroll-management-system-mini-project (visited on 01/12/2022).
- [115] Angel Jude Reyes Suarez. Payroll Management System Project in PHP With Source Code. URL: https://itsourcecode.com/free-projects/phpproject/payroll-system-php/ (visited on 01/12/2022).
- [116] Anuj Kumar. Hostel Management System in PHP. URL: https://phpgurukul. com/hostel-management-system/ (visited on 01/12/2022).
- [117] Projectworlds. Hostel Management System Project in PHP. URL: https:// projectworlds.in/free-projects/php-projects/hostel-managementsystem-project-in-php/ (visited on 01/12/2022).
- [118] Scribd. Results for "payroll management system". URL: https://www.scribd. com/search?content_type=documents&page=1&query=payrollmanagementsystem (visited on 01/12/2022).
- [119] Scribd. Results for "hostel management system". URL: https://www.scribd. com/search?content_type=documents&page=1&query=hostelmanagementsystem (visited on 01/12/2022).
- [121] YouTube Video. URL: https://youtube.com/ (visited on 09/09/2021).
- [122] National Programme on Technology Enhanced Learning. URL: https://nptel. ac.in/ (visited on 09/09/2021).
- [123] EduSkills Foundation. URL: https://eduskillsfoundation.org/ (visited on 09/09/2021).

- [124] EduSkills AWS Academy. URL: https://eduskillsfoundation.org/ index.php/aws-academy/ (visited on 09/09/2021).
- [125] EduSkills Cisco Academy. URL: https://eduskillsfoundation.org/ index.php/cisco-academy/ (visited on 09/09/2021).
- [126] SSE. National Service Scheme at SSE. URL: https://sseptp.org/nss/ (visited on 12/08/2021).
- [132] JNTUA. First Year B.Tech Course Structure and Syllabi under R20 Regulations. URL: https://www.jntua.ac.in/wp-content/uploads/2021/02/ JNTUA-R20-B.Tech-CSE-Course-Structure-2020-21..pdf (visited on 10/07/2021).
- [133] JNTUA. Second Year B.Tech Course Structure and Syllabi under R20 Regulations. URL: https://www.jntua.ac.in/wp-content/uploads/2021/ 09/II-B.Tech_.-Computer-Science-Engineering-R20-Course-Structure-Syllabi.pdf (visited on 10/07/2021).
- [134] JNTUA. Third and Fourth Year B. Tech Course Structure and Syllabi under R15 Regulations.
- [135] JNTUA. Front page of the JNTUA website. URL: https://www.jntua.ac.in/ (visited on 03/09/2021).
- [136] JNTUA. Methodology. URL: https://www.jntua.ac.in/methodology/ (visited on 03/09/2021).
- [137] SSE. Vision and Mission. URL: http://sseptp.org/vision-and-mission/ (visited on 03/09/2021).
- [138] SSE. Computer science & engineering department (SOSO). URL: http:// sseptp.org/computer-science-eng/ (visited on 01/07/2022).
- [139] JNTUA. Permant Affiliation Guidelines w.e.f the Academic Year 2021-22. URL: https://www.jntua.ac.in/wp-content/uploads/2021/08/JNTUA-P ermanent-Affiliation-Guidelines-2021.pdf (visited on 09/23/2021).
- [140] SSE. Facilities. URL: http://sseptp.org/facilities/ (visited on 03/09/2021).
- [141] JNTUA. Second Year B. Tech Course Structure and Syllabi under R19 Regulations. URL: https://dap.jntua.ac.in/wp-content/uploads/2020/05/ CSE-R19_2nd-Year-Course-Structure-Syllabus.pdf (visited on 10/07/2021).
- [151] Merriam-Webster Dictionary. Definition of buddy system. URL: https://www. merriam-webster.com/dictionary/buddysystem (visited on 02/11/2022).
- [161] EdTech Hub. About EdTech Hub. URL: https://edtechhub.org/aboutedtech-hub/ (visited on 11/05/2021).

- [162] Tim Unwin et al. "Guidance Note Introduction". In: Education for the most marginalised post-COVID-19: Guidance for governments on the use of digital technologies in education (2020). URL: https://edtechhub.org/wp-content/ uploads / 2020 / 09 / Education - for - the - most - marginalised - R eport-Guidance-note-Introduction-v8.pdf.
- [163] Salimah Samji et al., eds. PDIAtoolkit A DIY Approach to Solving Complex Problems. Version 1.0. Oct. 2018.



Appendix A

Questions of the semi-structured interview

A.1 Teachers

Objectives and values

- From your point of view, what are the strengths of the CSE study program?
- From your point of view, what are the weaknesses of the CSE study program?
- Do your students learn sufficient skills for future work environment?
- Would you like to share anything else?
- Would you like to ask me anything?

Processes

- How do you design your courses?
- How do you prepare for your courses?
- How do you teach your students? (Which teaching approaches do you use?)
- How do you evaluate/assess your students?
- Are there other assessment methods apart from the JNTUA tests?
- How do you assure the quality of your course?
- How do you evaluate your course?
- Do you gather feedback of your students about the course? If yes, how?

Technology

- Which infrastructure can you use for your courses?
- Which infrastructure can you use to prepare your courses?
- From your point view, is the infrastructure suitable for the CSE study program?

Information

- How do you select the study material you use for your classes?
- How do teachers exchange information with each other?

Management systems and structures

- Are you allowed to make changes on the curriculum? If you would be allowed to change the curriculum (for your course), what would you change?
- What are the further education/training opportunities for teachers of your institute?
- (How do you further educate or train yourself?)

Investment resources

- How much time do you have to prepare your classes?
- How much time to you spend in a class room?
- What are you working on besides your teaching responsibilities?

Staffing and skills

- Which subjects do you teach?
- Which experiences do you have regarding the subjects you teach?
- Which teaching approaches do you know about?

Milieu

• How does the local industry influence your courses?

A.2 Students

• In which semester and year are you?

Objectives and values

- What do you like about your study program?
- What do you not like about your study program?
- Would you like to share anything else?
- Would you like to ask me anything?
- What role does creativity play for you in your studies?
- What are your plans after graduation?

Processes

- How do you participate during SD/SE courses?
- How do your teachers teach their SD/SE courses?
- How have you been evaluated in your visited SD/SE courses?
- How can you give feedback about your visited SD/SE courses?

Technology

- Which technologies (e.g. computers, phones, etc.) are you using for your SD/SE courses?
- Which technologies in use do you own?

Information

• How do you get to know about offerings of co- or extra-curricular activities of the college?

Management systems and structures

- How can you participate in the development of the CSE study program?
- How can you participate in the development of the taught courses?
- If you are allowed to change the study program, what would you change?
- If you are allowed to change the way your teacher works with you, what would you change?
- If you are allowed to change the curriculum (for your courses), what would you change?

Investment resources

- What year and semester are you currently in?
- How many hours do you visit lectures per week?
- How many hours do you study per week?

Staffing and skills

- How do you apply learned (theoretical) knowledge of the SD/SE courses you have visited?
- What did you learn in your visited placement trainings?

Milieu

• What do you know about the job market in which you would participate after you graduate?

APPENDIX **B**

Course Description of "Advanced Data Structures and Algorithms"

course code	Advanced Data Structure	Advanced Data Structures & Algorithms		Т	Ρ	С
20A05301T			3 0 0		3	
Pre-requisite	Data Structures	Semester				
Course Objecti	ves					
Learn a	symptotic notations, and analyze the	performance of differen	t algorit	hms.		
 Unders 	tand and implement various data stru	ctures.				
•						
Course Outcon	nes					
After completio	on of the course, students will be able	to				
Analyze	e the complexity of algorithms and app	ly asymptotic notations	i.			
 Apply r 	non-linear data structures and their op	erations.				
•						
				9 hrs		
Unit-I Introduction to Algorithms, Pse complexity, Asy	Introduction to Algorithms Algorithms: eudocode for expressing algorithms, Pe ymptotic	erformance Analysis-Spa	ace com	g plexit	hrs y, Tir	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II	Introduction to Algorithms Algorithms: eudocode for expressing algorithms, Pe ymptotic Trees Part-I	erformance Analysis-Spa	ace com	plexit	hrs y, Tir	me
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II	Introduction to Algorithms Algorithms: eudocode for expressing algorithms, Per ymptotic Trees Part-I	erformance Analysis-Spa	ace com	g plexit	hrs y, Tir hrs	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks	Introduction to Algorithms Algorithms: audocode for expressing algorithms, Pe ymptotic Trees Part-I	erformance Analysis-Spa	ace com	9 plexit	hrs y, Tir hrs	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu	Introduction to Algorithms Algorithms: audocode for expressing algorithms, Pe ymptotic Trees Part-I res and algorithms: Concepts, Techniq	erformance Analysis-Spa	ace com	plexit	hrs y, Tir hrs	me
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu 2. Fundamenta	Introduction to Algorithms Algorithms: audocode for expressing algorithms, Perymptotic Trees Part-I res and algorithms: Concepts, Techniq Is of Computer Algorithms, Ellis Horow	erformance Analysis-Spa ues and Applications, G ritz, Sartaj Sahni and Raj	ace com A V Pai. asekhar	plexit	hrs ry, Tir hrs Galgo	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu 2. Fundamenta publications Pv	Introduction to Algorithms Algorithms: audocode for expressing algorithms, Perymptotic Trees Part-I res and algorithms: Concepts, Techniq Is of Computer Algorithms, Ellis Horow t. Ltd.	erformance Analysis-Spa ues and Applications, G ritz, Sartaj Sahni and Raj	ace com A V Pai. asekhar	plexit	hrs y, Tir hrs Galgo	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu 2. Fundamenta publications Pv Reference Bool	Introduction to Algorithms Algorithms: audocode for expressing algorithms, Perymptotic Trees Part-I res and algorithms: Concepts, Techniq Is of Computer Algorithms, Ellis Horow t. Ltd. ks	erformance Analysis-Spa ues and Applications, G ritz, Sartaj Sahni and Raj	ace com A V Pai. jasekhar	plexit	hrs y, Tir hrs Galgo	me
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu 2. Fundamenta publications Pv Reference Bool 1. Classic Data	Introduction to Algorithms Algorithms: audocode for expressing algorithms, Perymptotic Trees Part-I res and algorithms: Concepts, Techniq Is of Computer Algorithms, Ellis Horow t. Ltd. ks Structures by D. Samanta, 2005, PHI	erformance Analysis-Spa ues and Applications, G ritz, Sartaj Sahni and Raj	ace com A V Pai. jasekhar	splexit	hrs y, Tir hrs Galgo	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu 2. Fundamenta publications Pv Reference Bool 1. Classic Data 2. Design and A	Introduction to Algorithms Algorithms: Eudocode for expressing algorithms, Per ymptotic Trees Part-I res and algorithms: Concepts, Techniq Is of Computer Algorithms, Ellis Horow t. Ltd. ks Structures by D. Samanta, 2005, PHI unalysis of Computer Algorithms by Ah	erformance Analysis-Spa ues and Applications, G ritz, Sartaj Sahni and Raj p, Hopcraft, Ullman 199	A V Pai. asekhar 8, PEA.	s plexit ٤ am, (hrs y, Tir Bhrs	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu 2. Fundamenta publications Pv Reference Bool 1. Classic Data 2. Design and A 3. Introduction	Introduction to Algorithms Algorithms: Endocode for expressing algorithms, Per ymptotic Trees Part-I res and algorithms: Concepts, Techniq Is of Computer Algorithms, Ellis Horow t. Ltd. ks Structures by D. Samanta, 2005, PHI unalysis of Computer Algorithms by Ah to the Design and Analysis of Algorithm	erformance Analysis-Spa ues and Applications, G ritz, Sartaj Sahni and Raj p, Hopcraft, Ullman 199 ns by Goodman, Hedet	A V Pai. asekhar 8, PEA. niemi, T	s و العربي و MG.	hrs y, Tir Ehrs	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu 2. Fundamenta publications Pv Reference Bool 1. Classic Data 2. Design and A 3. Introduction Online Learning	Introduction to Algorithms Algorithms: audocode for expressing algorithms, Perymptotic Trees Part-I res and algorithms: Concepts, Techniq Is of Computer Algorithms, Ellis Horow t. Ltd. ks Structures by D. Samanta, 2005, PHI unalysis of Computer Algorithms by Ah to the Design and Analysis of Algorithm g Resources	erformance Analysis-Spa ues and Applications, G ritz, Sartaj Sahni and Raj p, Hopcraft, Ullman 199 ns by Goodman, Hedetr	A V Pai. asekhar 8, PEA. niemi, T	gplexit	hrs y, Tir Bhrs Galgo	ne
Unit-I Introduction to Algorithms, Pse complexity, Asy Unit-II Textbooks 1. Data Structu 2. Fundamenta publications Pv Reference Bool 1. Classic Data 2. Design and A 3. Introduction Online Learning https://www.to	Introduction to Algorithms Algorithms: Endocode for expressing algorithms, Per ymptotic Trees Part-I res and algorithms: Concepts, Techniq Is of Computer Algorithms, Ellis Horow t. Ltd. ks Structures by D. Samanta, 2005, PHI unalysis of Computer Algorithms by Ah to the Design and Analysis of Algorithm g Resources utorialspoint.com/advanced data strue	erformance Analysis-Spa ues and Applications, G ritz, Sartaj Sahni and Raj p, Hopcraft, Ullman 199 ns by Goodman, Hedetr ictures/index.asp	A V Pai. asekhar 8, PEA. niemi, T	gplexit	hrs y, Tir Bhrs	ne



APPENDIX C

Feedback Form

Name of Teacher				
Department				
Subject and Subject Code				
Year				
Knowledge of teacher on the subject	Excellent	Good	Fair	Poor
Clarity and understandability of	Excellent	Good	Fair	Poor
teacher's explanation				
Teacher's willingness to help	Excellent	Good	Fair	Poor
Approximate percentage of classes not	<10%	10% to $25%$	>25%	6
engaged by teacher in the subject				
Whether the teacher dictates note only	No		Yes	
without explanation				
Teacher's ability to organize lectures	Excellent Satisfactory		Inadequate	
Speed of presentation	Just Right	Too Fast	Too Slow	
Does the teacher encourage questioning	Yes	Sometimes	No	
Behaviour of teacher	Pleasant	Indifferent	Unpleasant	
Sincerity of teacher	Sincere	Not Sincere	Unable to judge	
Overall teaching effectiveness of the	Excellent	Good	Fair	Poor
teacher				
Indicate briefly strengths and weakness				
of teacher				
Any other relevant information (Man-				
nerism, peculiarities etc. of teacher)				


APPENDIX D

Guidance Notes

131



Guidance Note for Curriculum Designers in rural India

Computer Science and Engineering (CSE) Curriculum

Developed by: Lukas Bürstmayr, Paul Spiesberger, Raoul Vallon & Thomas Grechenig

August 16, 2022

This work is licensed under a Creative Commons "Attribution-ShareAlike 4.0 International" license.



1 Context

A challenge of the Indian higher education system that exists for more than a decade is a mismatch between the skills of the engineering graduates and the needs of the industry, especially in the field software engineering. The reasons and recommendations for this skill mismatch have been discussed in the past and explained in the literature and include the high privatization of higher engineering education. The majority of the private higher education sector is composed of over 38.000 affiliated colleges that implement study programs based entirely on the curricula set by their affiliating universities. The majority of these affiliated colleges are located in rural India. The authors of this Guidance Note examined 1 of 80 affiliated colleges in rural India implementing the same CSE curriculum of an affiliating university [1].

Based on the findings of this study, solutions were proposed by the authors that aim to bridge the respective gap, using existing literature and keeping challenges of students and teachers in mind. Based on these proposed solutions this Guidance Note was created that aims to enable curriculum designers to apply some of these proposed solutions.

Caveat: Solutions to problems differ and heavily depend on the context of each Higher Education Institution (HEI). The here stated solutions can only be seen as proposals and require constant local evaluation.

1.1 CSE Curriculum Designers

CSE Curriculum designers are persons who are responsible for creating or revising a Computational Science and Engineering (CSE) curriculum. For example, they define the objectives of a curriculum, the contents of the individual courses, the learning outcomes or the skills that students should have at the end of their studies of the respective curriculum.

1.2 Target Audience

The following guidance was developed for CSE curricula in rural India and is intended to help in the creation or revision of a CSE curriculum. However, if the implementing partner institution faces one or more of these challenge, then the following guidance can be applied there as well:

- Students do not meet the goals of the curriculum at the end of their studies.
- The implementing institution attempts to ensure student employability exclusively through extra-curricular courses.
- Students feel overburdened by the amount of perceived workload.
- (Potential) Teachers do not have relevant professional experiences or pedagogical skills to implement courses of the curriculum adequately.
- The intended teaching methods and assessment approaches are not executed sufficiently
- The implementing institution has no dedicated structures to evaluate or improve their implementation of the curriculum.

2 Guidance

The following guidance for curriculum designers is intended to help them design the curriculum in a way that enables implementing institutions, despite their challenges, to provide a high-quality study program in which students achieve the goals of the curriculum.

1. The designing HEIs should include information about suitable teaching approaches for the courses of the curriculum.

Example: Particularly for laboratory courses, the curriculum should include instructions on teaching them: First present the task, then let the students work it out by themselves and answer further questions, if any, and only afterwards present the respective solution.

2. The designing HEIs should determine topics for the final project work in a way that the respective topics are not too ambitious, but feasible for the students and assessable for the teachers of the implementing institutions.

Example: A small application for a specific task the students are familiar with instead of building a comprehensive software system.

3. The designing HEIs should include authentic assessment methods and tools for the examinations they prescribe to affiliated HEIs and the end of each semester.

Examples for authentic assessment tools: Presentations, projects, exhibitions, portfolios, case studies or reflection journals.

4. The designing HEIs should provide a handover protocol for teachers taking on one or more courses in the curriculum, along with the curriculum. In this regard, basic materials tailored to teachers should be provided for each course.

Example: Whenever a teacher's course assignment changes, the predeceasing teacher should make the used material available to his or her successor.

5. The designing HEIs should offer courses for (new) teachers of the partner institutions to equip them with basic pedagogical knowledge. Subsequently, teaching guidelines with pedagogical fundamentals as well as teaching and assessment approaches that are suitable for the courses of the curriculum should be developed and provided to the partner institutions.

Example: A course that teaches teachers how to use portfolios and reflection journals in their courses.

6. The designing HEIs should supply curriculum managers with information about possible quality assurance and quality enhancement measurements. Among other things, suitable methods for evaluating the implemented curriculum should be provided on the basis of concrete instructions

Example: Provide implementing institutions concrete ways of evaluating the workload of students and teachers as well as the possibilities of using these results afterwards.

7. The designing HEIs should provide its implementing institutions with a platform to help them exchange ideas, learn from each other and share Good Practices in a similar context.

Example: Establish an annual forum where the implementing institutions meet and report on their learnings of the previous academic year(s).

2.1 Further Readings

In order to follow the guidance, a few examples are listed below:

- 1. The study this Guidance Note is based on [1].
- 2. Comprehensive guide for teaching Computer Science [2].
- 3. Attributes of good teaching in higher engineering education in the context of India [3].
- 4. Authentic assessment of students' learnings [4].
- 5. Assessing course experiences, teaching quality and student engagement [5].
- 6. Suggestions and their costs when designing learning spaces [6].
- 7. Quality of higher engineering education from the students' perspective in India [7].
- 8. Students' perceptions about learning places [8].

References

- [1] Lukas Bürstmayr. "Solution proposals to Higher Software Engineering Education's Design-Reality Gaps in Rural India". Diploma Thesis. TU Wien, 2022.
- [2] Orit Hazzan, Noa Ragonis, and Tami Lapidot. Guide to Teaching Computer Science. An Activity-Based Approach. 3rd ed. Springer, Cham, 2020. ISBN: 978-3-030-39359-5. DOI: 10. 1007/978-3-030-39360-1.
- [3] Aabha Chaubey, Bani Bhattacharya, and Shyamal Kumar Das Mandal. "Attributes of good teaching in engineering education in Indian subcontinent". In: Sādhanā 43.11 (2018), p. 188. ISSN: 0973-7677. DOI: 10.1007/s12046-018-0946-3.
- Zahra Sokhanvar, Keyvan Salehi, and Fatemeh Sokhanvar. "Advantages of authentic assessment for improving the learning experience and employability skills of higher education students: A systematic literature review". In: *Studies in Educational Evaluation* 70 (2021), p. 101030. ISSN: 0191-491X. DOI: 10.1016/j.stueduc.2021.101030.
- [5] Hamish Coates. "6 Tools for effective student feedback". In: *Student Feedback*. Ed. by Chenicheri Sid Nair and Patricie Mertova. Chandos Learning and Teaching Series. Chandos Publishing, 2011, pp. 101–118. ISBN: 978-1-84334-573-2. DOI: 10.1016/B978-1-84334-573-2.50006-9.
- [6] Deborah Lynn Christoffersen et al. "Considerations for creating library learning spaces within a hierarchy of learning space attributes". In: *The Journal of Academic Librarian*ship 47.6 (2021), p. 102458. ISSN: 0099-1333. DOI: 10.1016/j.acalib.2021.102458.
- [7] Pradeep Kumar Choudhury. "Student assessment of quality of engineering education in India: evidence from a field survey". In: *Quality Assurance in Education* 27.1 (Jan. 2019), pp. 103–126. ISSN: 0968-4883. DOI: 10.1108/QAE-02-2015-0004.
- [8] Ronald Beckers, Theo van der Voordt, and Geert Dewulf. "Learning space preferences of higher education students". In: *Building and Environment* 104 (2016), pp. 243–252. ISSN: 0360-1323. DOI: 10.1016/j.buildenv.2016.05.013.



Guidance Note for Curriculum Managers in rural India

Computer Science and Engineering (CSE) Curriculum

Developed by: Lukas Bürstmayr, Paul Spiesberger, Raoul Vallon & Thomas Grechenig

August 16, 2022

This work is licensed under a Creative Commons "Attribution-ShareAlike 4.0 International" license.



1 Context

A challenge of the Indian higher education system that exists for more than a decade is a mismatch between the skills of the engineering graduates and the needs of the industry, especially in the field software engineering. The reasons and recommendations for this skill mismatch have been discussed in the past and explained in the literature and include the high privatization of higher engineering education. The majority of the private higher education sector is composed of over 38.000 affiliated colleges that implement study programs based entirely on the curricula set by their affiliating universities. The majority of these affiliated colleges are located in rural India. The authors of this Guidance Note examined 1 of 80 affiliated colleges in rural India implementing the same CSE curriculum of an affiliating university [1].

Based on the findings of this study, solutions were proposed by the authors that aim to bridge the respective gap, using existing literature and keeping challenges of students and teachers in mind. Based on these proposed solutions this Guidance Note was created that aims to enable curriculum managers to apply some of these proposed solutions.

Caveat: Solutions to problems differ and heavily depend on the context of each Higher Education Institution (HEI). The here stated solutions can only be seen as proposals and require constant local evaluation.

1.1 CSE Curriculum Managers

CSE Curriculum managers are people who are responsible for implementing a Computational Science and Engineering (CSE) curriculum at a HEI. This includes providing the necessary infrastructure, employing and managing teaching staff or ensuring and improving the quality of courses within the scope of their possibilities.

1.2 Target Audience

The following guidance was developed for CSE curricula in rural India and aims to help HEIs that implement a CSE curriculum. However, if the implementing institution faces one or more of these challenge, then the following guidance can be applied there as well:

- Students do not meet the goals of the curriculum at the end of their studies.
- The implementing institution attempts to ensure student employability exclusively through extra-curricular courses.
- Students feel overburdened by the amount of perceived workload.
- (Potential) Teachers do not have relevant professional experiences or pedagogical skills to implement courses of the curriculum adequately.
- The intended teaching methods and assessment approaches are not executed sufficiently
- The implementing institution has no dedicated structures to evaluate or improve their implementation of the curriculum.

2 Guidance

The following guidance is intended to enable the implementing institutions, despite their challenges, to provide a quality program of study in which students achieve the goals of the curriculum.

1. The implementing HEIs should provide guidelines for teaching that contain basic information about how students learn and possible ways of teaching them.

Example: Particularly for laboratory courses, the curriculum management should instruct teachers to first present the task, then let the students work it out by themselves and answer further questions, if any, and only afterwards present the respective solution.

2. The implementing HEIs should develop guidelines for the assessment of students beyond the mandatory mid-term and final examinations, including information on authentic assessment and the tools and methods appropriate for it, but also taking into account the workload perceived by students.

Examples for authentic assessment tools: Presentations, projects, exhibitions, portfolios, case studies or reflection journals.

3. The implementing HEIs should try to collaborate with industry experts within the curricular courses. The industry experts should bring practical experience to the study program.

Example: Industry experts shall supervise project reviews or work with teachers in the development and supervision of project work.

4. The implementing HEIs should avoid changing teachers' course assignments frequently. This will allow them to become more proficient in the subject area of their courses and deepen their respective knowledge in it. Thus, both students and teachers could benefit from the higher continuity.

Example: Increase teacher retention by taking their needs into account.

5. The implementing HEIs have to ensure that the teachers' interests and needs are addressed in order to prevent the necessity of reassigning courses due to a frequent loss of teachers. Nevertheless, the implementing HEIs should provide a handover protocol to assist newly assigned teachers in taking over a course in a way that minimizes knowledge loss through the handover process.

Example: Whenever a teacher's course assignment changes, the predeceasing teacher should make the used material available to his or her successor.

6. The implementing HEIs have to equip new teachers with pedagogical basics and provider guidelines for suitable teaching approaches and assessment methods. This could be achieved by offering courses for (new) teachers of the respective institution to equip them with with the contents just mentioned. Additionally, teaching guidelines with pedagogical fundamentals as well as teaching and assessment approaches that are suitable for the courses of the curriculum should be developed.

Example: A course that teaches teachers how to use portfolios and reflection journals in their courses.

7. A "buddy system" for teachers should be established so that newly hired teachers who have little or no teaching experience are mentored by experienced teachers. Through this

system, new teachers are supposed to be integrated into the HEI from the very beginning and learn from experiences already acquired by their more senior colleagues.

Example: New teachers will be accompanied by a senior teacher in preparing or delivering course units during the first weeks.

8. The implementing HEIs should establish a internal structure or body that regularly evaluates the curriculum and reports possible improvements to the curriculum management based on the evaluation results.

Example: Assign a person to focus exclusively on the evaluation of study programs and the use of these evaluation results. This evaluation should at least consider feedback on the study program by both students and teachers. Additionally, the perceived and objective workload as well as the satisfaction of students and teachers should be monitored and taken into account.

2.1 Further Readings

- 1. The study this Guidance Note is based on [1].
- 2. Comprehensive guide for teaching Computer Science [2].
- 3. Students' perceptions about learning places [3].
- 4. Forms of collaboration between the industry and HEIs [4].
- 5. Attributes of good teaching in higher engineering education in the context of India [5].
- 6. Quality of higher engineering education from the students' perspective in India [6].
- 7. Suggestions and their costs when designing learning spaces [7].
- 8. Assessing course experiences, teaching quality and student engagement [8].
- 9. Handbook for relatively inexperienced teachers [9].
- 10. Retention of teachers in India [10].
- 11. Authentic assessment of students' learnings [11].

References

- [1] Lukas Bürstmayr. "Solution proposals to Higher Software Engineering Education's Design-Reality Gaps in Rural India". Diploma Thesis. TU Wien, 2022.
- [2] Orit Hazzan, Noa Ragonis, and Tami Lapidot. Guide to Teaching Computer Science. An Activity-Based Approach. 3rd ed. Springer, Cham, 2020. ISBN: 978-3-030-39359-5. DOI: 10. 1007/978-3-030-39360-1.
- [3] Ronald Beckers, Theo van der Voordt, and Geert Dewulf. "Learning space preferences of higher education students". In: *Building and Environment* 104 (2016), pp. 243–252. ISSN: 0360-1323. DOI: 10.1016/j.buildenv.2016.05.013.
- [4] Dhruba Borah, Khaleel Malik, and Silvia Massini. "Are engineering graduates ready for R&D jobs in emerging countries? Teaching-focused industry-academia collaboration strategies". In: *Research Policy* 48.9 (2019), p. 103837. ISSN: 0048-7333. DOI: 10.1016/j.respol. 2019.103837.

- [5] Aabha Chaubey, Bani Bhattacharya, and Shyamal Kumar Das Mandal. "Attributes of good teaching in engineering education in Indian subcontinent". In: Sādhanā 43.11 (2018), p. 188. ISSN: 0973-7677. DOI: 10.1007/s12046-018-0946-3.
- [6] Pradeep Kumar Choudhury. "Student assessment of quality of engineering education in India: evidence from a field survey". In: *Quality Assurance in Education* 27.1 (Jan. 2019), pp. 103–126. ISSN: 0968-4883. DOI: 10.1108/QAE-02-2015-0004.
- [7] Deborah Lynn Christoffersen et al. "Considerations for creating library learning spaces within a hierarchy of learning space attributes". In: *The Journal of Academic Librarianship* 47.6 (2021), p. 102458. ISSN: 0099-1333. DOI: 10.1016/j.acalib.2021.102458.
- [8] Hamish Coates. "6 Tools for effective student feedback". In: Student Feedback. Ed. by Chenicheri Sid Nair and Patricie Mertova. Chandos Learning and Teaching Series. Chandos Publishing, 2011, pp. 101–118. ISBN: 978-1-84334-573-2. DOI: 10.1016/B978-1-84334-573-2.50006-9.
- [9] Heather Fry, Steve Ketteridge, and Stephanie Marshall, eds. A Handbook for Teaching and Learning in Higher Education. 4th Editio. London: Routledge, 2014, p. 474. ISBN: 0-203-89141-4. DOI: 10.4324/9781315763088.
- [10] MS Pabla. "Teachers Retention in Higher Educational Institutes". In: PARIPEX Indian Journal of Research 3.7 (2014). DOI: 10.36106/paripex.
- Zahra Sokhanvar, Keyvan Salehi, and Fatemeh Sokhanvar. "Advantages of authentic assessment for improving the learning experience and employability skills of higher education students: A systematic literature review". In: *Studies in Educational Evaluation* 70 (2021), p. 101030. ISSN: 0191-491X. DOI: 10.1016/j.stueduc.2021.101030.